NONI GRAMMATICAL STRUCTURE WITH SPECIAL REFERENCE TO VERB MORPHOLOGY

BY LARRY M. HYMAN



SOUTHERN CALIFORNIA OCCASIONAL PAPERS IN LINGUISTICS NO. 9 August 1981

PUBLISHED BY THE DEPARTMENT OF LINGUISTICS
UNIVERSITY OF SOUTHERN CALIFORNIA
LOS ANGELES, CALIFORNIA 90007

To My FRIEND MARK BINTUM of Nkor

TABLE OF CONTENTS

•
•
)
3
5
0
7

PART V:	COM	PLEX S	SENTE	NCES													
Chapter	19.	Relat	tive	Claus	es				٠								91
Chapter	20.	Seque	entia	l and	Sub	junc	tive	Cla	uses							į.	95
Chapter	21.	Тетро	ral	and C	ondi	tion	C1a	uses									100
Chapter	22.	Inter	roga	tives	and	Foc	us							•			104
REFERENC	ES .	•	:1	•	٠				•	•	•	e.	E I	•	<u></u>	F	109
ENGLISH-	NONI	WORD	LIST		٠	•			•	•	•						110
INDEX											•	•		•			120

PRFFACE

The present volume, *Noni Grammatical Structure*, is the third in a series of SCOPIL publications stemming from work on the languages of the Grassfields Bantu region of Cameroon. Though not a complete grammar, this work attempts to present the major grammatical structures of an important language, which although not properly Grassfields Bantu, belongs to the closely related "Beboid" group (Hombert 1980).

Work was carried out on Noni part time during an eleven week period in 1977 with Mark Bintum serving as principal consultant. Because of the time limitations placed on this study, the results presented here cannot be considered definitive. Nevertheless, we feel confident that the materials described herein will be of use to comparativists and future Noni scholars--cf. the promising start made by Richards (1981). Particular attention is paid to the complex verb morphology, an area which has been studied in depth only in a few languages--none Beboid--in the area.

The grammatical sketch that follows is organized in five parts:

Part I describes the consonants, vowels and tones found in the language.

Part II describes the noun class system and the various pronouns and modifiers which occur in the noun phrase.

Part III describes verb forms, the various tenses and aspects as they occur in different clause types.

Part IV presents the syntax of the simplex clause: its word order properties, expression of grammatical relations and verbal auxiliaries.

Part V goes beyond the simplex clause to treat complex sentences involving relative, sequential, temporal and condition clauses. The section concludes with a discussion of interrogatives and focus.

Following the above 22 chapters and the references, an extended English-Noni wordlist is provided which should be of interest to comparativists and specialists of the Grassfields Bantu borderland.

This project has been made possible by a National Science Foundation grant no. BNS76-81261 which allowed travel to Cameroon as well as research at the University of Southern California. We gratefully acknowledge this support as well as the fellowship the author received from the John Simon Guggenheim Foundation for work on comparative Grassfields Bantu grammar in 1979-80.

A number of persons in Cameroon greatly facilitated our study. First, Dr. Emmanuel Sounjock of the then CE.REL.T.RA (now C.R.E.A.) in Yaounde gave generously of his time and expressed great interest in our project. We thank him and the Cameroonian authorities for the research permits we were granted.

We also are very grateful to Fr. Engelbert Kofon for introducing us to his Noni language, for sharing his materials with us, and for his personal comments on an earlier--now superceded--study of the Noni noun classes (Hyman 1975).

But above all it is our friendship with Mr. Mark Bintum which made this grammar a reality. Mark was the principal Noni consultant both for the 1975 study and for this study. He gave selflessly to this project and was at times more effectively devoted to it than the author himself.

We have made Mark Bintum wait too long to see the fruits of his devoted labor. For all of his giving, patience and (polite) pressure, I thank him and dedicate this book to him.

Larry M. Hyman

Los Angeles, 8/24/81

PART I: PHONOLOGY

1

THE SOUND SYSTEM

1. Vowels. The Noni language possesses a system of seven short and seven long vowels:

i		и	ii		uu
е		0	ee		00
3		o	33		၁၁
	ā			aa	

The above orthographic vowels are pronounced as follows:

- (1) i is pronounced [i], e.g. -ff 'take'. It is pronounced [i:], e.g. -fff [ff:] 'to be fed up with.
- (2) e is usually pronounced [ə], more rarely [e], e.g. -cé [cé] ([cé]) 'pass the night'. ee is usually pronounced [e:], although the pronunciation [ə:] is occasionally heard for some words, e.g. cèe [ce:] ([ce:]) 'palm-kernel'.
- (3) ϵ is pronounced $[\epsilon]$, e.g. $-|\epsilon|[|\epsilon|]$ 'become wet, sink'. $\epsilon\epsilon$ is pronounced $[\epsilon:]$, e.g. $-|\epsilon\epsilon|[|\epsilon:]$ 'deceive'.
- (4) a is pronounced [a], e.g. -sa [sa] 'go ahead'. aa is pronounced
 [a:], e.g. -saa [sa:] 'to tear'.
- (5) ɔ is pronounced [ɔ], e.g. -kɔ [kɔ] 'snore'. ɔɔ is pronounced [ɔ:], e.g. -kɔɔ́ [kɔ̂:] 'catch'.
- (6) o is pronounced a very tense [o], e.g. -do [do] 'say'. oo is pronounced [o:], e.g. -doo [do:] 'be numerous'.
- (7) u is pronounced [ω], e.g. yun [yon] 'thatching grass'. When preceded by a stop or w, u also conditions a labio-dentalization of the consonant, e.g. -kfu [kf ω] 'sift (something moist)'. Although predictable, this labiodentalization is expressed orthographically in the consonant sequences tf, kf, bv, dv, gv, mv and wv. uu is pronounced [u:], e.g. -yúú [yú:] 'be clear, to dawn'. In a few cases labio-dentalization is observed before uu, e.g. kfùù [kfù:] 'rope'.
- 2. Consonants. The following consonants are distinguished in Noni, arranged in the appropriate rows and columns. Rows: voiceless non-continuants, voiced non-continuants, voiceless fricatives, approximants, and nasals. Columns: labial, dental, palatal and velar. These consonants are presented in their orthographic representations except for § (written sh) and γ (written gh). They are pronounced as indicated in the following paragraphs.

- (1) Voiceless non-continuants: t is pronounced [t], e.g. -tew [tew] 'prefer'. c is pronounced [tš], e.g. -cám [tšám] 'chew'. k is pronounced [k], e.g. -kíé [kíé] 'laugh'.
- (2) Voiced non-continuants: b is pronounced [b], e.g. -b55 [b5:] 'bear a child'. d is pronounced [d], e.g. -dee [de:] 'cook'. J is pronounced [dž], e.g. -jéw [džéw] 'go mad, become drunk'. g is pronounced [g], e.g. -ge [gə] 'make, do'.
- (3) Voiceless fricatives: f is pronounced [f], e.g. few [few] 'to measure'. s is pronounced [s], e.g. -séé [sé:] 'tell'. The sound [š] is represented orthographically as sh, e.g. -shé [šé] 'spend the day'.
- (4) Approximants: w is pronounced [w], e.g. -wám [wám] 'scream'. I is pronounced [t], e.g. -léy [léY] 'enter'. y is pronounced [j], e.g. -yén [jén] 'see'. The sound [γ], an unrounded velar approximant, is represented orthographically as gh, e.g. -ghew [$\gamma \wedge^{w}$] 'be jealous'. It occurs relatively rarely in Noni.
- (5) Nasals: m is pronounced [m], e.g. -ma [ma] 'tell (a riddle)'. n is pronounced [n], e.g. -nε [nε] 'to leave'. ñ is pronounced [n], e.g. -ñá [ná] 'give', and is not written ny because of such oppositions as nyèm [njèm] 'dry season' vs. ñèm [nèm] 'animal'. η is pronounced [n], e.g. -nám [nám] 'to pinch', and occurs rarely in syllable-initial position.
- (6) Glottal stop: The glottal stop occurs very rarely in Noni and has been recorded in the following words: Žy? 'no', mba?'button', cá? 'granary made of bamboo', nfò? (name of a society), and nwà? (=nwà) 'book'. Except for 'no', all words with glottal stop appear to have been borrowed, most likely from Lamnso?. ? can only appear in syllable-final position. It may appear after falling tone short vowels occurring before pause, e.g. nam ye = nam ye? 'this animal'.

In addition to the above, a few other consonant types occur in the language. We have already mentioned the fact that t, k, b, d, g, m and w are labiodentalized before u, e.g. /butu/ 'day' is written boutfu and pronounced $[b^vot^f o]$. In addition, the labio-velar stops [kp] and [gb] occur as alternate realizations of k+w and g+w, respectively: kwen 'firewood' is realized either $[k^wen]$ or [kpen]; gwɔʻɔ́n 'bamboo' is realized $[g^wj:n]$ or [gbj:n], etc. We have chosen to standardize this variation with orthographic kw and gw.

3. Syllable structure. The majority of Noni syllables begin with a consonantal onset, continue with a vocalic nucleus, and terminate either in Ø or in one of a limited set of final consonants, as summarized in the following formula:

(N)
$$C_1$$
 (G) V_1 (V_2) (C_2)

The consonantal onset may consist of up to three consonants: a non-syllabic nasal (N), a stem-initial consonant (C_1) , and a w or y glide (G),

e.g. $-\eta g \circ y$ 'break' (by snapping), $-\eta g w \circ y$ 'go sour'. The remaining source of initial consonant sequences, that involving orthographic C+f and C+V, has been analyzed as labiodental affrication. Of the consonants presented in section 2, only the glottal stop cannot initiate a syllable. The rare consonant gh occurs only in this position, while very few syllables have η as their C_1 consonant. Cases of C+y are relatively rare, e.g. -fyéy 'mix'. w occurs frequently after velars in both nouns and verbs, e.g. -gwe 'fall', $\eta w \circ w$ 'square granary'. It has been recorded once after sh (-shwaw 'to scratch') and after t, b and f in nouns only, e.g. two 'witch', fwe 'pregnancy'.

NC1 sequences have three sources:

- (1) In nouns, especially (though not exclusively) in genders 1/2 and 9/10, a NC₁ sequence is often noted, e.g. nton/bonton 'pot/pots' 1/2, nd \sqrt{u} 'reloth/cloths' 9/10 (but also kensan/binsan 'net/nets' 7/8).
- (2) Verbs with a first person singular subject take a homorganic nasal agreement marker (including auxiliary and consecutivized verbs), e.g. me nyemté mbinè 'I am singing and dancing' (cf. wvù yemté binè 'he is singing and dancing').
- (3) A few verb stems and adverbials inexplicably begin with a NC $_1$ sequence, e.g. -mby 4 y 'tell a lie', nt 4 ay 'well' (adv.).

The only syllables lacking a consonantal onset are prepositions, conjunctions and certain noun prefixes which consist of a vowel alone, e.g. ϵ 'if', ϵ - (class 6 prefix, as in ϵ -fom 'fat').

The *vocalic nucleus* consists in most cases of a single short or long vowel, although some sequences of two unlike vowels are possible (see below). If there is no final consonant, both short and long vowel syllables are frequently attested, e.g. -ka 'clear one's throat', -kaa 'grind'. When there is a final consonant, long vowels are found relatively rarely, and only before m and n. Most cases of a long vowel + m involve either the -m positional extension on verbs (e.g. -j)im 'lie down') or the class 12 and 15 -m suffix on nouns (e.g. meèm 'oil' 15 [Bantu 6a]). Syllables ending in a long vowel + n are rare, e.g. ntèèn 'many'. In cases where a nasal is followed in the same word by a homorganic consonant + vowel, it is always assigned to the following syllable. Thus, -kaaŋkè 'be frying' (from -káŋ 'fry') is syllabified as -kaa.ŋkè.

In addition to the rare glottal stop discussed above, only the consonants w, y, m, n and η can occur as the *final consonant* (C2). w occurs after all vowels (although instances of uw are rare):

```
-jíw 'to poison' gvùw 'chest'
-kew 'gather firewood' -kow 'become wealthy'
-kew 'start' -gow 'grind'
-kaw 'promise'
```

All vowels occur before final y except i:

It should be noted that ϵ_W and ϵ_Y are pronounced $[\Lambda^W]$ and $[\Lambda^Y]$, respectively, after the velar consonants k, g and gh, e.g. $[k\Lambda^W]$ 'start', $[k\Lambda^Y]$ 'groan in pain'.

All seven (short) vowels contrast before final m, n and $\eta\colon$

```
.i im
       'back'
                                -kfúm
                                       'to heat'
.ièm
       'war, grave'
                                -kom
                                        'to touch'
      'part friends forever'
-kem
                                mcp-
                                       'look for'
              -kam squeeze
-bin
      'to dance'
                                -kfun
                                       'to hit'
-ben
      'rise'
                               -kon
                                       'to slander'
-bến 'call'
                                -bon
                                       'stab'
                 -kan 'scrape'
-cin
      'tremble'
                               -tfún
                                       'be inclined'
-yeŋ
      'thigh'
                               -kon
                                       'chase'
gwen
      'root'
                               -kəŋ
                                       'like'
                 -kán 'fry'
```

4 .

 ϵm is pronounced [Am] after velars, e.g. [kAm] 'part friends forever', while $\epsilon \eta$ is pronounced [An] after all consonants, e.g. [gwán] 'root'

THE TONE SYSTEM

1. Each syllable carries contrastive tone in Noni. Three level (register) tones characterize the language: H (high), M (mid) and L (low). These three levels are represented by means of the three accent marks \acute{a} , \overline{a} and \grave{a} , as seen in the following examples:

```
      bíé 'fish (pl.)'
      fwéw 'thorn'

      ble 'goats'
      kwen 'firewood'

      blè 'fish (sg.)'
      gwèw 'shoulder'
```

In our orthography we shall propose that M tone be left unmarked, unless there is some explicit need to mark it (e.g. where there is a ML falling tone on one vowel; see below). Thus, 'goats' will be written bie, and 'firewood' will be written kwen.

The language also makes use of a number of contour tones involving a change from one of the three tone levels to another within the same syllable. Of the six logical possibilities (LH, LM, MH, HL, MH, ML), only MH does not exist in the language. The remaining five contour tones are all found on monosyllables, as in the following noun examples:

```
LH:
        twě
            'witch'
                                  bìén
                                       'bile'
LM:
        wan 'child'
                                 bìe
                                       'goat'
             'mistake' (=only example)
HL:
        jây
            grazing grass'
HM:
        gay
                                 gás
                                       'spear'
ML:
        lew 'oath'
                                 mεèm
```

Except in certain class 5 nouns, the HM contour tone is rare in Noni, as is the HL tone. LH, LM and ML all frequently occur on both nouns and verbs. Finally, two nouns have been found with a complex rising-falling (LHL) tone: mun 'arrow', ncî 'heel'.

In all cases the domain of tone is the syllable. Thus, in syllables beginning with a NC sequence, the nasal always carries the phonetic pitch of the beginning point of the tone of its syllable, e.g.

[Ń]: mbásé 'soups'

 $\left[\overline{N}\right]$: ndvu 'cloths' ncom 'drums' $\left[\overline{N}\right]$: ncom 'drum' nd $\sqrt[3]{u}$ 'cloth'

The starting point of a contour tone generally seems to begin in the C_1 consonant, especially in the case of LM and LH, and especially when the C_1 is wory, e.g. L tone $[\grave{y}]$ in both \grave{y} ew 'house' and yew 'in the house'.

Where a contour tone is realized over two vowels, we shall mark the beginning point on the first vowel and the end point on the second, e.g. bien 'bile'. As seen in twe 'witch' and jây 'mistake' above, a stands for a LH rising tone and â for a HL falling tone, both realized on a single vowel. As seen in other examples such as wan 'child', other contours require two

tone marks when there is only one vowel to receive them. Ideally our orthography should have the following single contour marks: a (LM), a (HM) and a (ML).

- 2. There is no downdrift in Noni. Thus, a H-L-H or a M-L-M sequence will be realized with both H (or M) tones on the same phonetic pitch level. The following variations should however be noted:
- (1) While H and L do not vary appreciably in their realization, M is optionally lowered to 'M (a lowered mid tone) in two environments: (a) when preceded by L and followed by H, and (b) when preceded by H and followed by L. That is, both L-M-H and H-M-L are likely to be realized with the M tone slightly lower than its normal level in other contexts (e.g. H-M).
- (2) The LH rising tone is realized as a phonetic LM before a H or L tone or before pause. It is realized as phonetic LH only before a M tone, e.g. fòwðy [___,] 'at the market', fòwðy fome [___,__] 'at my market'. Similarly, the HL falling tone is frequently realized ML after L or after pause, although this lowering of HL to ML seems to be optional, varying within the speech of the same speaker.
- (3) The LM rising tone is always realized L'M (i.e. rising to a lowered mid pitch level). Similarly, the ML falling tone is always realized as 'ML. Thus, the ending point of LM and the beginning point of ML are slightly lower, phonetically, than the M tone occurring in non-contour contexts, as summarized below:

W	RITTEN	PRONOUNCED	
	yěw	[ȳəʷ]	'in the house'
	ýew .	[yau]	'house'
	jây	[ˈʃâʏ]~[ˈʃ <u>a</u> ʔ]	'mistake'
	lew	[låw]	'oath'
cf.	kwen	[k₩ə̄n]	'firewood'

(4) A L-H sequence is realized L-LH (phonetic L-LM except before M) when the two syllables occur in close grammatical association, e.g. wáy 'market', but: fòwåy 'at the market' (from fò+wáy).

Aside from the above, nouns do not undergo tonal alternations in context, but rather carry their same isolation tone when preceded or followed by other tones. (There are, however, some tonal changes that take place when deriving a plural from a singular; see Chapter 3). Verbs, on the other hand, undergo various tonal alternations in different parts of their paradigm, as noted in chapters dealing with the various tense/aspects of the verb. It should be noted that in these alternations, M is sometimes related to H, sometimes to L.

PART II: THE NOUN PHRASE

3

Noun Classes

Noni, like neighboring related languages, possesses a full system of noun classes, which condition agreement on modifiers and pronouns in various parts of the grammar. In this chapter singular-plural formation is outlined, followed by a treatment of pronouns and noun modifiers in subsequent chapters. The numbering used for these classes is, unless otherwise indicated, to be interpreted as representing the noun classes carrying the same numbers in Narrow Bantu. In the case of the plural class of class 11 [=Bantu class 19], and two of the locative classes, no Bantu equivalent is known. Emphasis here is on the present properties of the noun classes. A separate study on the significance of these classes for Bantu studies is available in Hombert (1980).

In the following sections each class will be given a number. A noun class is either singular or plural in nature in most cases. A singular/plural pairing is termed a *gender*.

1. Gender 1/2. Nouns in gender 1/2 have a \emptyset prefix in the singular, and a bo- prefix in the plural. Roughly two thirds of the nouns in this gender have an initial NC₁ sequence. The homorganic nasal is not interpreted as a prefix, however, since it occurs in both the singular and the plural:

nfàn WITH NASAL: 'chief' banfah 'chiefs' n ćdm 'dwarf cow' nédméd 'dwarf cows' WITHOUT NASAL: cšn 'thief' bacán 'thieves' bvulè 'lion' bòbvulè 'lions'

A considerable number of human nouns fall into this gender, although most 1/2 nouns are not humans. This gender also includes many borrowings, e.g. nama 'tobacco' (English), saka 'news' (Mungaka). The tone of the bo- prefix is generally predictable: it is normally L except in the case when the stem begins with a H tone (as in 'thieves' above). Since very few nouns begin with a H tone in 1/2, M tone bo- will be rare. The plural form 'chiefs' above is an exception. Other exceptions include: bofilm 'people' (sg. wè), bokenè 'women' (sg. kwoòn), bonfaà 'slaves' (sg. nfwà). The singular/plural forms twě/bocéw 'witch/es', in addition to their segmental irregularity in deriving the plural from the singular, show the same tonal irregularity as 'thief/ves' above. It is interesting to note that all tonally irregular plurals of this type involve human nouns. A different kind of tonal irregularity is seen in the following examples (the only of their kind): wanga/bòwanga 'rabbit/s', mun/bòmun 'arrow/s'. Here we see that with the addition of the plural prefix, the tones of the singular shift position (to the right in the case of 'rabbits', to the left in the case of 'arrows').

The following nouns irregularly undergo a modification of their \mathcal{C}_1 consonant in deriving the plural from the singular:

wè	'person'	míiñcd	'people'
wan	'child'	mccd	'children'
twě	'witch'	wascd	'witches'
kwoòn	'woman'	bakεnè	'women'
nfwà	'slave'	bonfaà	'slaves'

In all of these exceptional nouns, the singular involves a w (either as C_1 or following the C_1), which is not present in the plural (cf. PGB *ù-/ba-).

In addition to the human nouns, gender 1/2 contains a second definable group of nouns which are nominalizations derived from verbs. In all but two cases (given below), nominalizations involve a NC1 initial:

```
'message'
                                           pl.
                                                bàntom
                                                            (-tom 'send')
 WITH NASAL:
                  ntom
                                                            (-gow 'grind')
                        'grinding stone'
                                           p1.
                                                wćpnćd
                  wégn
                                                            (-cón 'steal')
                  cšŋ
                        'thief'
                                           p1.
                                                bocón
WITHOUT NASAL:
                                                            (-kan 'swear')
                  kàn
                        'swear word'
                                           pl.
                                                bàkàn
```

(The noun Tew 'oath' may be derived from the verb -lew 'avoid, restrict', although the tones do not appear to line up.)

2. Gender 3/4. Containing considerably fewer nouns than the large 1/2 gender, gender 3/4 nouns are characterized by a C_1 w sequence in the singular, with the w disappearing in the plural:

```
twén 'vine branch' pl. tén
kwen 'firewood' pl. ken
gweè 'ceiling' pl. geè
```

The three nouns above represent the three tone patterns found on monosyllabic nouns in this gender. Only two bisyllabic 3/4 nouns have been found:

```
mbwesem 'green grasshopper' p1. mbesem
ngwénsé 'earthworm' p1. ngénsé
```

Only one irregular noun has been found in 3/4:

```
wế 'field' pl. gế
```

The number 'hundred' seems once to have been a 3/4 noun:

```
gwéé 'hundred' gée fé 'two hundred'
```

3. Gender 3/13. A few class 3 nouns take their plural in class 13 (prefix ji-) instead of class 4. Of the six nouns in this gender, four involve irregularities in the plural:

```
pl. jidwεè
REGULAR:
               dwεè
                     'bridge'
               bwaw
                     'rat (sp.)' pl. jibwaw
                                      jiwáci
               way
                     'market'
                                 pl.
IRREGULAR:
               ŋgóy
                    'fire, gun'
                                 pl. jingóci
                                               'guns'
                     'mountain' pl. jikomè
               kom
               w∨ún
                     'tail'
                                 pl.
                                      liwante
```

4. Gender 5/6. Class 5 is marked by either \emptyset or an -e suffix. Plural class 6 is marked by an ϵ - prefix. The following nouns represent the tonal changes that occur in going from the singular to the plural in both bisyllabic and monosyllabic nouns:

```
BISYLLABIC:
                  lówe
                         'bean'
                                          pl.
                                                εlów
                 temé
                         'axe'
                                          p1.
                                                εtem
                 mbεnè
                         'breast'
                                          p1.
                                                emben.
MONOSYLLABIC:
                 ກນັ້ນ
                         'knee'
                                          pl.
                                                ខ្យារំ
                 เล้
                         'compound'
                                          pl.
                                                ela
                 leè
                         'pumpkin'
                                          pl.
                                                εlēh
                         'name'
                 die
                                          pl. εdie
```

In bisyllabic nouns, singular H-M, M-H and M-L correspond, respectively, to plural M-H, M-M and M-ML. In monosyllabic nouns, singular HM, H and ML (of which there is only one example) correspond, respectively, to plural M-H, M-M and M-ML. A fourth monosyllabic tonal class, M, becomes M-M in the plural.

A number of segmental irregularities are noted in nouns of this gender. Three nouns exhibit a final η in the plural (reminiscent of certain verb forms treated in Chapter 10):

```
góo 'spear' pl. ɛgóŋ
gíi 'rice, guinea corn' pl. ɛgíŋ
leè 'pumpkin' pl. ɛ[ē)
```

In addition to the above, the following exceptional singular/plural alternations have been noted:

tíi	'belly'	p1.	εtέw
díw	'day'	p1.	ediow
tε	'stone'	pl.	εta
bóó	'angle'	pl.	Ebow.
jisé	'eye'	pl.	εjisε

As can be seen in some of the above examples, not all class 5 nouns end in a final -e. In some cases it appears that this suffix has assimilated to the preceding vowel (e.g. 950 'spear', from earlier *g5ne). In other cases there may not have been a final -e. Note, finally, that the final vowel of nsoyi 'cooking stone' (pl. ϵ nsoy) is -: rather than -e because of a general process in Noni whereby suffixal -e becomes [i] after y (cf. the progressive aspect formation process involving the change of -ye to -yi, rather than to -ye).

5. Gender 5/13. Some nouns whose singular is in class 5 take a plural in class 13, rather than in class 6.

```
nfóón 'leaf' pl. jinfóón
lemè 'work' pl. jilemè
biné 'dance' pl. jibiné
```

Unlike the class 13 in gender 3/13, the 13 of 5/13 does not introduce segmental irregularities. One noun can occur in either 5/6 or 5/13:

```
lemé 'tongue' pl. siem /.jilemé
```

6. Gender 7/8. By far the largest gender, in terms of the number of nouns belonging to it, is 7/8. Class 7 is characterized by the prefix ke-, which can take M or L tone. Class 8 is characterized by a like tone prefix bi- (sometimes realized be-):

```
kebow 'hole, room' pl. bibow
kecaw 'belt' pl. bicaw
keban 'palm tree' pl. biban
```

Approximately one third of the nouns in class 7 can be realized with the prefix ε - (identical to class 6, but still taking class 7 agreement) instead of ke-. It cannot be predicted which nouns allow this alternative realization of their prefix. It seems, though, that many of the most common words fall into this category:

```
kebś / sbś 'hand' ketó / stó 'ear' kete / ste 'tree' kege / sge 'tooth'
```

Only one noun has been found which must have ϵ - as its prefix:

```
elemé / *kelemé 'blood'
```

The following unexpected segmental changes occur in singular/plural formation in gender 7/8:

```
pl. bigew
         'tooth'
 ege'
 efowe
         'medicine'
                          pl. bifóó
                          pl. bikelè
 εκειὲ
         'bag'
                          pl. biléém
 εlεmé
         'blood'
                          pl. bikaa
 εkalε
         'foot'
                          pl. bilémté
· elemte 'wound'
```

It should be noted that many of the class 7 nouns which allow ϵ - as their prefix also end in what appears to be an older - ϵ suffix (e.g. 'medicine', etc. in the above examples).

7. Gender 9/10. Nouns in gender 9/10 are characterized by a lack of a prefix (or suffix), but by tonal alternations in deriving the plural from the singular. The tonal alternations encountered in this gender are illustrated below:

```
iom
    'antelope'
                         pl.
iòm
                              bíé
bìè
    'fish (sg.)'
                         p1.
    'goat'
                         p1.
bie
                              bie
                         pl.
                              bwé
bwě
    'dog'
```

As seen above, almost all 9/10 nouns are monosyllabic, a few exceptions being the following:

```
mbàsè 'vegetable' pl. mbásé
mfumè 'caterpillar' pl. mfumèn
cuma 'locust' pl. cuma
```

When the singular has L tone, the plural can be with either ML or H tone. When the singular has LM tone, the plural has M tone, and when the singular

has LH tone, the plural has H tone. The four monosyllabic nouns above reconstruct with an earlier prefix tone (L in the singular, H in the plural) and an additional final stem tone, as follows: *'jòm'/'jòm' 'antelope/s', *'bie'/'bie' 'fish', *'bie'/'bie' 'goat/s', *'bwé'/'bwé' 'dog/s' (<PGB i-/i-).

A little less than one third of the nouns in 9/10 begin with a $\ensuremath{\text{NC}}_1$ sequence, e.g.

```
njèè 'sheep' pl. njeè
nsàn 'friend' pl. nsán
ndvu 'cloth' pl. ndvu
```

Most animals are in the 9/10 gender, with the most common ones generally occurring without a homorganic nasal (e.g. 'fish', 'hen', 'dog', 'snake', 'goat'). Nouns which begin with a nasal consonant do not show a N vs. NN opposition, e.g. nam 'animal', pl. nam.

The nouns can (pl. can) and shem 'liver, fruit (earlier: heart)' (pl. shem) clearly show that 9/10 was once characterized by an i- prefix. These nouns appear in related languages with initial [k] and [t], respectively, which have undergone palatalization in Noni (cf. the realization of the number 'three' in Chapter 7).

8. Gender 11/12 (PGB 19/18). Nouns in gender 11/12 are characterized by a fe- (sometimes realized fi-) prefix in the singular, and a mun- prefix (and sometimes an additional -m suffix) in the plural:

```
fèmbì 'kolanut' pl. mùmbìm 'kola trees' fyeè 'thing' pl. mwēm
fèncaw 'ring, wire' pl. mùncaw
```

The prefix tone can be either L or M. When a noun ending in M takes a -m suffix in the plural, its M tones goes up to H:

```
fetene 'squirrel' pl. munteném
fèñò 'knife' pl. mùñòm
feñíi 'bird' pl. muñíím
```

11/12 contains many diminutivized nouns, which can come from any of the other genders (except 14/15), as seen below:

```
fènɔ / mùnɔm 'knife/ves' < nɔ / nɔ 'machet/s' 9/10
fete / muntém 'stick/s' < kete / bete 'tree/s' 7/8
fèntèw / muntèw 'small village/s' < ntèw / bɔntèw 'village/s' 1/2
```

What is interesting is that in deriving the diminutive, the same segmental irregularities found in the plural of the basic class arise:

```
fekenè / mukenè dim. of: kwoòn / bokenè 'woman/women' 1/2
fewáci / muwáci dim. of: wáy / jiwáci 'market/s' 3/13
fegóo / mungón dim. of: góo / egón 'spear/s' 5/6
```

9. Gender 14/15 (PGB 14/6a). Nouns whose singular are in class 14 begin with a bvu- prefix, which usually carries M tone. Relatively few class 14 nouns take a plural, although when possible, this plural is always in class 15 (prefix mon-; suffix -m, with the same tonal changes as in class 12):

```
'face'
                          pl. monshim
byushiw
         'place'
                         pl. mondvúúm
hvudvúu -
                          pl. muŋkóm
bvukśw
         'ladder'
         † fufu!
                            (no pl.)
bvulem
bvutow
         'honey'
                            (no pl.)
bvul≅ŵ
         'intestines'
                            (no pl.)
```

Included in class 14 (and also not taking a plural) are abstract and quality nowns, most of which are derived from verbs, taking a bvu- prefix and an $-\epsilon$ suffix:

```
< -dēfš
                                       'become long'
byudéfè
          'length'
                                       'become fat'
bvufánè
          'fatness'
                           < -fan
bvukfúfè
          'shortness'
                           < -kfúfé
                                       'become short'
                           < - Tîtě
                                       'become heavy'
bvujítè
          'weight'
```

The eight days of the week also belong to this class, though with a H tone on their prefix (note also the unexpected $[\check{z}]$):

```
bvútfū 'day 1' bvûnkaàdēh 'day 5'
bvûnkaà 'day 2' bvúžíidēh 'day 6'
bvúží 'day 3' bvúsów 'day 7'
bvúkèmā 'day 4' bvúmbòn 'day 8'
```

bién 'bile'

Finally, it can be noted that class 14 is normally lacking from Grassfields Bantu languages.

Single class genders. In addition to the above singular/plural genders, some nouns occur in a singular class without a corresponding plural, and some occur in a plural class without a corresponding singular. The reason why a corresponding plural or singular does not exist is usually semantic (e.g. it doesn't make sense to speak of 'fatnesses'). Examples are given below:

```
'profit'
                                    SEW
              bòn
                    'brain'
class 1 :
              nkån 'cornbeer'
                                    bilen 'groundnut'
             (no class 2 noun exists without a class 1 singular)
                                    fwéfwé 'wind'
              abên
                    'tusk'
      3
                                            'fire' (13 = 'guns')
                                    ngóy
              fwem
                    'chalk'
              fīw
                                    bábá
                                           'gall-bladder'
                    'kidney'
    4=10:
                                   mbyay 'lying'
              n.] Tm
                    'smoke'
      5
             (none)
                                           'twins'
      6
              efom 'fat'
                                    €mam
                                          'ear'
                                    ketó
      7 :
              kèmbèw
                      'cloud'
                                          'arm'
              kedéw
                      'beard'
                                    keyi
             (an alternate word for 'ear', kentonton, has the regular
              plural form bintonton; the plural of 'arm' is irregular
              and suppletive: hkenken, class 4=10).
             (none)
      8 :
                                    abèw 'shoulder'
      9
              jàŋ
                    'rain'
```

class 10 : (see class 4)

11 : (none)

12 : mùngày 'strength'

13 : jitâ 'ribs'

14 : byùyèw 'flatness' byùfèé 'intelligence'

(quality nouns; see above)

15 : meèm 'oil' monkfunyè 'pus'

mbyum 'wine' monkonè 'eyebrows'

mum 'flour' monjam 'urine'

(mass-liquid nouns) monshéém 'sand'

Locative classes. In addition to the above 15 noun classes, Noni has three locative noun classes, which are numbered 16, 17 and 18. Of these, only 16 has a consistent prefix (fo-). All three classes vary in meaning from 'in', 'at', 'to', etc., with locative class apparently being unpredictable. These three classes are best illustrated by means of a possessive phrase:

class 16 : fòjisé fo wan 'the forehead of the child' (lit. at/on eye class 17 : èbɔ jo wan 'in the hand of the child' of child) class 18 : yew dvu wan 'in the house of the child'

The markers fo, jo and dvu are conditioned by the locative meaning of each phrase, as seen from a comparison with the following:

class 5: jisé ji wan 'the eye of the child' class 7: ɛbɔ ke wan 'the hand of the child' class 9: yew yi wan 'the house of the child'

In the above examples it is seen that these locative classes are derived from other noun classes, much as the class 11 diminutive derives from other classes (e.g. kete 'tree' 7 becomes fete 'stick' = "small tree" 11). But just as some class 11 nouns do not have a basic non-diminutive correlate, some locatives do not come from basic nouns, e.g.

class 16: fòfa 'on the head' (cf. ɛkɔlɛ 'head')
fòfalaan 'at the summit' (e.g. of a mountain, tree)

class 17: Emû 'in the mouth' (cf. dlow 'mouth')

èkǔ 'under'

class 18 : cin 'around the waist'

Only three cases have been found of class 18: yew 'in the house' (cf. yew 'house' 9/10), con 'in the hut' (cf. con 'hut' 9/10), con 'around the waist'. It should be noted that while class 16 corresponds to class 16 *pa- in Bantu, no correlates exist for Noni classes 17 and 18 either in Bantu or in other neighboring languages.

In deriving a locativized noun, two possible alternate strategies can be used: (i) one can change the noun class to 16, 17 or 18, according to the noun, in which case the fo, jo and dvu agreement markers appear; or (ii) one can maintain (at least in many cases) the original noun class identity of the non-locativized form and simply add a locative suffix (-|é after vowels, -é after consonants), e.g.

fàjem fo wan fàjem wan é 'at the child's grave'

While not every locativized form allows this second alternative, whenever a locative is not converted to class 16, 17 or 18, the locative suffix is obligatory, most frequently co-occurring with the prepositions & 'in' and yi 'on':

in the book'
 in the book'
 in my car'
 in my car'
 in my car'
 in my car'
 in the child's house'

In some rare cases the locative suffix appears without a preposition, e.g.

พอิกู wềm 'my country' พอิกู wềm é 'in my country'

The prepositions & and y' and the postposition le'e are not considered to be markers of noun classes, since they do not condition agreement markers of their own.

PRONOUNS

1. The independent (simplex) pronouns in Noni are as follows:

		eingular		plural	
1st person		me		bèsèn beènè	(excl)
2nd person		ćw		bèn	
3rd person	LOG	wen		bòwen	
	class 1 3 5 7 9 11 14 16 17 18	wvù wvú jí kể yì fể bvú (fổ) (jổ) (dvú)	2 4 6 8 10 12 13 15	bố yí yố bí yí mú jí mố	

The above table presents both personal pronouns as well as the independent forms of pronouns for each noun class. A few comments are in order.

- (a) Noni distinguishes between exclusive and inclusive 'we'. The form besen is used when the speaker wishes to exclude the hearer, while the form beene is used when the speaker wishes to include the hearer. As we shall see below, there is an exceptionally complex set of compound pronouns which can be used to identify the individual make-up of plural personal pronouns.
- (b) Noni has both a singular and a plural logophoric (LOG) pronoun. These forms (wen and bowen, respectively) are used in reported speech, whenever a pronoun in an indirect discourse refers back to the person or persons reporting the event. Examples are contrasted with the regular or non-logophoric subject pronouns below:
 - sg. wvù dòó le wvù béè gèn fòwǎy 'he says that he [another] went to market' wvù dòó le wen béè gèn fòwǎy 'he says that he [himself] went to market'
 - pl. bố dốc lẽ bố bết gèn fòway 'they say they [others] went to market' bố dốc lẽ bồwen bết gèn fòway 'they say they [themselves] went to market'

In the first sentence of each set, the regular singular and plural independent pronouns are used (wvù 'he' and bɔ 'they', respectively). The resulting meaning conveyed is that the person(s) going to the market is not the same as the speaker(s) reporting the event. In the second sentence of each set, the logophoric pronouns are used (wen 'he/she [reporting person]' and bowen 'they [reporting persons]'). The result is that the same person(s) reporting the event personally went to the market. It is interesting to note that these pronouns

are probably borrowed from Oku, Aghem or another Ring language, where [wén] is the class I demonstrative pronoun meaning 'this (one)' (cf. the treatment of demonstratives and logophoric pronouns in Hyman 1979a). While Ring languages use this form for the non-logophoric meaning, Noni seems to have been first sensitized to the logophoric distinction, and then borrowed the form [wén] with the opposite meaning.

(c) The pronoun forms for classes 16, 17 and 18 (the locative classes) are given in parentheses to indicate that they are not used in subject position. They are however used in many of the other contexts where the other class pronouns are acceptable (e.g. as a verb complement [object]).

Except for the locative classes, then, the above forms are used as subjects and objects, after prepositions, and after the verb 'to be', as illustrated with the pronoun me 'I/me' below:

subject: me nóà ngwè 'I have fallen'
object: wvù kfùně me 'he is hitting me'
after prep: bá nóà ñà ké ɛ me 'they have given it [cl. 7] to me'
after 'be': È nǔ me 'it's me'

In the form 'they have given it to me' we have a class 7 object pronoun ke as well as a prepositional object pronoun me 'me'.

The locative classes can also be used as objects with the meaning of 'to it', 'in it', etc. In addition, the class 18 pronoun dvú has the partitive meaning 'of it' or 'with it' as seen in comparing the following two sentences involving the verb 'to give':

```
bố nổo mà me bố kèngòm 'they have given me plantains' bố nổo mà me dyú 'they have given me it'
```

The first sentence literally means 'they have given me with plantains'. Since dvú "replaces" bố kèngòm 'with plantains', it is best translated 'with it'. This pronoum therefore is a prepositional object, i.e. standing for a preposition plus an inanimate object from a non-locative class. In the above utterance it stands for bố + kế, which cannot occur with the verb 'to give'. (The form bố-kế is acceptable as a complex pronoun meaning 'they including it [cl. 7]'; see below.) The partitive meaning of dvú is seen in a comparison of the following sentences:

```
bố nổô diế kế 'they have eaten it'
bố nổô diế dvú 'they have eaten some' ['of it']
```

Finally, note the meaning of dvú in the following example:

```
bố nổo mã wan dvú 'they have given to the child too'
```

In this case dvú means 'too' (in addition to giving to someone else). Understood is that dvú stands for 'with it', and the 'it' refers to the giving of something to someone else. It is thus not hard to see how the prepositional and partitive senses have led to this nuance.

It is interesting to note that the class 1-15 pronouns given above can be used only as independent pronouns. That is, they do not co-occur with the noun to which they refer. As we shall see in Chapter 16, the only subject-verb agreement found in Noni is the homorganic nasal prefixed to all verbs having

me 'I' as their subject. The class pronouns are found with slight tonal modifications to be used as relative clause markers: with LM tone if the relativized noun is the subject of the relative clause, or with L tone if it is not (see Chapter 19).

2. In addition to the above simplex pronouns, Noni is rich in compound pronouns. These pronouns consist of a first part, which indicates the person (1st, 2nd, 3rd), and a second part, which indicates number and the possibility of logophoric reference. The first parts are, respectively, beè- (1st person), bèn-e- (2nd person), and bò- (3rd person). In the case of bèn- and bò-, we are dealing, of course, with the plural pronouns 'you pl.' and 'they'. The possible second parts are the pronouns wò 'you sg.' and bèn 'you pl.', which can be added only to beè- (1st plural); wvù 'he/she' and bò 'they', which can be added to all three first parts, and wen 'LOG sg.' and bòwen 'LOG pl', which also can be added to all first parts. The pronoun beenè 'we [incl]' is derived from beè+bèn in an irregular fashion. In exactly one situation we are able to find a pronoun with three parts: when there is a first, a second and a third person involved, we derive forms such as beènèbò 'we [including you and he, she or them]'. The result is the following sixteen compound pronominal forms:

	dual		plural	
1+2 :	beèwò	'you sg. & I'	beènè	'you pl. & I/we'/'you sg. & we'
1+3:	beèwvù	'he & I'	beèbɔ̈́	'he/they & we'/'they & I'
2+3:	bèn ɛ ˈwvù	the & you sg.	bènèbŏ	'he/they & you pl.'/'they
3+3 :	bŚwvù	'he & he'	Ď ó bở	<pre>% you sg.' 'he/they & they'</pre>
1+2+3 :		171	beènèbš	'he/they & you sg./pl. & I/we'
1+LOG :	beèwen	'he & he LOG'	beèbàwen	'he/they & they LOG'/'you pl. & he LOG'
2+LOG:	benewen	'you sg. & he LOG'	bèns bòwen	'you sg./pl. & they LOG'/'you pl. & he LOG'
3+LOG :	bówèn	'he & he LOG'	bố bồwen	'he/they & they LOG'/'they & he LOG'
1+2+LOG :		t	peènè Èbòwen	'he/they LOG & you sg./pl. & I/we'

As can be seen, dual forms are distinguished from plural forms. By "dual" is meant that there are only two persons represented in the compound pronoun; plural compound pronouns refer to more than two persons. Appropriately, there are no dual forms for 1+2+3rd person or for 1+2+LOG, since a minimum of three people is required. What is important in the above forms is that these compounds do not reflect the composition of subgroups in the plural forms: they merely indicate which persons are involved in the total group. Thus, for example, beeb can refer to 'he and we', 'they and we' or 'they and I'. Any combination of 1st and 3rd person pronouns yielding at least three participants is equally representable by this form. This is particularly striking in the case of logophoric pronouns, as seen in the following two sentences:

wvù dòó le beèbòwen béè gèn fòwðy 'he says that he and we [or they and I/we] went to market'

In the first example the subject of 'say' is plural and so is the compound pronoun in the lower clause. The result is that the bowen part of the compound pronoun refers back tothe subject bo, as we would expect. In the second sentence, however, one of the possible interpretations is 'he says that he and we went to market', in which case the bowen part is in plural form not because the subject of the verb 'say' is plural, but because the subject wwo 'he' combines to form a plural compound. Thus, we can say that each part contributes its features (person, number, logophoric reference), but that the resulting compound "jumbles" up the features and does not reflect the internal composition of the two (or three) conjoined groups.

Pronoun compounding thus works as follows: (a) For person, if there is a [1st person], use bee. If not, but if there is a [2nd person], use ben-\(\varepsilon\). Finally, if there is neither a [1st person] nor a [2nd person], use bee. (b) For number, if both parts are marked [+singular], use a singular pronoun we, with or wen (according to the identity of the participant whose person may not be represented in the first part of the compound-bee is considered to be non-logophoric). If either of the components is plural, or if there are three singular components, use the plural pronouns in second position.

Finally, it should be noted that non-personal pronouns can combine with $b\delta^2$, e.g.

1 2 3	bó-w∨ù bó-bó bô-w∨ú bô-ví			11 12 13 14	bó-fě bó-mů bó-jľ bó-bvů
4	bô-yí	9	bố−y l̃		
5	bố~jĬ	10	bô-yî	15	bó-mŏ

The meaning of these forms is 'they including it/them' or 'with it/them'. Note that the class 6 pronoun is Eyó except in subject position. Also, the floating L tone which occurs between bó and the following pronoun is assigned to the pronoun if it is H tone and beginning with a consonant other than w or y. In the latter case, the L is deleted if the pronoun has L tone, or it is assigned to bố if the pronoun has H tone. Note that the pronunciation bố-bố, bố-jǐ, etc. is also possible, where the L is realized on both components.

It should be noted in all of the above discussion on logophorics that the logophoric pronouns are used only in reported speech (and related constructions). There is otherwise no reflexive pronoun. Instead, either the noun 'body' or 'head' is used:

wvù yɛné ye yé lé 'he sees himself' (lit. he sees his body) bɔ́ kee bikoo bibɔ́ɔ́lɛ́ 'they know themselves' (lit. their heads)

Occasionally a verb will allow the verb extension -En with a reciprocal or reflexive meaning, e.g. bź yEnén é ið 'they see each other/themselves' (see Chapter 9).

Possessives

1. In Noni the possessor always follows the possessed noun. The following noun class concords are observed when the possessor is a noun:

class	concord	example	gloss
1	(wù)	ntòn (wù) wàn	'the child's pot'
2	31cd	bàntàn bo wan é	'the child's pots'
3	(wu)	gwວ໌ວ໌ກ (wu) wan	'the child's bamboo'
	(yi)	gśśn (yi) wan	'the child's bamboos'
5	jilέ	temé ji wan é	'the child's axe'
6	3	εtem ε wan	'the child's axes'
7	keΙέ	kebó ke wan é	'the child's hand'
	bilέ	bibó bi wan é	'the child's hands'
9	(yì)	bìe (γì) wan	'the child's goat'
	(yi)	bie (yi) wan	'the child's goats'
11	feΙέ	feko fe wan É	'the child's cup'
12	muΙέ	muŋkႆm mu wan ຢ	'the child's cups'
	jil€	jinfoon ji wan έ	'the child's leaves'
	bvu[€		'the child's place'
15	m⊃lé		'the child's places'
16	fo .		on the child's head'
17	jo		'in the child's mouth'
18	dvu	yěw d∨u wan	'in the child's house'

A number of observations can be made from the above examples:

- (1) The possessive markers of classes 1, 3, 4, 9 and 10 (which have either the shape [wu] or [yi]) are optionally deleted. Thus, one says either nton wù wan or nton wan 'the child's pot'.
- (2) Classes 2, 5, 7, 8, 11, 12, 13, 14 and 15 all involve a second possessive marker | £, which directly follows the possessor noun. As seen in the above examples, the | of this marker falls when the possessor noun ends in a consonant. Thus, compare bonton bo wan £ 'the child's pots' with bonton bo we | £ 'the person's pots'. In this second example, we not only observe an | in the second possessive marker, but note also that its tone has become LH, in accordance with the tone rule discussed in Chapter 2.
- (3) Concerning the locative classes 16, 17 and 18, it is often possible to have an alternate possessive construction which, instead of the markers fo, jo and dvu, employs a suffix lé, e.g. fòjem fo wan or fòjem wan é 'at the child's grave'. As seen in the latter example, the l of lé also falls when it is directly preceded by a consonant. This second construction is the only one possible when a locative does not transfer a noun into one of the locative classes, e.g. è fetelè fe wan é 'in the child's heart'.
- 2. Possessive pronouns also follow the possessed noun in Noni. A complete table is given on page 20. Besides distinguishing first, second and third person, singular and plural, Noni differentiates between coreferential 'his/her' vs. non-coreferential 'his/her' (subscripted 1 and 2, respectively),

	m	your sg.	his/her ₂	his/her ₁	TOG Sg.	our	your pl.	their	LOG p1.
-	wèm	ÇM	, 0 M	-	wen	wesen	wene	, (ψψ) b	bòwen
2	pei	™ oq	рêм	bêŋ	bowené	bòsésèn	bònên	\$1,ccqcq	bobòwené
3	wêm	*C*	*0	1	wen	wésén	wene	çq(nm)	рэмеп
4	yém	,cy	, è	1	wen	yésén	yéné	çq(1))	рэмел
S	dyem	₩O/p	dyêw	dyêŋ	jiwené	jìsésèn	jìnên	jib531£	jibòwenê
9	eyém	çka	εyé	-	Ewen	yżsésèn	yɔnên	eb5/yob5	ерэмеп
7	Kem	(oy	kêw	kêŋ	kewenć	kèsésèn	kènên	kebáálé	kebòwené
∞ .	byem	₩o∧q	byêw	byên	biwené	bìsésèn	blnên	5155516	bibòwenê
6	yèm	ć	ye	!	wen	yèsèn	yènè	(v)) bš	bòwen
10	yém	çÁ	yé	:	wen	yésén	yéné	(yl) b3	bòwen
11	fyem	₩ov +	fyêw	fyên	fiwené	flsésèn	flnên	fib331£	fibòwenê
12	mweil	*OE	шwêw	mwêŋ	muwen£	mùsésèn	mùnên	∄1°Ççqnш	эмерумен
13	dyem	₩ O/VÞ	dyêw	dyên	jiwené	jìsésèn	jìnên	j1b551£	jibòwené
14	Dwell pwell	<u>fo</u> q	bwêw	bwêŋ	bvuwenέ	bvùsésèn	bvùnên	51ccduvd	bvubòwené
. 15	mem	M O <u>m</u>	mêw	mêŋ	mowené	mòsésèn	mònên	3155dcm	mobòwené
16	fo me	ćw c	им с			to bèsèn	fo bèn	cd ch	
17	jo me	Ćw cį	ÚVW CĘ			jo bèsèn	jə bèn	çq cí	
18	dvu me	ćw u∨b e	ùvw u∨b ć	•=		dvu bèsèn	dvu bèn	çq nvb	

but only in classes 2, 5, 7, 8, 11, 12, 13, 14, and 15 (N.B. the same classes that take 16 when the possessor is a noum). This difference is illustrated in the following two sentences:

wvù nóò dié bvulem bwên 'he has eaten his [own] fufu' wvù nóò dié bvulem bwêw 'he has eaten his [s.o. else's] fufu'

This distinction is not found in the third person plural pronoun, where the corresponding sentence bố nổờ die byuism byubốốis means 'they have eaten their [own or s.o. else's] fufu'.

Noni also has special reported speech (or logophoric) pronouns which are used whenever the possessor is the person reporting some event. We therefore can distinguish the following:

wvù dòó le me nóò ndié bvulem bwêw 'he says I ate his [s.o. else's] fufu' wvù dòó le me nóò ndié bvulem bvuwené 'he says I ate his [=own] fufu'

In the second example, the logophoric possessive pronoun clearly indicates that the subject doing the reporting is also the possessor of the fufu.

The following observations can be made from the table on page 20:

- (1) Tonal differences divide the noun classes into three groups (excluding the locative classes—see below): (a) classes 1 and 9, which generally begin with L tone and do not involve any suffix; (b) classes 3, 4, 6 and 10, which have H tone possessive stems and do not involve a suffix; and (c) classes 2, 5, 7, 8, 11, 12, 13, 14 and 15, which begin either M or H (depending on the pronoun) and appear to involve a L tone suffix in several forms.
- (2) In the 'your sg.' and 'his/her₂' forms, classes 2, 5, 7, 8, 11, 12, 13, 14 and 15 show a -w suffix. It is exactly these classes which exhibit the lé after noun possessors; it is also these classes only which allow the distinction between 'his/her₁' and 'his/her₂'.
- (3) The logophoric pronouns and the pronoun 'their' all seem to be nouns, since they involve the !é postposition (written here as part of the possessive pronoun). Thus, class 2: bowené 'LOG sg.', bobóje 'their', bobòwené 'LOG pl.'.
- (4) The forms observed in locative classes 16, 17 and 18 involve the appropriate class concord followed by the *independent* pronouns, e.g. fo me, jo besen, dvu bó etc. In this respect they are preposition-like.
- (5) Interesting variations are found in the 'our' and 'your pl.' forms. First note that the final n of 'our' is optional, e.g. boom bosesen or boom bosese 'our children'. The meaning of this possessive pronoun is necessarily exclusive (i.e. excluding the hearer). If one wishes to indicate that the children belong to us (meaning, the speaker and the hearer), one uses the exclusive pronoun followed by the inclusive independent pronoun, i.e. boom bosese beene (which can also be simplified to boom boses beene). The root for 'our' appears to be -esen for classes 1, 3, 4, 9 and 10, and -sesen for the remaining classes. Similarly, the root for 'your pl.' appears to be -ene for classes 1, 3, 4, 9 and 10 and -nen for remaining classes. Note that in this case class 6 does not function with 1, 3, 4, 9 and 10. Also, 'your pl.' in classes 1, 3, 4, 9 and 10 appears to have a -e suffix. In the other cases it was classes other than these which had a suffix (e.g. -w) or postposition (16). In classes 2, 6, 7, 8, 11, 12, 13, 14 and 15, there has been a reduplication of the root consonant.

Because of all of the differences noted between classes 1, 3, 4, 6, 9 and 10 vs. classes 2, 7, 8, 11, 12, 13, 14 and 15, we shall refer to the former group as *glide* classes, and the latter as CV classes. In devising this terminology we take note of the w- and y- concord of the former classes, but the CV- concord of the latter (e.g. bo-, ji-, bvu- etc.).

The nouns cee 'father' (sg. & pl.) and bwee/bobwee 'mother/s' have the following irregular possessives:

1 cèm 1 cò 1 cè 1 cee wèsèn 1 cèn 1 cee bó	'my father' 'your sg. father' 'his/her father' 'our father' 'your pl. father' 'their father'	bà í cee bàsésèn bà í cee bànên bà í cee bá	o 'our fathers' 'your pl. fathers' 'their fathers'
bwem bwo' bwe' bwee wèsèn bwee wènè bwee bo	'my mother' 'your sg. mother' 'his/her mother' 'our mother' 'your pl. mother' 'their mother'	bàbwee bàsésèn bàbwee bànên bàbwee bá	'our mothers' 'your pl. mothers' 'their mothers'

Notice that while these exceptional nouns appear to take class 1 concords in the singular and class 2 concords in the plural, the forms for 'their fathers' and 'their mothers' is with bo rather than with bobootie.

Finally, the only way to have a complex possessor pronoun is to first use one of the plural possessive pronouns, followed by a complex independent pronoun e.g.

bònwà bòsésèn beèwvù 'his and my books' (lit. our books, he and I) bònwà bònên bènêwvù 'his and your books' (lit. your pl. books, you and he)

DETERMINERS

1. Among its determiners Noni distinguishes three demonstrative pronouns: 'this/these' (near hearer), 'that/those' (near speaker), and 'that/those' (far from speaker and hearer). The forms encountered with the different noun classes are seen in the following table:

class	'this' (n.s.)	'that' (n.h.)	'that' (far)
1	พงโนโก	же	we'
2	ban	ba	bà ba'
2 3	wvun	₩€	we'
4	yin	yε	ye ye
4 5	jin	diε	
6 7	Eyan	εуа	jìjT Èya
	kin	kĺε	kì kī'
8 9	bin	biε	bì bī'
9	y ` [n	yε	ve'
10	yIn	ýε	ye ye
11	fin	fiε	ffT
12	mvun	mwe	mùmū
13	jin	die	Jì JT`
14	bvun	bwe	bvùbvū
15	man	ma	mòma
16	fεn	fê_	fòfe
17	jan	èkē	fèfe
18	dvun	kε	fèfe

Although it is difficult to establish a single underlying form for each of the three demonstrative roots, the following can be said:

- (1) The demonstrative 'this/these' can be represented as /-n/. The vowel that appears before the final -n is, except for class 16, the historical vowel of the class marker and the one found in related languages. Thus, classes 2, 6 and 15 (=PGB 6a) are characterized by an a, although in most other agreement markers the vowel of these markers is realized as o. Class 17 apparently also had an *a historically. Note also that classes 7 and 11 (=PGB 19) now show the vowel i. In other constructions these classes show variation between i and e. The concord markers are L for classes 1 and 9, M for the remaining classes.
- (2) The demonstrative 'that/those' (n.s.) probably reconstructs with the vowel *-a. Except for classes 2, 6 and 15, where the concord marker also had the vowel *-a, the 'near speaker' demonstratives are realized with the vowel ϵ today. There is no tonal difference between classes 1 and 9 and the other classes.
- (3) The demonstrative 'that/those' (far) involves a reduplication process in the classes 2, 5, 7, 8, 11, 12, 13, 14, 15 and the locatives. The tone is ML on the main syllable and L on the preceding reduplicated syllable if present (including the è- of class 6). Again, classes 2, 6 and 15 show a final a, while classes 7 and 11 use | instead of e. Classes 1, 3, 4, 9 and 10 appear to have a -e. Reduplicated forms lack a suffix, while class 6 has the prefix è- and also no suffix.

In the 'near hearer' column, the variant $\dot{y}\dot{y}\ddot{a}$ is frequently heard as an alternative to $\dot{c}\dot{y}\ddot{a}$ in class 6. The only other alternatives to the above forms involve locative demonstratives. First, the form $\dot{k}\ddot{e}$ can be substituted for fèfe in classes 17 and 18 ('far' column). Since the three locative adverbs are fen 'here', fe 'there' (n.h.) and fèfe 'there' ('far'), it is not clear where the forms $\dot{k}\ddot{e}$, $\dot{k}\ddot{e}$ and $\dot{c}\dot{k}\ddot{e}$ come from. Note finally, in this regard, that the demonstratives given above can acquire a locative sense by prefixing a H tone class marker to them identical in segmental shape to the independent class pronouns, as seen below:

'here it is/here they are'	'there it is/there they are' (n.h.)	'there it is/there they are' (far)
พน์พ∨นัก b≾ban	พน์พ่ะ boba	พน์พ ต ๖๖๖ฮ
		wúwe
		y í ye`
jíjín	j (j ε	j [j <u>T</u>
yoʻyan	yáya	yśya
kékĭn		kékT`
		b∫ b <u>T</u>
		у ј у <u>е</u> `
		y (ye)
		féf <u>T</u>
		múm u
្រ័ ្រា		j í jT`_
		bvúbv u
		móma
		f≾f <u>e</u> `
j ójà n		jók <u>e</u>
ďvůdvůn	dvűke¯	d∨úkē`
	they are' wúwvǔn bóbān wúwvūn yíyìn yíyìn yíyìn bíbin yíyìn bíbin yíyìn bíbin múmin jíyìn féfin múmin jíyòn n fófin mómin jíjòbin	they are' they are' (n.h.) wiwvin bobban wiwvin yiyin yiji yiji yoyan kekin bibin bibin yiyin yiyin yiyin yiyin yiyin yiyin yiyin fefin mimun jijin bvibvun moman fofen jojan

In the first column the 'near speaker' demonstratives acquire LH tone in classes 1 and 9 and LM tone in the remaining classes. These compare with LM and M tone, respectively, in the simple demonstratives. The tones of the 'near hearer' demonstrative become LM for all classes, as opposed to the M tone they carry in all but the locative classes of the simple demonstratives. There is no tone change on the demonstratives in the 'far' column.

Both sets of demonstratives given above can be used either alone or after a noun, e.g. $wv\bar{u}n$ 'this one', $w\bar{u}n$ wv $\bar{u}n$ 'this child'.

2. There is no definite article in Noni per se. There are, however, different means available to specify that a referent has already been mentioned or talked about. First, the 'near hearer' forms in the first table can be used in this capacity. Thus, kete kis can either mean 'that tree near you' or 'the tree we have been talking about'. This use of the 'n.h.' demonstratives with a referential sense is common in the languages of this area. It appears to be the case that the longer demonstrative forms can only have the 'near hearer' meaning.

Another means of acquiring a definite reading is to use the form -dvú 'the one being talked about', e.g. feñíi fédvú 'the bird being talked about'. This form takes the same H tone class prefixes as the long demonstrative forms, i.e. wúdvú, bɔơơơ, wúdvú etc. It is not clear what, if any, difference there is in using this form as opposed to the referential use of the 'near hearer' demonstratives.

There is no indefinite article in Noni. However, there are the following forms which are sometimes used for the purpose of non-definite reference:

class	'any, some, certain'	'another, a certain'		
1	mù	w∨ẩmù		
2	ûmcd	êmed		
3	mû	wvumû		
4	m Î	y imî		
5	.j Imû	,i î mê		
6	ŷmû	yəmê		
7	kemû	kemê		
8	bimû	bimê		
9	mì	vímì		
10	m î	vimî		
11	femû	femê		
12	mumû	mumê		
13	jimû	jimê		
14	b∨umû	bvumê		
15	. mamû	êmcm		

In the first column classes 1, 3, 4, 9 and 10 lack a prefix, but appear to have, at least in classes 4, 9 and 10, an -i suffix. (Classes 3, 4 and 10 may also have a H tone prefix which would account for the HL falling tone.) In the second column, a class prefix is used in all classes. While there is some overlap (and possible confusion) between the two sets of forms, they may be differentiated as indicated.

3. The forms for the interrogative determiner - 'a 'which' are as follows:

1	พน์เล่	6	yálá	11	félá
2	bólá	7	kélá	12	mula
3	wűłá	8	bílá	13	jílá
4		9	vílá	14	bvúlá
5	jílá	10	yílá	15	málá

As in the case of the other determiners, these forms are used either after a noun (e.g. mbòn wúlá 'which bushcow?') or alone (e.g. wúlá 'which one?').

7

OTHER MODIFIERS

1. Adjectives. There are very few true adjectives in Noni. As we shall see, most adjectives are derived from verbs. A very few are not, such as the adjective fs 'new' which is presented first after the verb 'to be' in the following examples:

```
'this book is new'
    nwà wvùn nú fwe
                             'these books are new'
    bànwà ban nú bofe
                              'this field is new'
    wế wvun nú fwe
3
                              'these fields are new'
    ηέ yin nú fiε
                              'this axe is new'
    temé jin nú jifε
                              'these axes are new'
    stem syan nú yafs
                              'this bracelet is new'
    kèsăw kin nú kefs
7
                              'these bracelets are new'
    bìsăw bin nú bifε
8
                              'this drum is new'
    ncòm yin nú fìε
9
                              'these drums are new'
    ncom yın nú fie
10
                              'this knife is new'
    fèπò fin nú fefε
11
                              'these knives are new'
    mùກີວັm mun ກນ໌ mumfeْm
12
                              'these guns are new'
    jingúci jin nú jife
13
                              'this place is new'
    byudvúu byun nú byufe
14
                             'these places are new'
    mandvúúm man nú mamfém
```

It is noted in the above forms that the adjective stem carries a M tone, and that there is an obligatory class prefix: a M tone CV prefix for classes 2, 5, 6, 7, 8, 11, 12, 13, 14 and 15; or a modification on the initial consonant in the remaining classes (with an additional initial L tone in classes 1 and 9). Historically, these irregular forms were: * \hat{v}_{-} f ϵ (class 1), *u-f ϵ (class 3), *i-f ϵ (classes 4 and 10) and *i-f ϵ (class 9).

The same forms seen above can be used attributively directly after the modified noun. Alternatively, an additional class marker occurs which creates a relative copular construction:

```
'a book which is new'
    nwà wvů fwè
                         'books which are new'
    sted éd éwnéd
                         'a field which is new'
    wế wvú fwe
3
                         'fields which are new'
4
    ηέ yi fie
                         'an axe which is new'
5
    temé ji jifε
                         'axes which are new'
6
    εtem yɔʻɔfε
                         'a bracelet which is new'
7
    kèsaw ké kefe
                         'bracelets which are new'
    bisăw bi bife
8
                         'a drum which is new'
    ncòm yĭ fìe
9
                         'drums which are new'
    ncom yí fiε
10
                         'a knife which is new'
    fèñò fé fefε
11
                         'knives which are new'
    mພໍກິວໍm mú mumfe
12
    jingúci jí jife
bvudvúu bvú bvufe
                         'guns which are new'
13
                         'a place which is new'
14
                         'places which are new'
    atcm čm můvybncm
```

As seen in class 6, the expected form stem yo yofs 'axes which are new' must

undergo contraction to become ϵtem ysof ϵ . Optional contractions are also possible in the following classes:

```
2 bònwà bósfe 8 bìsảw bíife 13 jingúci jíife
5 temé jíife 11 fèñò féefe 14 bvudvúu bvúufe
7 kèsảw kéefe 12 mùñóm múumfe 15 mondvúúm mósfe
```

Although the modifier 'new' seems to be unto a class of its own, which we might label "adjectives", the class is extremely limited. As seen above, the class markers carry M tone. Such is not the case with another "adjective", meaning 'great', which appears reduplicated when modifying a plural noun, but which takes H tone class markers:

```
mbòn wvúbáy 'a great bushcow' (cl. 1)
bòmbòn bóbáy(báy) 'great bushcows' (cl. 2)
```

A non-reduplicated form can appear with a plural noun, but this indicates that the plural referents are to be taken as a single group, e.g. 'a great group of bushcows'.

The modifier waa 'small' (pl. boom) is derived from the noun wan 'child' and therefore precedes the noun in an associative construction, e.g. waa can 'small monkey' (pl. boom bo can é). Similarly, the noun-like form ngay 'old' combines with we 'person' to form the compound ngay we 'old person' (pl. ngay boni). As seen in the following forms,

```
kwoòn wù ngày wè 'old woman' (1it. woman who is an old person)
bokenè bó ngày boñi) 'old women'
```

this form does not combine directly with other nouns.

The bulk of adjective and adjective-like notions are expressed by verbs in Noni. In the following sentences,

```
me ndèfé lò 'I am tall'
me njìté lò 'I am heavy'
```

the progressive verb forms keé-dèfè 'to be tall' and keé-jìtè 'to be heavy' are used to express what would be predicate adjectives in English. The same verb forms appear in a relative construction to express what would be attributive adjectives in English:

```
wan wvu defe 'a tall child' (lit. a child who talls)
ŋkɛ̃m wvu jìte 'a heavy load' (lit. a load which heavies)
```

Similarly, consider the following expression of color terms by verbs:

```
fèmbì fè ysslé 'a red kolanut' (<-yéé 'become red')
fèmbì fè báá 'a white kolanut' (<-bá 'become white')
fèmbì fè yillé 'a black kolanut' (<-yíí 'become black')
```

Because these are verb forms, when these modify a plural noun, they often take special suffixes, e.g.

```
wan wvu fan 'a fat child' (-fan 'become fat')
boom bo fanken 'fat children' (-fanken (pl.))
```

2. Momerals. The following are the numerals 'one' through 'ten', as they appear in isolation:

man	'1'	soocan	161
fεε	121	sɔɔshwî	171
tεε	131	ñàn	181
nε'	141	bvùùkε	191
tin	151	yoofè .	1101

The numbers '11' through '19' are obtained by placing now before the numerals '1' through '9':

```
ncòw soocan
                                        1161
ncòw mwàn
           1111
                                        1171
                         ncòw sooshwî
ncòw féé
           1121
                                         1181
           1131
                         ncòw mần
ncòw te
                         ncòw byùùkε
                                        1191
           1141
ncòw ne
           1151
ncòw tĩn
```

The 'tens' are formed by placing mbaan before the numerals '2' through '9':

```
1601
            1201
                          mbaansoocan.
mbaanf 16
                                          1701
                          mbaansooshwî
mbaanshe
            1301
                          mbaanñan
                                          1801
            1401
mbaanñe*
                                          1901
                          mbaanbyùùkε
mbaanshen
           1501
```

(The modifications that are observed on the numerals '1', '2', '3', '4' and '5' will be explained below.) In order to add numerals to these 'tens', the same word now is used: mbaanfie now fee '22' etc.

In addition, the larger numbers gwéé '100' (pl. géé) and ntfùkè '1000' round out the system. The follow exemplifies the system with the number '1978' (the year I had hoped this grammar would be finished):

```
ntfùkè géé bvùùke bố mban sooshwî ncòw ñàn '1978'
```

As seen in this form, there is some phonetic variation in pronouncing '70'.

These numerals all appear after the noun they modify. However, there is noun class agreement on the numerals '1' through '7'. First, numeral '1':

```
1 mwan 9 mian
3 mwan 11 femwan
5 Jiman 14 bvumwan
7 kemwan 15 moman
```

As can be seen, 'l' is realized with L tone in classes 1 and 9, ML tone in the remaining singular classes (and the mass class 15). There is some variation among speakers, apparently, although the above seems to be fairly general: the initial consonant will be m- if the prefix is of the shape Ci- or Co-; 'l' will be realized mian in class 9; in all other cases, its form is -mwan (including classes 1 and 3). We have already remarked on the Cw and Ci initial of the adjective fe 'new'. The same explanation holds here and in several of the other numeral forms. Thus, mwan reconstructs as u-man and mian reconstructs as i-man. Apparently the mw- form has spread to other classes unless blocked by the nature of the class prefix.

The	forms	for	the	numerals	121	through	171	are	seen	in	the	following
table:												TOTTOWING

	'two'	'three'	four	'five'	'six'	'seven'
2 4 6 8 10 12 13 15	bofée féé~fíé efáá bifée fíé muféém Jifée moféém	botetè te ~ she eta bitetè she muntentém jitetè montentém	bonew Tena Tena Tena Tena Tena Tena Tena Tena	botTh tTh~sheh stTh bitTh sheh muntTh jitTh montTh	bosościan sościan Esościan bisościan sościan musościan Jisościan mosościan	bosooshwî sooshwî Esooshwî bisooshwî sooshwî musooshwî jisooshwî

Let us consider the above forms by numeral:

(1) The numeral '2' has a root -fee which can undergo both segmental and tonal modifications. In class 10 (and optionally in class 4) it becomes fie, and in class 6 it becomes efac. It acquires a final -m in classes 12 and 15 and apparently a tonal suffix in classes 2, 8 and 13, where its H tone becomes HM. The numeral '2' can optionally be followed by the additional syllable -ten. In this case, everything stays the same, except that classes 2, 8 and 13 show H instead of HM tone:

2	bofééten	10	fléten
4	fééten	12	muféémten
6	εfááten	13	jifééten
8	bifééten	15	moféémten

Unless this additional syllable is related to the numeral '5', it is not clear what its origin is.

- (2) The numeral '3' has a root -te (perhaps pronounced long when enumerating). It undergoes palatalization to become she in class 10 (and optionally in class 4), and acquires an additional suffix with one of two shapes: -tè in classes 2, 8 and 13 vs. -ntém in classes 12 and 15. It is pronounced eta in class 6, with the final -a seen in demonstratives as well.
- (3) The numeral '4' has a root $-n\overline{\epsilon}$ which acquires a final -w in classes 2, 8 and 13 and a -m in classes 12 and 15. The initial consonant is palatalized to \widetilde{n} in class 10, and optionally in class 4. The final vowel becomes a in class 6.
- (4) The numeral '5' has a root -tTh, which palatalizes to sheh in class 10 and optionally in class 4. As seen, the vowel quality also changes.
- (5) The numeral '6' is probably a compound historically. It involves the extremely rare MH tonal contour and is characterized only by a M tone class prefix in classes 2, 6, 8, 11, 13 and 15. The same remarks cover the historical compound involved in the numeral '7'. $-\text{shw}^{\circ}$ varies with $-\text{fw}^{\circ}$.

Numerals higher than '7' remain invariant when modifying nouns of different classes.

It is not easy to construct ordinal numbers in Noni. The following expressions are used to express the notions 'first', 'second' and 'last':

```
ŋwà wvǔ fwě 'the first book' (lit. book of ahead)
ŋwà wvù bǐ 'the second book' (lit. book that follows)
ŋwà wvǔ Èjěm 'the last book' (lit. book of behind/back)
```

Among the quantifiers used in Noni are the invariant forms nteen 'many' and bay 'much' (='great'), e.g.

```
bòmbòn ntèèn 'many bushcows' lemè báy 'much work' ken ntèèn 'many firewoods' meèm báy 'much oil'
```

The quantifier 'whole/all' has the following forms:

1	tfùu	tfulè
2	11ocd	bocilè
3	tfuu	tfulè
4 5	cii	cilè
	jicii	jicilè
6	εcii	εcilè
7	kecii	kecilè
8	bicii	bicilè
9	cii	cilè
10	cii	cilè
11	fecii	fecile
12	munciim	muncilém
13	jicii	jicilè
14	bvucii	bvucilè
15	iiocm	moncilém

When used with a singular class, these forms mean 'whole', e.g. mbòn tfulè 'a whole bushcow'. When used with a plural class, these forms mean 'all', e.g. bòmbòn bocii 'all the bushcows'. It is not clear what the difference between the two sets of forms is. Fr. Kofon has suggested that the forms ending in -lè emphasize the wholeness, i.e. 'in its entirety'.

Finally, the interrogative quantifier 'how many' has the forms below:

2	ašmod nódmód	'how many bushcows?'
4	ken mέε	'how many firewoods?'
6	εtem εmáa	'how many axes?'
8	bite bimέε	'how many trees?'
10	bíé mέε	'how many fish?'
12	muñim muméem	'how many birds?'
13	jiduεὲ jimέε	'how many bridges?'
15	maàmcm máam	'how many oils?'

As seen, the only modifications on the root mes occur in class 6, where the vowel is aa, and in classes 12 and 15, where a final -m is added.

8

WORD ORDER AND SUMMARY

1. With almost no exception, all modifiers follow the noun in Noni:

possessive: bònwà bem 'my books'
demonstrative: bònwà ban 'these books'
adjective: bònwà bofs 'new books'
numeral: bònwà botstè 'three books'

When more than one of the above four types of modifiers co-occur after a noun, the general word order is as follows:

```
NOUN + POSS + DEM + { ADJ / NUM }
```

That is, the possessor is expected to occur first after the noun, followed by any demonstrative and then, in either order, an adjective (such as 'new') and a numeral. These possibilities are illustrated below:

POSS + DEM bànwà bem ban 'these my books' bànwà bem bofe POSS + ADJ 'my new books' POSS + NUM bànwà bem botetè 'my three books' 'these new books' DEM + ADJ bònwà ban bofε DEM + NUM bònwà ban botetè 'these three books' ADJ + NUM bònwà bofe botetè 'three new books' NUM + ADJstated évided évided

POSS + DEM + ADJ : bànwà bem ban bofe 'these my new books' POSS + DEM + NUM : bòŋwà bēm̀ ban bɔtεtè 'these my three books' POSS + ADJ + NUM ; bànwà bēm bofe botetè 'my three new books' POSS + NUM + ADJ : sted étated med éwnéd 11 DEM + ADJ + NUM : bònwà ban bofε botεtè 'these three new books' DEM + NUM + ADJ : sted étated ned éwnéd

POSS + DEM + ADJ + NUM : bằn wà bểm ban bafa batatè 'these my three new books'
POSS + DEM + NUM + ADJ : bằn wà bểm ban batatè bafa " " " " " "

Orders other than the ones shown above are either unacceptable, not quite right, or have a different meaning, e.g. bɔnwà bɔtɛtè ban can mean 'these Book III's', but not 'these three books'.

Both the adjective 'new' and numerals such as 'three' can be extended through an additional H tone class prefix:

bòŋwà bóbofε ~ bòŋwà bóofε
 'the new books'
 bòŋwà bóbotetè ~ bòŋwà bóotetè
 'the three books'

The above forms have been translated with a definite article, which is the likely interpretation of extended adjectives and numerals. However, as seen in the following forms, the additional H tone class prefix appears to have an interpretation as a reduced relative clause:

bònwà bóbotetè bóbofe 'the three books which are new' bònwà bóbofe bóbotetè 'the new books which are three'

The second example might also be translated 'the new books of which there are three'. As seen in the following pair of sentences,

bònwà bóbofs botstè 'three new books' bònwà botstè bóbofs 'three new books'

a combination of this extension with word order works to provide different emphases: when the adjective is extended and the numeral unextended, emphasis is on the modifier which directly follows the noun (just as the focus position in a sentence falls directly after the verb; see Chapter 22). However, a slightly different result is observed when the numeral is extended and the adjective unextended:

bònwà bóbotetè bofe 'the three new books' (taken together) bònwà bofe bóbotetè 'three of the new books'

The second example might be interpreted as 'new books, of them three'.

The interaction of word order and such extended forms is quite complex and deserves further study. In some cases the presence of the additional H tone class prefix makes an otherwise unacceptable sequence grammatical, e.g.

bວິກູwà bem bobsfs ban 'these my new books' *bວິກູwà bem bofs ban

While such extended forms may be regarded as reduced relative clauses, full relative clauses do occur with the verb nu 'be', e.g.

bònwà boo nú bofε 'books which are new'

The difference between this phrase and bonwa bobofε is slight, if any.

The remaining modifiers such as ntèèn 'many', bomê 'other' (class 2) and bocii 'all' (class 2) occur in the ADJ/NUM slots, e.g.

bònwà bofε ntèèn 'many new books' bònwà ntèèn bofε " " "

The interrogative -mee 'how many' appears to fall in this same slot:

bànwà bafe bamée 'how many new books?' bànwà bamée bafe """""""

In certain cases the extended forms are preferred when combined with these modifiers, e.g. bonwa bobofs bomes. Because the extended adjective, numeral and other modifiers alter the word order properties of the noun phrase, the above statements should be taken as rough generalizations only.

2. In the preceding chapters we have seen that almost all noun modifiers take noun class agreements, but that these agreements can vary considerably according to whether the modifier is a possessive pronoun, demonstrative, etc. Even within the same category there is considerable variation, e.g. a tonal distinction between classes 1 and 9 and the remaining classes in the 'n.s.' demonstrative, but not in the forms for the 'n.h.' and 'far' demonstratives. A table is given on the following page which is designed to summarize the different kind of class concords found on the various modifiers.

Cl.	N. pref.	PRO	Poss. N	'his'	'this'	'another'	'new'	11'/'4'
1	`w(N)-	wvù	(wù-)	w' -	wvù-	w∨úù	`w_	`w_
2	bo-	bś	31cd	b'w	ba-	bo~	bo-	wcd
3	w_	wvú	(wu-)	w'-	wvu-	w∨uù	w_	w_
4	Ø	yſ	(yi-)	y'-	y 1 –	yi1	У_	(Y-)
5	Øe	jí	ji- −1€	dy '₩	ji-	j!-	j!-	ji− -ŵ
6	ε-	yś	€-	E-3	εya-	yo-	yo-	€a
7	ke/ε-	ké	ke1€	k ′- -₩	ki-	ke-	ke-	ke₩
8	bi-	Ьí	bi− −l€	by'w	bi-	bi-	bi-	bi− -ẁ
9	`(N)-	γÌ	(y ì -)	y ` -	yì-	yf1	` y_	`Y-
10	'(N)-	уľ	(yi-)	y ' -	y 1 –	yiì	У_	У_
11	fe-	fé	fe1€	fy'w	fi-	fe-	fe-	few-
12	munm	mű	- mu− −1€	mw ∕ẁ	mvu-	mu-	mumm	mu− −m̀
13	Ji-	jľ	ji− -1€	dy'₩	jI-	Ji-	ji-	j i − -₩
14	bvu-	b∨ů	bvulé	bvw	bvu-	bvu-	bvu-	b∨u ^w -
15	m(ис) m	mś	31cm	m'ẁ	ma-	-cm	mmcm	m⊃m̀
16	fò-	fś	fo-	fo-	fε-			
17	(-3)	jś	jo-	jo-	ja-			
18	Ø	dvú	d∨u-	dvu-	dvu-			

In the above table the symbol $^{\rm W}$ stands for labialization of the initial consonant of the noun, while the symbol $^{\rm Y}$ marks the palatalization of the initial consonant.

PART III: VERB STRUCTURE

9

VERB FORMS

1. Like other Bantu and Bantoid languages, the verb in Noni consists of a verb radical and one, or sometimes more suffixes. The verb radical (R) is almost always monosyllabic and can have any of the following syllable structures:

CV : CVV : CVC :	bi cii kfun	'follow' 'drag' 'hit'	tố kốố tổn	'come' 'catch' 'refuse'
CGV :	gwe	'fall'	kwé	'die'
CGVV:	kwaa kwaw	'grate' 'think'	kwển	'be sufficient'

As also seen, there is a tonal contrast in verb radicals. Verb forms will be cited with the tones they carry after the infinitive prefix kèé. Thus, monosyllabic verb radicals are seen to take either a M or a H tone in the infinitive construction, e.g. kèé-bi 'to follow', kèé-tś 'to come', etc.

Verb suffixes can be one of two types: (a) lexical or (b) derived. Lexical verb suffixes are those which occur on a verb form but do not constitute a separate, meaningful part. Examples are cok-& 'bathe', fum-& 'bake in ashes' and thu-se 'give back'. While the -& suffix in the first two examples does not have a meaning of its own, and while the -& suffix in 'give back' does not appear to be related to the causative suffix of the same shape, these endings still function as suffixes (e.g. in progressive aspect formation; see Chapter 10). Perhaps they once did have a consistent meaning, e.g. fum 'bury' may be the same morpheme historically as the first syllable of fum-& 'bake in ashes'. Similarly, cow 'soak, dip' may be related to cok-& 'bathe' etc.

Derived suffixes can be added with a change of meaning. Thus, the suffix -cs adds the meaning 'here and there' or plural action: lo 'seize' vs. lo-cš 'seize from several people'; gaw 'divide' vs. gaw-cš 'divide in several places'. Other derivational suffixes are given in the remaining sections of this chapter. In the following chapter we shall examine in detail the progressive aspect suffix which changes the aspect of a verb, e.g. k55 'catch' vs. koo-lè 'be catching'.

We shall see that a verb form may have more than one suffix, although this is not common. Where this most frequently arises is in the derivation of the progressive aspect (Chapter 10). We shall adopt the following terminology: the initial syllable of the verb will be termed the radical (R). The radical plus any unanalyzable second syllable will be termed a base. The base followed by any additional suffix(es) is simply referred to as the verb form. In studies of Bantu languages verb suffixes are sometimes referred to as "extensions". We shall occasionally use this term to refer to non-lexical verb suffixes. The different verb extensions that have been discovered in Noni are presented and exemplified in the following paragraphs. It is important

to note that while numerous examples have been found of verbs taking each of the suffixes to be discussed, one cannot predict with certainty which verbs will take which suffixes. Thus, each radical will have to be marked in the lexicon for the suffixes it accepts.

2. Aspectual extensions. The first group of verb suffixes has five members, all of which are aspectual in nature (cf. the progressive suffix in the next chapter). These are as follows:

```
-cs 'attenuative' -tsn 'bifurcative'

-ys 'distributive' -RED 'frequentative'

-ksn 'iterative'
```

In all but the frequentative, the suffix has the shape -CV or -CVn. As will be seen, the frequentative is formed by a reduplication process. The reduplicated syllable is, however, considered a suffix.

2.1. Attenuative -cc. Radicals which take the -cc suffix divide into two classes semantically. The first class shows that -cc changes the meaning of the verb in such a way that the extent of the action is weakened or attenuated. Examples are as follows:

```
kám
       'squeeze'
                               kámcέ
                                       'squeeze a little'
 nám
       'pinch'
                             námcé
                                       'pinch a little'
 kfun
      'hit'
                              kfunce
                                       'hit a little'
 COW
       'soak'
                               COWCE
                                       'soak a little'
ghẩw
       'be superior'
                              gháwce
                                       'be slightly superior'
 fan
      'become fat'
                               fancε
                                       'become a little fat'
```

The suffixed forms can be translated 'a little', 'slightly', 'lightly' etc. In some cases the meaning 'a little from' or 'a little out of' is acquired:

```
séé
      'cut with knife'
                              séécé
                                       'cut a little out of'
saa
      'tear'
                              saàcé
                                       'tear a little out of'
WCP
      'grind'
                              35wcg
                                       'grind a little out of'
díé
      'eat'
                              diéce
                                       'eat a little from'
cźn
      'steal'
                              cáncé
                                      'steal a little out of'
```

In several cases the attenuative suffix indicates that a "smaller job" is involved. This sometimes necessitates a different gloss for the suffixed form:

```
yow
      'skin' (an animal)
                              YOWCE .
                                      'peel' (a banana etc.)
      'spread to dry'
san
                                      'turn on the second side to dry'
                              sance
tóm
      'embroider' (<'sew')
                              tómcέ
                                      'mend' (='re-sew')
bám
      'mould'
                              bśmcε
                                      're-mould' (after an accident or crack)
      'swallow' (liquid)
m i
                              mTce
                                      'swallow' (something solid)
```

The first example best exemplifies this change from a bigger to a smaller job. Skinning an animal is a similar task to peeling a fruit, but the proportions are much greater. The next three examples indicate a second application of a job. Because it is a re-doing, as it were, the job*is by definition smaller (e.g. the drying of the second side of an object is a smaller job than the drying of the first side; the moulding of a pot the first time is a larger job than the re-moulding that takes place in a repair, etc.). The last example is intended to indicate that swallowing a liquid is a big (continuous) job, while

swallowing or gulping a solid object is a smaller (singular) job.

Approximately 70% of the attenuative verb forms attested have this meaning 'a little' etc. The remaining 30% have the meaning 'here and there' or 'little by little', both of which indicate plural action. This interpretation derives from the attenuative meaning: instead of performing an action all at once as a single event, one attenuates the action into a sequence of smaller events. This is seen in the following examples:

```
'defecate several times'
                                ñTcč
      'defecate'
ñ١
                                shoce 'clean one after the other'
sho
      'clean'
                                       'touch several times'
                                k<del>o</del>mcš
      'touch'
kom
                                tance
                                       'count one after the other'
      'count'
tán
                                       'trim several things'
                                kfűcε
      'trim'
kfúv
                                       'spit several times'
                                túcε
      'spit'
túy
                                qawce 'div. in several places, share w/sev. people'
      'divide, share'
gaw
```

The one verb son 'slap' has been found with both meanings. Thus, sonce means either 'slap a little/lightly' or 'slap several times lightly'. We suspect that more of these attenuative verb forms have the meaning 'several times lightly' rather than simply 'several times'.

2.2. Distributive -ye. The distributive suffix -ye indicate that several actions were involved in the event. This can mean doing one action several times (e.g. hitting someone several times) or doing several equivalent actions once [or more] each (e.g. hitting different people):

```
séévé 'cut several times'
     'cut with a knife'
                                     'uproot several times'
                              bááγε
báá
     'uproot'
                              cimyé
                                      'dig several times'
cim
     'dig'
                                      'bury several things'
                              fomγξ
fom
     'bury'
                                      'smoke several pieces of meat'
                              támνέ
     'smoke' (meat)
tám
                                      'slap several times'
     'slap'
                              sonye
son
                                      'curse several times'
                               3výc I
Low
     'curse'
                                     'pay several people'
                               laŵyἔ
law
    'pay'
```

Among the verbs recorded with more than one possible interpretation are:

```
tóm 'shoot' tómyé 'shoot sev. things, sev. times at 1 thing' lóm 'bite' lómyé 'bite sev. times, or sev. people bite' nám 'pinch' námyé 'pinch sev. times or people' 'stab' 'stab sev. times or several people'
```

It is thus possible to get a plural action marked by the distributive suffix with either (a) a plural subject, (b) a plural object, or (c) a singular subject and a singular object, but a plural verb (e.g. 'he stabbed me several times').

Because several actions are involved, the distributive meaning may develop a secondary meaning of completeness. A few such cases have been noted:

```
káw 'chop' káwyế 'chop up completely' saa 'tear' saàyế 'tear completely' (or sev. times) bấŋ 'cover' bấŋyế 'cover up completely' yúúyế 'become clear' (water)
```

2.3. Iterative -ken. The iterative suffix -ken is similar to the distributive suffix in that it is used to indicate plural action. Unlike the distributive suffix, however, it is used almost exclusively with intransitive verbs:

```
kwé
      'die'
                              kwékén 'die one after the other'
awe
      'fall'
                              gweken 'fall several times'
     'belch'
                              bεwken 'belch several times'
bεw
kέy
     'groan' (in pain)
                              kéykén 'groan several times'
jay
     'urinate'
                              jāýkčn 'urinate several times'
nane 'leave'
                              nεĥken 'leave separately'
```

The key to the interpretation of this suffix is in the first and last examples. The intransitive iterative forms to the right can be used with a singular subject to indicate 'several times one after the other', or they can be used with a plural subject to indicate either 'several times one after the other' or, simply, 'separately'. Thus, consider the following sentences:

```
me nóò ngwè 'I have fallen' (several times)
bó kwekéné lò 'they are dying' (separately)
bó kwéé lò 'they are dying' (together)
```

The use of -ken in the second instance of 'I have fallen' indicates that the singular subject has undergone the action more than once. The iterativity of the actions is an automatic consequence of the subject's being singular. Not so in the next two sentences. When -ken is present (in the progressive form, in this case), iterativity is assured: the people involved are dying one by one, i.e. separately. When -ken is not present, however, it must be the case that they are dying as a group, i.e. together. This is more obvious in a sentence such as bố nổô gwè 'they have fallen' (all at once, as a group) vs. bố nổô gwèkèn 'they have fallen' (one by one, separately).

This notion of 'separateness' is revealed in the following iterative forms:

```
fíé 'become burnt' fíékén 'become burnt in separate spots'
kaa 'grind' kaàkén 'grind coarsely' (into separate grains)
jíím 'become extinguished' jímkén 'sev. becoming extinguished on their own'
díé 'eat' díékén 'feed oneself'
```

Finally, note in the following examples that a handful of verbs appear to have the iterative suffix as part of their lexical form:

```
fúmkén 'vomit' niànkén 'spy'
baykén 'thunder' coònkén 'praise'
```

2.4. Bifurcative -ten. The suffix -ten indicates the bifurcation or splitting in two of an object:

```
bón 'fold'
                              bốốntến 'fold in two'
saa
    tear!
                             saàtĚn
                                       'tear in two'
    'divide'
                             gawten
                                       'divide in two'
san 'split'
                             santěn
                                       'split in two'
som 'cut'
                             somtén
                                       'cut in two'
QO
     'break fufu'
                                       'break fufu in two'
                             götén
```

The following verbs have a lexical suffix -ten:

```
cóśntén 'sharpen to a point' sómtén 'settle a dispute'
| léétén 'rest' fomtén 'be sticky' (meat in decomposition)
| jTmtěn 'perspire' (cf. fóm 'to be fatty' [like pork])
```

2.5. Frequentative RED. A dozen verbs or so have been found which derive the meaning of 'continuously', 'often', 'many times' by reduplicating their root syllable:

```
nifmnifm 'wander continuously, a long time'
n.i îm
     'wander about'
      'drip'
                             cémcém
                                       'drip continously' (e.g. rain)
cém
                             Tem em 'be slippery, slip several times'
l em
      'slip'
                             cémcém 'be sickly'
cém
     'be sick'
tέy
     'hang up'
                             tévtév 'hang up several things'
iáv 'make a mistake'
                             jáyjáy
                                     'make sev. mistakes, one mistake
                                         several times!
```

There clearly is some overlap between the frequentative in certain cases and the distributive or iterative suffixes, e.g. when expressing the notion of plural action. Since so few verbs allow reduplication of this type, those susceptible to frequentative marking must be so indicated in the lexicon.

3. Relational extensions. The second group of verb suffixes consists of four members, all of which have to do with expressing grammatical relations and transitivity:

```
-se 'causative' -m 'positional'
-ke 'transitive' -en 'reciprocal'
```

The last of these, the reciprocal extension, has a number of different (unpredictable) variants: -en, -nen, -yen and -sen. (Perhaps -ken and -ten are ultimately related, although they are extensions of a very different type in present day Noni.)

3.1. Causative -se. Many, if not most intransitive verbs can take a causative suffix -se to form transitive verbs:

```
'become black'
                              víísé
                                       'blacken'
                              flésé
fíé
      'become burnt'
                                       'make burn'
béé
      'cry'
                               béésé
                                       'make cry'
bέm
      'be scary'
                               bémsé
                                       'scare'
1 Ém
                              lémsé
                                       'injure'
      'become injured'
lon
      'become old'
                              lohsě
                                       'make old'
fan
      'become fat'
                              fāhsě
                                       'fatten'
kwen 'be sufficient'
                              kwénsé
                                       'add to'
I'm 'become extinguished' I'msé
                                       'extinguish' (tr.)
```

Some verb radicals of the shape CV double their vowel when adding the causative suffix, while others do not. There is no way to predict this difference.

```
bi 'accompany' bTsě 'add to'
bí 'fly' bíísé 'make fly'
la 'become lost' lasě 'lose'
bá 'become white' báásé 'whiten'
```

```
cé 'spend the night' césé 'keep sth. overnight' cí 'restrict' cíísé 'pass a law'
```

Very few transitive verbs take the causative suffix, although the following have been found:

```
foo 'borrow' foose 'lend'
sam 'borrow' samse 'lend'
fen 'lock up' fense 'stop s.o. from doing sth.'
dan 'cross' danse 'put sth. across'
```

3.2. Transitive -ke. Approximately 10-12 verbs have been found which use -ke as a transitive suffix:

```
shii
       'sit, come down from'
                              shilke 'bring down from'
Ιέm
       'become injured'
                              lέmké
                                      'injure s.o.' (=|£msé)
sám
       borrow!
                              sámké
                                      'give out' (cf. samsé)
iì im
       'lie down'
                              jīkě
                                      'lay sth. down'
       'stand' (intr.)
I i im
                              éké
                                      'stand sth. up'
ngvůům 'kneel'
                                      'bend, make kneel' (=ngvúsé)
                              ngvűké
yiém
      'lean against'
                             véké
                                      'lean sth. against'
i i im
      'become extinguished, jimké
                                      'extinguish, (=jimsé)
          dark'
                                        darken
```

Most of these forms with -ke have intransitive correlates with the positional extension -m (see next section). Also, as indicated, variants with -se exist for a number of the examples (cf. Chapter 10, where a relationship is shown to exist between nonprogressive -ke and progressive -se).

3.3. Positional -m. The following is a complete list of verbs which have been found with the positional suffix -m:

```
jìím 'lie down' ngyuum 'kneel'
shìim 'hang oneself' ciim 'be firm' ('stand on its own')
liim 'stand' jiim 'get dark, become extinguished'
ñìim 'hide oneself' yiém 'lean against'
```

The vowel preceding this -m is always double, suggesting that -m might once have been pronounced -me (as it is pronounced in nearby languages such as Kom and Babanki). While the simple root does not exist by itself, it is clear that -m adds the meaning of 'positioning oneself'. (The sole exception is 'become extinguished', which does not refer to human beings.) As seen in 3.2, when a non-reflexive meaning is required, these verbs end in -ke.

3.4. Reciprocal $-\epsilon n$, $-n\epsilon n$, $-y\epsilon n$ and $-s\epsilon n$. A number of suffixes ending in $-\epsilon n$ express a reciprocal meaning:

```
ŋam 'pinch'
                              námén 'pinch each other'
kón 'chase'
                              konen 'chase each other'
kon 'like'
                              končn 'like each other'
táá 'look at'
                             tááyén 'look at each other'
kee
    'spy'
                             keèyén 'spy on each other'
kii
     'call'
                             klinen
                                     'call to each other' (=kfiyen)
gay 'greet'
                             gaynen 'greet each other'
```

jáy 'make a mistake' jáysén 'mistake one for the other'

While the above examples demonstrate the reciprocal origins of these variants, the most common meaning they acquire is 'together' and even 'several times' (as in the earlier group of extensions):

```
béé 'cry' bééyén 'cry one after the other'
foo 'borrow' foòyén 'borrow sev. things from sev. people'
few 'measure' fewsén 'measure sev. things together'
fyey 'mix' fyeysen 'mix things together'
```

In fact, $-\epsilon n$ has an iterative meaning when placed after verbs with the positional suffix:

```
jilměn 'lie down (pl.)' nilměn 'hide selves' shilměn 'hang selves' ngyúúmén 'kneel (pl.)' límén 'stand (pl.)' címén 'be firm (pl.)'
```

Without the -En suffix, the bare positional verb would indicate that the plural subject takes its position as a group, rather than severally.

3. The above appear to be the major extensions found in Noni. There are occasional cases of unidentifiable suffixes, and many more cases of suffixes which resemble the above extensions, but which cannot be shown to have an independent meaning. Occasionally a verb will have more than one of the above suffixes. Thus, consider the following two verbs:

```
cé 'spend the night' shé 'spend the day'
césé 'keep overnight' shésé 'keep all day'
céshé 'keep sev. things all day'
overnight'
```

The forms céshé and shéshé derive from cé+sé+yé and shé+sé+yé. Finally, the same radical can sometimes take many different suffixes as seen below:

```
| lem 'to slip, slide' (also = 'to work')
| lemce 'slip a little'
| lemye 'slip in several places'
| lem | lemse 'slip several times, be slippery'
| lemse 'cause to be slippery'
```

PROGRESSIVE ASPECT

- 1. In the preceding chapter it was seen that a verb radical can take a number of suffixes and that these suffixes related either to aspect or to grammatical relations. The most commonly occurring aspectual suffix is the progressive marker. Although it is believed that this marker can best be represented underlyingly as /-te/, it has so many different realizations that it is best to study this suffix by addressing its allomorphs separately. In the following sections it will be demonstrated that the different realizations of the progressive suffix can be predicted on the basis of the phonological and morphological shape of the verb form to which it is added. This verb form may either be simplex (i.e. consisting solely of a monosyllabic radical) or complex (i.e. consisting of a radical and a suffix). These two cases are illustrated in the following sections followed by a discussion of verbs whose progressive forms are irregular.
- 2. Simplex verb forms. In the case of simplex verb forms, which consist solely of a monosyllabic verb radical, the shape of the progressive suffix is determined on the basis of the last segment(s) of the radical.
- 2.1. Radicals ending in a short vowel. If a verb radical ends in a short vowel, the progressive (henceforth, PROG) form is obtained by doubling the vowel, e.g.

bi ge lé kfù fó tó tá	'follow' 'make' 'sink' 'sift' 'itch' 'come' 'shoot'	bìi gèè leè kfùù foò toò taà	'be following' 'be making' 'be sinking' 'be sifting' 'be coming' 'be shooting'
---	---	--	--

As in the last chapter, verb forms are cited with the tones they carry after the infinitive prefix kèé-, e.g. kèé-bi 'to follow', kèé-bì' 'to be following'.

The above forms show the doubling of all seven phonological short or single vowels in Noni. In addition, the few verbs that consist of a sequence of unlike vowels should be noted as not undergoing any change in forming the PROG:

```
fié 'become burnt' fiè 'be becoming burnt' dió 'fight' diò 'be fighting'
```

The reason that the second vowel is not doubled is that this would produce a sequence of three vowels, which is not permitted within a word in Noni.

2.2. Radicals ending in a long vowel. When a verb radical ends in a long (=double) vowel, the PROG is obtained by adding the suffix -|e:

```
cli 'drag' clile 'be dragging'
bee 'cry' bèèle 'be crying'
dɛɛ 'cook' dɛɛ̂le 'be cooking'
túú 'carry (on head)' tuule 'be carrying (on head)'
```

```
soo 'knot' sòòlè 'be knotting' k55 'catch' kɔɔlè 'be catching' táá 'look at' taalè 'be looking at'
```

2.3. Radicals ending in -m. When the verb radical ends in -m, the PROG form is obtained by adding the suffix -te:

```
'be digging'
                                cimtè
cim:
      'dig'
                                cemtè
                                        'be dripping'
cém
      'drip'
                                        'be becoming sick'
                                cemtè
cém
      become sick
                                        'be groaning'
                                dvùmtè
dvum
      'groan'
                                         'be sending'
                                tomtè
tóm
      'send'
                                         'be pouring'
                                dòmtè
      'pour'
mcb
                                        'be chewing'
                                camtè
cám
      'chew'
```

2.4. Radicals ending in -n. When the verb radical ends in -n, the PROG form is obtained by adding the suffix -e:

```
'be dancing'
      'dance'
                                 binè
bin
                                        'be rising'
                                 bènè
      'rise'
ben
                                        'be calling'
                                 bεnè
bển
      'call'
                                        'be hitting'
                                 kfùnè
      'hit'
kfun
                                 tonè
                                         'be refusing'
      'refuse'
tổn
                                        'be burning'
                                 tonè
tốn.
      'burn'
                                         'be lacking'
                                 tànè
      'lack'
tan
```

2.5. Radicals ending in $-\eta$. When the verb radical ends in $-\eta$, the PROG form is obtained by doubling the preceding vowel and by adding the suffix -ke:

```
'be trembling'
                               ciinkė
cin
      'tremble'
                                         'be being inclined'
                               tfuuŋkè
tfún
      'be inclined'
                               koonkè
                                         be chasing
      'chase'
kổŋ
                                         'be roasting'
                               yòònkè
      'roast'
yo ŋ
                                         'be frying'
                               kaanke
      'fry'
kán
```

2.6. Radicals ending in -w. When the verb radical ends in -w, the PROG form is obtained by doubling the preceding vowel, dropping the -w, and adding the suffix -ke:

```
'be blackening'
                                jiikè
jiw
     'blacken'
                                fèèkè
                                        'be measuring'
     'measure'
few
                                Ιεεκè
                                       'be dodging'
lέw
     'dodge'
                                       'be soaking'
                                còòkè
     'soak'
COW
                                gàà kè
                                        'be grinding'
    'grind'
wcp
                                       'be paying'
                                Tààkè
     'pay'
law
```

2.7. Radicals ending in -y. When the verb radical ends in -y, the PROG form is obtained by adding the suffix -te:

```
'be coughing'
                                keytè
      'cough'
kéy
                                         'be protecting'
                                cèytè
      'protect'
CEY
                                         'be trimming'
                                kfuytè
kfűy
      'trim'
                                boytè
                                         'be exiting'
      'exit'
bóy
                                         'be accusing'
                                toytè
toy.
      'accuse'
                                kàytè
                                         'be tying'
      'tie'
kay
```

- 3. Complex verb forms. A complex verb form consists of a verb radical and a suffix. In all cases the resulting form is bisyllabic. In a few cases longer forms have been noted. As seen in the following sections, a slightly different set of progressive formation rules apply when the verb form has a suffix.
- 3.1. Verb forms ending in $+(C) \in n$. When the verb form ends in a suffix of the shape $-\epsilon n$, $-\gamma \in n$, $-n\epsilon n$, $-\epsilon n$, $-\epsilon n$, $-\epsilon n$, $-\epsilon n$, i.e. in a suffix ending in -n, the PROG form is obtained by adding the suffix $-\epsilon$. In this case the same PROG suffix is found as in simplex verb forms ending in -n:

```
d∨ū'nĕn
        'rumble'
                              dvùmkènè 'be rumbling'
fúmkén
        'vomit'
                              fumkènè
                                       'be vomiting'
¡Thten 'perspire'
                              iìmtènè
                                       'be perspiring'
        'be quick'
vānsčn
                              yànsènè
                                       'be being quick'
kΤcἔn
        'contradict'
                              kicinè
                                       'be contradicting'
táávén 'look at e.o.'
                              taayine 'be looking at each other'
```

It will be noted that in addition to the suffix -e, the PROG forms involve a vowel change in suffixes ending in -n: non-PROG +(C) ϵ n corresponds to PROG +(C)ene. In the last two examples the PROG forms end in +cine and +yine because the consonants c and y are palatal.

3.2. Verb forms ending in $+(C)\epsilon$. When the verb form ends in $+(C)\epsilon$, and the optional C is not palatal, the PROG form is obtained by replacing the final $-\epsilon$ with $-\epsilon$:

```
tele
        'burst'
                                tèlè
                                         'be bursting!
kenkε
        'get'
                                         'be getting'
                                kenkè
jitě
        'become heavy'
                                jìtè
                                         'be becoming heavy'
fume
        'bake in ashes'
                                fume
                                         'be baking in ashes'
shīwsč
        'shake'
                                shiwsè
                                         'be shaking'
ŋaŋlĕ
        'crawl'
                                ŋàŋlè
                                         'be crawling'
Ιéwtέ
        'run'
                                lewte
                                         be running
```

3.3. Verb forms ending in $+c\epsilon$ and $+y\epsilon$. When the verb form ends in $+c\epsilon$, and the C is palatal, i.e. the suffix is $+c\epsilon$ or $+y\epsilon$, the PROG form is derived by changing these suffixes to $+c\epsilon$ and $+y\epsilon$, respectively. An intermediate stage with $+c\epsilon$ and $+y\epsilon$ is, of course, possible, with the ϵ being then raised to 1 because of the preceding palatal consonant.

```
tómcé
        'mend'
                                tomel
                                         'be mending'
IEnce
        'trample'
                                lènci
                                         'be trampling'
kwawce
        'remember'
                                kwàwci
                                        'be remembering'
kόcέ
        'wrap up'
                                koc i
                                         'be wrapping up'
nεcε
                                nèci
        'straighten'
                                         'be straightening'
káwνέ
        'chop up'
                                kawyi
                                         'be chopping up'
                                kwaay i
kwaàyĚ
        'scrape ground'
                                         'be scraping ground'
saayĚ
        'tear sev. times'
                                sààyì
                                         'be tearing several times'
dáyέ
        'forget'
                                daayi
                                         'be forgetting'
```

As seen in the last form, daayl 'be forgetting', verb forms of the shape CV-ye normally lengthen the vowel of the radical as -ye changes to -yi in the PROG.

3.4. Verb forms ending in +se. When a verb radical of the shape CV has a suffix +se, the vowel of the radical is lengthened in forming the PROG. If the radical already has a long vowel, or if the radical is of the shape CVC, the PROG aspect is marked solely by a tonal change, as indicated below:

```
vèèsè
                                         'be sweeping'
yεse
        'sweep'
                               tfuusè
                                         'be answering'
tfúsé
        'answer'
                               ngvuusè
                                         'be bending sth.'
        'bend sth.'
ngvúsé
                               taasè
                                         'be sewing'
tásé
        1 sew1
lέwsé
        'divert'
                               lεwsè
                                         'be diverting'
                               dòmsè
                                         'be prodding'
damsě
        'prod'
sámsé
        'lend'
                               samsè
                                         'be lending'
                               fààsè
fəàsě
        'lend'
                                         'be lending'
```

3.5. Verb forms ending in -fe. While it is not clear whether -fe should be considered a suffix, or whether the f belongs to the radical, verb forms which end in -fe form the PROG aspect by lengthening the radical vowel, dropping the f, and adding -te:

```
hífé
                              biitè
                                      'be asking'
       'ask'
                                      'be spoiling'
béfé
       'spoil'
                             beetè
téfé
       'advise'
                             teetè
                                      'be advising'
kfúfé 'become short'
                                      'be becoming short'
                             kfuutè
cófé
       'chisel'
                             cootè
                                      'be chiseling'
                                      'be throwing down'
tófé
     'throw down'
                             tootè
```

3.5. Verb forms ending in -e. When a verb form ends in -e (and the preceding consonant is not s or f), the PROG form is obtained by replacing the -e suffix with -y1:

```
kéwé
       'roll up'
                                kewy i
                                       'be rolling up'
jέwé
       'plant'
                               jewyi
                                       'be planting'
                                       'be speaking'
jēmě
       'speak'
                               jèmγì
                               yèny i
                                       'be opening'
veně
       'open'
koné
       'reward'
                                       'be rewarding'
                               konyì
iThě
       'look at'
                               jìnyì
                                       'be looking at'
```

3.6. Verb forms ending in -VVm. Verb forms ending in -VVm, i.e. those taking the positional extension +m, form the PROG by adding -e. This differs noticeably from the -te suffix which was assigned to simplex verbs ending in -m.

```
ji Îm
                                         'be lying down'
        'lie down'
                               ilime
shiim
        'hang oneself'
                               shilmè
                                         'be hanging oneself'
                               liime
                                         'be standing'
I I Im
        'stand'
ñiim
        'hide oneself'
                               ñììmè
                                         'be hiding oneself'
                               ngvuume 'be kneeling'
navúúm 'kneel'
```

3.7. Reduplicated verb forms. Most of the attested reduplicated verb forms have a radical of the shape CVC. They all take a suffix -e to form the PROG aspect, rather than following the pattern established for the different CVC shapes in simplex verb forms:

```
gemgem 'fence in'
                             gemgeme 'be fencing in'
cémcém 'be sickly'
                             cemcèmè
                                      'be being sickly'
léwléw
       'dodge'
                             lewlèwè
                                      'be dodging'
lāwlāw
       'twinkle'
                                      'be twinkling'
                             làwlàwè
cowcow 'dampen'
                             còwcòwè
                                      'be dampening'
```

If the same rules had been followed as on simplex verb forms, the first and last PROG forms above would have been *gèmgèmtè and *còwcookè. As indicated, these are incorrect. The one reduplicated form that has been found that does not have a CVC radical is tuttu 'carry(child on shoulder)', whose PROG form is tuutùùlè, i.e. taking the suffix -le as CVV radicals normally do.

4. Exceptions. Despite the fact that the above statements cover the majority of Noni verbs, there are exceptions in the way in which certain verbs form the PROG. These exceptional verbs may (as indicated) optionally follow the appropriate rule, or in other cases, must be exceptions. The following table gives an idea of the relative numbers of regular vs. irregular verbs according to their phonological and morphological structure:

Verb Structure	# Regular	# Irregular
CV	39	10
CVV	30	5
CVm	32	.0
CVn	22	1
$CV_{\mathfrak{I}}$	19	0
CVw	29	5
CVy	32	1
+(C)εn	28	0
+(C)ε	14	5
+ce	26	. 0
+yε	6	1
+se	27	3
-fe	9	2
+e	17	9
+m	8	0
RED	9	0
	347	42

Thus, out of a corpus of 389 verbs, only 42 are irregular, i.e. approx. 10%. These exceptional verbs are discussed in the following sections.

4.1. Radicals ending in a short vowel. There are nine verbs whose radical ends in a short vowel which can optionally take a -le suffix to form the PROG. Alternatively they can undergo the regular rule and lengthen their vowel:

```
bί
     'flv'
                               bil/bilè
                                           'be flying'
cí
     'legislate'
                              ciì/cilè
                                           'be legislating'
fε
     'cure an invalid'
                               fèè/fèlè
                                           'be curing an invalid'
fe
     'chat'
                              fèè/fèlè
                                           'be chatting'
bvu 'interrupt'
                              bvùù/bvùlè
                                           'be interrupting'
     'sift'
kfu
                              kfùù/kfùlè
                                           'be sifting'
fo
     'pour'
                              fòò/fòlè
                                           'be pouring'
kó
     'gather up'
                              koò/kolè
                                           'be gathering up'
ka
     'clear one's throat'
                              kàà/kàlè
                                           'be clearing one's throat'
```

Perhaps one can add the verb 'eat' here, which has a vowel sequence, but which fails to remain the same in the PROG:

```
dié 'eat' diekènè 'be eating' (intr.)
```

As seen, the suffix -ken is added to form the PROG. By itself, dieken 'means' feed oneself'. A PROG form die(e) has been recorded with an object present.

4.2. Radicals ending in a long vowel. There are two small groups of verbs that end in VV, but which do not take -|e. In the first, two verbs take -te instead:

```
b55 'bear a child' bootè 'be bearing a child' y55 'build' yotè 'be building'
```

It is significant that both of these verbs are irregular in the [-focus] past tenses, where their forms are boke and yoke, respectively (see Chapter 12).

The second group of irregular CVV verbs consists of three verbs ending in -ee which exhibit an η followed by a -e suffix in the PROG aspect:

béé	'accept'		'be accepting'
séé	'tell'		'be telling'
téé	'call'	tenè	'be'calling'

These verbs are also irregular in both [+focus] and [-focus] past tenses. All of the exceptions to the normal addition of PROG -le to CVV verbs have to do with a historically lost velar which reappears (cf. class 5 in nouns in Chapter 3, section 4.)

4.3. Radicals ending in -n. One verb ending in -n has been found to require an additional -ten suffix (=our 'bifurcative' extension) in the PROG:

```
bón 'fold' bontènè 'be folding'
```

The verb form bonten means 'fold in two'. What is curious about this verb is that the suffixed form is required in the PROG but not in the non-PROG aspect.

4.4. Radicals ending in -w. Eight verbs ending in -w have been found which either require or at least allow a PROG form obtained by adding -te directly to the radical, as indicated:

```
'be poisoning'
     'poison'
                               iiwtè
.i (w
                                            'be being mad, drunk'
                              jewtè
jέw
     'be mad, drunk'
                                            'be leaking'
                               jowtè
    'leak'
jów
                                            'be making a click'
                               cowtè/cookè
     'make a click'
CÓW
                              sòwtè
                                            'be investigating'
     'investigate'
WCZ
                               cawtè
                                            'be picking out'
cáw
     'pick out'
                                            'be dividing'
                               gàwtè/gààkè
     'divide'
gaw
                              tawtè/taakè
                                            'be becoming strong'
taw 'become strong'
```

The difference in PROG forms between the regular and irregular radicals ending in -w may be due completely or in part to the fact that many of the irregular verbs had final labial consonants (e.g. -b in 'divide' and 'become strong'), while the regular verbs had a velar. The etymologies for many of these roots are not clear.

4.5. Radicals ending in -y. One verb ending in -y exceptionally loses the -y and takes a -le suffix in the PROG form:

```
féy 'pass, surpass' felè 'be passing, surpassing'
```

Because this verb takes exceptional tones in several tenses, it may be analyzable as fei, in which case it becomes exceptional to a different rule.

4.6 Verb forms ending in -(C) ϵ (where C*palatal). Several verbs end in -(C) ϵ but do not convert this to -(C) ϵ , as seen earlier. First, there are three verbs which replace - ϵ with -y:

```
ciné 'leave (sth.)' ciny' 'be leaving (sth.)'
nEné 'leave' nêny' 'be leaving'
fúlé 'blow' fuy' 'be blowing'
```

As seen in 'be blowing', I drops out before y. Two more irregular verbs ending in $-\epsilon$ are seen below:

```
deft 'become long' deete/defene 'be becoming long' shume 'carry on pole' shume 'be carrying on pole'
```

The first verb can optionally follow the rule for verbs ending in -fe or can convert itself to defen and then become defene. The second verb drops its - ϵ and then takes -te, as radicals with final -m do by rule.

4.7. Verb forms ending in $-y\epsilon$. One verb ends in $-y\epsilon$ but drops the $-\epsilon$ and adds -te (as in 'carry on pole' in the last group):

```
bνύγε 'burst open' bνυγτε 'be bursting open'
```

Perhaps this indicates that the verb should be analyzed as bvúy-é rather than bvú-yé. The - ϵ then drops exceptionally, causing the verb to take the -te suffix in the PROG.

4.8. Verb forms ending in -se. The following two verbs end in -se, but do not follow the general pattern seen earlier:

```
dósé 'be tired' dosè 'be being tired' cinsé 'make round & smooth' cinsènè 'be making round and smooth'
```

In 'be tired' the vowel o should double in the PROG, but doesn't. In 'make round and smooth', an -n is first added and then the PROG -e by general rule.

4.9. Verb forms ending in -fe. Two verbs ending in -fe have been found which instead of undergoing the general pattern (lengthening of radical vowel, dropping of f, adding of -te), lengthen their radical vowel and keep the f:

```
yéfé 'rub' yeefè 'be rubbing' cófé 'take (out) from' coofè 'be taking (out) from'
```

4.10. Verb forms ending in -Ce. There are a number of exceptional verbs ending in -(C)e (where $C \neq s$ or f). One verb ends in -e but does not acquire the expected -y:

```
sémé 'cease' (of rains) semtè 'cease' (of rains)
```

Instead, the -e drops and -te is assigned by general rule. In six other verb forms. -c; is found instead of -y; in the PROG:

qwelě	'harvest'	gwèci	be harvesting
keně	'set fire'	kèènci	'be setting fire'
lentě	'cross (river)'	litaci	'be crossing (river)'
ngélé	'crunch'	ngecl	'be crunching'
nεlě	'straighten out'	nèci	'be straightening out'
kólé	'wrap up'	koci	'be wrapping up'

We have seen that there are two verb extensions -yɛ 'distributive' and -cɛ 'attenuative'. The PROG forms of these extensions are -yi and -ci, respectively. Most verbs which end in -(C)e (other than -se or -fe) are assigned -yɛ, in the PROG, which of course becomes -yi. The above six verbs, however, are assigned -cɛ, which becomes -ci. Note that 'be setting fire' and 'be crossing' have a -nc- sequence which causes the preceding vowel to double (as happened with V η + ke in section 2.5 above). The change from ee to Ii cannot be explained in 'be crossing'. Finally, 'be harvesting', 'be crunching', 'be straightening out' and 'be wrapping up' all lose their I in the PROG forms.

Now note the following two irregular verbs, also having an internal - | -:

```
belé 'thank' bèèlè 'be thanking' belé 'weed, pluck' besyì 'be weeding, plucking'
```

Although these two verbs are identical except for tone, they are different even with respect to their irregularities in the PROG. 'Thank' exceptionally lengthens its radical vowel and keeps -!e in the PROG, while 'weed, pluck' changes -e to -y|, but drops its -!- and lengthens the radical vowel. In this regard it is like 'be blowing' in section 4.6, except that this latter form does not lengthen its radical vowel.

Finally, there are several verbs ending in the 'transitive' extension -ke, whose PROG forms are derived as seen below:

```
bànsè
                                       'be turning (tr.)'
bāhkě
       'turn (tr.)'
                                       'be dispersing (tr.)'
                               sànsè
sankě
       'disperse (tr.)'
                                       'be standing sth.'
                               lewsè
léké
       'stand sth.'
                               càwsè
                                        'be putting away'
cākě
       'put away'
                                        'be laying sth. down'
                               jiwsè
ίΤκě
       'lay sth. down'
                               shìwsè
                                       'be bringing sth. down'
       'bring sth. down'
shî kě
                                       'be leaning sth. against'
                               yewsè
yéké
       'lean sth. against'
```

All of the PROG forms seen above involve a -se suffix, instead of the -seen in the non-PROG forms. If the radical has the structure CVC, -ke is simply dropped. If not, we substitute -se for -e, yielding an intermediate structure CVkse, which then becomes CVwse. It is hard to explain these changes, and for this reason the above group was included among the irregular, rather than regular, verbs. We saw in Chapter 9 that some of the -ke forms have alternatives with -se. Thus, we may have an imperfect merger of two extensions, rather than a natural phonological development.

4.11. Inherently PROG verbs. The rules given above cover the majority of Noni verb forms, with the exceptions being limited in number. In addition, at least two verbs, nù(ù) 'be' and keè/keỳ 'know' are only rarely used, if at

all, when not in the PROG aspect.

5. The rules. It is a very striking feature of Noni that the PROG suffix has so many different realizations. Some of these different realizations have to do with the substitution of one suffix for another in forming the PROG. Thus, as seen in 3.5, the suffix -ye is substituted for the non-PROG suffix -e in PROG formation. To this -ye is added the true PROG suffix, which causes -ye to become first -ye and then -yi, as described. Similarly, in 4.10, we saw that -se is substituted for -ke in forming the PROG aspect.

In this section the different realizations of the PROG suffix are accounted for, informally, by a sequence of ordered rules. We begin with the assumption that the PROG suffix can be represented by the form /+te/. The rules which now follow apply in the order given, but affect only the PROG suffix. Because of this last restriction it is clear that these are really morphologized. Thus, the following rules represent only an abstract account of these phenomena, one which is believed to capture the historical processes which have led to alternations observed.

5.1. Rule 1. The first rule changes +te to +le when preceded by a CVV verb radical:

$$t \rightarrow I / \#CVV+ e$$

5.2. Rule 2.. The second rule deletes the t of the PROG suffix in three different environments:

t
$$\rightarrow$$
 \emptyset / # $\left\{ \begin{array}{c} C \ V \ (n) \\ X + Y \end{array} \right\} + \underline{\hspace{1cm}} e$

In the first line between braces in the environment of rule 2 the t of the PROG suffix is deleted after a verb radical of the shape CV or CVn. In the second line the t is deleted whenever the verb form contains a morpheme (+) boundary (the X and Y variables stand for non-null elements). This internal + boundary can be either a radical-suffix boundary or the boundary which occurs between the two parts of a reduplicated verb. It should be noticed here that because rule 1 precedes rule 2, the reduplicated verb tuitiu 'carry (child on shoulder)' has the PROG form tuutuule, rather than *tuutuue.

5.3. Rule 3. The third rule takes the output CV+e from rule 2 and assimilates the PROG +e to the preceding single vowel of the verb radical:

e
$$\rightarrow$$
 [α F] / # C [α F] + ____

5.4. Rule 4. Rules 4 and 5 are written transformationally. As seen in the following formulation,

the fourth rule affects radicals which end in -n and -w, both of which are

[+back]. When such radicals occur in the PROG aspect, their vowel lengthens, the t of the PROG suffix becomes k (i.e. [+back]), and, if the final radical consonant is [-nasal], i.e. w, it drops. Thus, as we saw in 2.5 and 2.6, respectively, cin 'tremble' becomes PROG clinke, and jiw 'blacken' becomes PROG jiike.

5.5. Rule 5. The fifth rule is similar in its formalization to rule 4. but it cannot be collapsed with it:

When a verb radical ends in -f, the PROG form is obtained by lengthening the preceding vowel and deleting the -f. As seen in this rule, one analysis assumes that verb forms such as bifé 'ask' are actually monosyllabic underlyingly, i.e. /bif/. Because -f cannot occur at the end of a word, -e is introduced in just such cases. Thus, it does not figure in rule 5. If it had to be included, then rule 5 would have to precede the second line of rule 2, because we would not want hypothetical /bifé+te/ to become *bifee. On the other hand, while this form must become beetè 'be asking', the two exceptional forms yéfé 'rub' and cófé 'take (out) from' in 4.9 must become yeefè and coofè in the PROG. Perhaps these have a true +e suffix. In this case, yéfé+te, for example, would first become yefee and then yeefè 'be rubbing'.

5.6. Rule 6. The sixth rule accounts for sequences of vowels which occur in the suffixes, e.g. -cs+e. Depending on how one accounts for the vowel lengthening which sometimes characterizes the verb radical (as a "compensatory process"?), rule 6 can be formalized as in (a) or (b):

(b)
$$\langle aF \rangle \not 0 + CV + e$$

 $\langle aF \rangle \not 0$
 $\langle aF \rangle \not 0$

The formulation in (a) simply says that a (suffix) vowel drops before +e. The affected vowel will automatically be a suffix vowel, because radical vowels have already assimilated a following +e by rule 2. The formulation in (b) is designed to take care of the alternations between tfúsé 'answer' and tfuusè 'be answering' seen in 3.4. Perhaps it is the case that the radical vowel lengthens only if the sequence of suffix vowels is e+e. This would make 'be forgetting' in 3.3 an exception.

5.7. Rule 7. The last rule we shall consider here assimilates +e to i when preceded by a palatal consonant (c or y):

The above rules will generate the regular alternations. In addition, many of the "exceptional" PROG verb forms can be accommodated with only minor adjustments.

VERB CONJUGATION I: INTRODUCTION

1. Tone patterns. In this and the following three chapters Noni verb forms will be described and illustrated as they appear in different tenses, in different constructions, and when negated. We shall refer to these different forms as conjugations, noting that this term covers tense, aspect, mood and polarity (negative vs. affirmative) as these features are realized in main vs. relative vs. consecutive clauses (in Chapters 12, 13 and 14, respectively).

As a preliminary to the study of tense, aspect etc. it is necessary to point out a few features of the tone system as it applies to verb forms. Chapter 2 it was seen that Noni has a basic three-level tone system with L, M and H tone. Various contours are also allowed. It has already been demonstrated (in Chapter 9) that verb radicals show a two-way tonal opposition. In simplex verb forms this opposition has been represented as one between M and H tone, e.g. cow 'dig' vs. cow 'make a click'. In addition to this Zexical tonal contrast, each verb form takes one of four tone patterns. We shall refer to these four patterns as A, B, C and D. Thus, in order to predict the tones of a given verb in a given conjugation, we need to know (a) its lexical tone, and (b) its grammatical tonal class (A, B, C or D). We shall adopt the convention of referring to the class of verbs having the lower lexical tones as L tone verbs, and those having the higher lexical tones as H tone verb. Thus, 'dig' is a L tone verb, while 'make a click' is a H tone verb. The realization of the different verb tones can now be summarized in the following table:

pattern	monosy	llabic	bisyl	labic	formula	
	L	Н	L	Н	(tentative)	
Α	L	LM	L-L	M- L	`R`	
В	LH	Н	L-H	M-H	`R'	
C	M	Н	ML-H	H-H	R	
D	ML	Н	ML-L	HL-L	'R'	

The patterns are numbered in order of frequency, with pattern A being the most common. As can be seen, tentative formulas have also been proposed. Since there are only four patterns, it is reasonable to propose a two-way tonal contrast occurring both pre- and post-radically, as indicated. Thus, pattern A consists of a pre-radical L, the radical tone (either L or H) and a post-radical L, and so forth. These formulas remain tentative as an abstract analysis of the tonal system of verbs, although considerable support is available also from noun tones (though not without problems).

In the above table we have distinguished monosyllabic vs. bisyllabic verb forms. The reason for this is clear: the same pattern will have a different realization according to the number of syllables in the complete verb form (=the verb radical, any lexical or grammatical suffixes, including the PROG suffix, if present). In cases where the verb form has more than two

syllables, the second tone goes on the second and all subsequent syllables.

It will be noted from this table that in three of the patterns (B, C and D), a monosyllabic H tone verb will appear as H tone, but that in the corresponding bisyllabic forms, three distinct tonal patterns are observed (respectively, M-H, H-H and HL-L). This is due to the fact that tonal contours are limited in Noni. In pattern B, the monosyllabic H tone pattern should have been MH, while in pattern D it should have been HL. While HL is only rarely permitted in Noni, MH is virtually unattested in the language. (There is some question as to whether it can occur in the numeral 'six'.) While patterns B, C and D merge on monosyllabic H tone verbs, they are of course distinguished on monosyllabic L tone verbs. All of these patterns will be exemplified in the following sections and chapters.

2. Citation forms: the infinitive. It is customary to establish a citation form for verbs. While in nouns we have simply cited the singular form, the choice for a citation form in verbs is not as obvious. In the preceding chapters we have cited the verb forms with the tones they take after the infinitive prefix kèé-, e.g.

```
kèé-kfun 'to hit' kèé-kfùnè 'to be hitting' kèé-tón 'to refuse' kèé-tonè 'to be refusing' kèé-dòmsè 'to push' kèé-dòmsè 'to be pushing' kèé-címsé 'to pacify' kèé-címsè 'to be pacifying'
```

The above examples include monosyllabic L and H tone verbs followed by L and H bisyllabic verbs. As seen, the non-progressive forms on the left take pattern C, while the progressive forms on the right take pattern A. Thus, although we have cited the tonal contrast between monosyllabic verbs as one between M and H, e.g. kfun 'hit' vs. ton 'refuse', these citation forms actually represent the realization of this tonal contrast when occurring with pattern C.

Throughout our description of the verb conjugations we shall continue to use the infinitive tones for the purpose of citing verb forms. We shall also continue to use the above four simplex and complex verb forms to illustrate all of the tonal patterns in each tense, construction etc.

3. Imperatives. In some languages the second person singular imperative offers the least marked (i.e. affixless) verb form. As seen in the following examples, this is not the case in Noni:

```
kfùně 'hit!' È kfùnè 'be hitting!'
toné 'refuse!' È tonè 'be refusing!'
dòmsě 'push!' È dòmsè 'be pushing!'
cimsé 'pacify!' È cimsè 'be pacifying!'
```

The PROG forms on the right have a preposed marker & (found also in some consecutives; see Chapter 14) followed by the PROG verb forms with the expected tone pattern A. The non-PROG forms, however, have tone pattern B and a suffix if the verb form is monosyllabic. As seen in the following forms,

```
bić 'follow!' (bi 'follow')
clilč 'drag!' (cii 'drag')
```

this suffix takes the form - | E after a CVV verb radical and the form - E after either a CV or a CVC verb radical. In many cases the imperative suffix appears to be optional, e.g. yem or yeme 'sing!'. In a few verbs it takes an excep-

tional form (e.g. sho 'clean', sho'le 'clean!'). However, it cannot be present if either (a) the verb form has more than one syllable; or (b) a monosyllabic CV verb belongs to the H tone class. In the latter case the imperative verb carries a M tone: wo 'drink', wo 'drink!'. Since these suffixless imperative forms carry M tone, it is necessary to first assign the suffix, to derive a pattern B hypothetical form wo-£ and then delete the suffix (cf. the optional variants of 'sing!' given above. Finally, there is some evidence that in cases where the suffix is optional, the suffixed variant has a weaker imperative force, e.g. cem 'dig!' (a command) vs. cemé 'dig!' (an instruction), both from cem 'dig'.

Negative imperative forms are considerably more complex than their affirmative counterparts:

```
fò wò kfun kε
                'don't hit!'
                                   fò wò kfùnè kε
                                                   'don't be hitting!'
fà wà tổn kε
                'don't refuse!'
                                   fò wò tonè kε
                                                   'don't be refusing!'
fà wà dām̀sĕ kε
                'don't push!'
                                   fà wà dàmsè kε
                                                   'don't be pushing!'
fò wò címsé kε
               'don't pacify!'
                                   fò wò cimsè kε
                                                   'don't be pacifying!'
```

The structure of these utterances is NEG + 'you sg.' + verb + NEG. As in the infinitives, tone pattern C is used for non-PROG forms, while pattern A is used for PROG forms. While the subject pronoun wò 'you sg.' is indicated here with L tone, it can alternative be realized with LM tone, e.g. fò wò ton kɛ 'don't refuse!'.

Other imperatives (e.g. 1st and 2nd person pl.) use tone pattern C as well, e.g. ben kfun 'hit!' (you pl.), b5 ton 'let's refuse!'. The latter form appears to be derived from beews 'you and I' + ton. Notice the lack of the imperative suffix in these forms, as in the negatives.

4. Overview of the system. The exact realization of the different verb forms are of course affected by considerations of tense, aspect, mood and polarity. In addition, there are significant differences in the realizations of the different tenses etc. in main vs. relative vs. consecutive clauses. For this reason, the treatment of verb conjugations is presented in three separate chapters, with one chapter devoted to each clause. The affirmative and negative forms are given for each tense, as are the progressive and non-progressive. We shall have need, therefore, of the following features:

REL : relative/non-relative
CNS : consecutive/non-consecutive
PROG : progressive/non-progressive
NEG : negative/affirmative

NEG : negative/affirmative FUT : future/non-future TOD : today/non-today

NEAR : near/far

FOC : "even" focus/"uneven" focus

The first two features, REL and CNS, define the three constructions which will serve as the basis for the following three chapters: Chapter 12 will deal with main clauses, which are [-REL, -CNS], Chapter 13 with relative clauses, which are [+REL, -CNS], and Chapter 14 with consecutive clauses, which are [-REL, +CNS]. (The fourth feature combination, [+REL, +CNS], does not exist.) The feature specification [+NEG] refers to negative polarity, while [-NEG] refers to affirmative polarity. The feature PROG is addressed below.

The following three features, FUT, TOD and NEAR, define the eight distinguishable tenses or time distinctions recognized by the language. These

tenses have the feature specifications seen in the following table:

	P_3	P_2	P_1	P_{o}	F_{o}	F_1	F_2	F_3
[FUT]	-	-	-	-	+	+	+	+
[TOD]	-	_	+	+	+	+	-	-
[NEAR]	-	+	-	+	+	-	+	-

The first feature, FUT, distinguishes future from non-future tenses. As seen, F_0 , F_1 , F_2 and F_3 are all [+FUT], while P_0 , P_1 , P_2 and P_3 are [-FUT]. In the abbreviations for these tenses, F stands for "future", while P stands for "non-future" (i.e. either past or present, as indicated below).

Both the [+FUT] and [-FUT] tenses show four degrees of remoteness. The designation "zero" in P_0 and F_0 indicates 0 degree of remoteness from the time of speaking, while the integers 1, 2 and 3 indicate further degrees, in the order indicated. The most common time references indicated by these tenses are the following:

Po : present/immediate past 'to have just X'ed'

P₁ : earlier today

P2 : yesterday or some days ago

Pz : a long time ago

Fo : immediate future 'to be about to X'

F₁: later today

F2 : tomorrow or some days from now

Fz : a long time from now

The key to this analysis is seen in the two meanings of P_0 : if the P_0 is [+PROG] it has a present tense meaning, e.g. b5 kfùně lò 'they are hitting. If the same tense is [-PROG], it has an immediate past or present perfect meaning, e.g. b5 kfǔn lò 'they have (just) hit'. Normally, when the PROG suffix is added to a past or future tense, we obtain the progressive 'ing' reading for that tense, e.g. 'they hit' vs. 'they were hitting', 'they will hit' vs. 'they will be hitting'. It is in the P_0 that the PROG form changes a past tense into a present tense. The past tense which is so changed is, of course, a perfect of sorts, a past tense which has a close tie to or relevance for the present. Because of this relationship between the present and the immediate past, the most major tense distinction in the language is that between [+FUT] and [-FUT] (cf. Anderson's (1979) treatment of nearby Aghem, which has certain features in common, but which differs in having a completive/non-completive aspect, rather than a progressive/non-progressive).

Following the treatment of Bamileke-Dschang by Hyman (1980), the different degrees of past and future will be distinguished by means of the features TOD and NEAR. Tenses which refer to events occurring on the same day of the discourse are [+TOD], i.e. P_0 , P_1 , F_0 , F_1 , while tenses which refer to events not occurring on the same day are [-TOD], i.e. P_2 , P_3 , F_2 , F_3 . Finally, within each of the TOD specifications there is a near vs. far variant: [+TOD, +NEAR] indicates present or immediate past or future; [+TOD, -NEAR] indicates earlier and later today; [-TOD, +NEAR] indicates some days ago or from now; and [-TOD, -NEAR] indicates a long time ago or from now.

The reason why the feature NEAR is used is because of the subjectivity that may in some cases lead a speaker to use one or the other tense. An event

which took place some weeks ago, for example, will be in the P_2 ([-TOD, +NEAR]) if the speaker chooses to regard the time point as relatively close to today; it will be in the P_3 ([-TOD, -NEAR]) if he chooses to regard the time point as not close to today, and so forth. In other words, except for the today/non-today distinction, it would be incorrect to define tenses in terms of objective time reference.

The last feature which will be used in the following chapters is FOC. In main clauses only, and only among [-FUT] tenses, there is a distinction between "focused" and "non-focused" verb forms. While this distinction will come up in later chapters, it can be illustrated by means of the following sentences:

```
[+FOC] : bố cí nóỏ kfùn wàn 'they hit a child' (P2)
[-FOC] : bố cí kfǔn wan " " " " "
```

Both sentences report the same event with the same time reference. However, the internal focus structure is different. The first sentence is neutral, and could be uttered out of the blue. The second sentence appears only in answer to a question (e.g. 'whom did they hit?'). The [+FOC] variant has the auxiliary nob which indicates even focus among the arguments of the sentence and the action of the verb. It places emphasis, if anywhere, on the fact that the action took place (cf. Watters' (1979) notion of "truth value focus" for Aghem). Because the [+FOC] variant has a foregrounding function, it is usually not found in relative clauses (but cf. Chapter 19), which are considered to have a backgrounding referential function.

In the following chapters the different conjugations will be presented in the following order:

```
I. [-PROG]
   A. [-FUT]
      1. [-NEG]
         a. [-FOC]
             +FOC]
      2.
          +NEG
      [+FUT]
      1.
          [-NEG]
      2.
           [+NEG]
II. [+PROG]
   A. [-FUT]
      1.
          [-NEG]
          +NEG]
      [+FUT]
      1.
         -NEG]
          +NEG]
```

This is done first for main clauses ([-REL, -CNS]) in Chapter 12; then for relative clauses ([+REL, -CNS]) in Chapter 13; and then for consecutive clauses ([-REL, +CNS]) in Chapter 14. A summary is provided in Chapter 15.

VERB CONJUGATION II: MAIN CLAUSES

1. [-PROG, -FUT, -NEG, -FOC]. In this chapter we shall present the verb conjugations for main clauses. We shall follow the outline given at the end of Chapter 11 and therefore begin with non-progressive, non-future, affirmative tenses. Since these show a focus distinction, as alluded to in the previous chapter, we shall begin with the [-FOC] variants.

The tenses included under this heading include the P_0 , P_1 , P_2 and P_3 . We shall continue to use the verbs kfun 'hit', ton 'refuse', domse 'push' and cimsé 'pacify' to illustrate the two tone patterns on monosyllabic and bisyllabic verb forms.

1.1. [-FOC], P_O . As seen in the following sentences, the [-FOC] P_O has a \emptyset tense marker and takes tone pattern B:

```
bố khun lò 'they have just hit'
bố tổn lò 'they have just refused'
bố dồmsẽ lò 'they have just pushed'
bố cimsẽ lò 'they have just pacified'
```

The marker 13 'just' is required in order to make these sentences complete. The same will be seen in the other [-FOC] tenses in main clauses.

1.2. [-FOC], P_1 . This tense utilizes the auxiliary bé and tone pattern A:

```
bố bế kfùn lồ 'they hit' (earlier today)
bố bế tốn lồ 'they refused'
bố bế chmse lồ 'they pushed'
bố bế chmse lồ 'they pacified'
```

The tone pattern on monosyllabic H tone verbs such as ton 'refuse' is irregular. For this reason we shall consider the [-FOC] P_1 to be marked by a suffix with L tone, - \dot{V} , whose tonal effect is seen only on toh (which would otherwise have been realized ton in pattern A).

1.3. [-FOC], P_2 . This tense utilizes the auxiliary ci (probably from the main verb ce 'spend the night') and tone pattern B:

```
bố cí kfún lò 'they hit' (yesterday, some days ago)
bố cí tốn lò 'they refused'
bố cí dòmsẽ lò 'they pushed'
'they pacified'
```

1.4. [-FOC], P_3 . This tense utilizes the auxiliary to (from the main verb to 'come') and tone pattern B:

```
bố tổ khín lồ 'they hit' (a long time ago)
bố tổ tổn lồ 'they refused'
bố tổ dồmsẽ lồ 'they pushed'
bố tổ címsế lồ 'they pacified'
```

Note that the LH tone on the verb in 'they refused' is the result of the tone

rule referred to in Chapter 2 whereby a H tone becomes LH in close conjunction with a preceding L. We shall see a number of applications of this rule in the verb forms in this and successive chapters.

- 2. [-PROG, -FUT, -NEG, +FOC]. The above four non-progressive affirmative tenses have [+FOC] counterparts which are complete and cannot cooccur with 13.
- 2.1. [+FOC], $P_{\mathcal{O}}$. This tense is formed with the auxiliary not and tone pattern A:

bố nổờ khủn 'they have hit' bố nổờ dòmsè 'they have pushed' bố nổờ tồn 'they have refused' bố nổờ cĩmsè 'they have pacified'

The origin of the marker nóò, which will be seen in other [+FOC] tenses, is nú 'copula, focus marker' + ϵ , as can be seen in the slow variant bố nú ϵ kfùn 'they have hit'.

2.2. [+FOC], P_1 . This tense utilizes the auxiliary bee, which is from be+ ϵ , and tone pattern A:

bố bết kfun 'they hit' (today) bố bết dòmse 'they pushed' bố bết tồn 'they refused' bố bết cimse 'they pacified'

2.3. [+FOC], P_2 . This tense adds not to the auxiliary of seen in the [-FOC] variant. The tone pattern changes to A:

bố cí nóô kfun 'they hit' (yesterday, some days ago)

bá cí nóà tồn 'they refused'

bố cí nóỏ dòmsè 'they pushed' bố cí nóỏ cimsè 'they pacified'

2.4. [+FOC], P_3 . This tense adds not to the auxiliary to seen in the [-FOC] variant. Again the tone pattern changes to A:

bố tố nóô kfun 'they hit' (a long time ago)

bố tố nốc tồn 'they refused'

bố tố nóỏ dòmse 'they pushed'

- bố tổ nổô cimse 'they pacified'
- 3. [-PROG, -FUT, +NEG]. Since there are no focus variants in the negative, the following non-progressive P_0 P_3 forms correspond to both focus variants of affirmatives above:
- 3.1. [+NEG], P_O . In all of the non-progressive, non-future negatives the negative marker bá (variant báá) is inserted between the tense auxiliary (if any) and the verb. In addition, ke is added at the end of the clause:

bố bấ khun kẽ 'they haven't hit'

bố bấ tốn kε 'they haven't refused'

bo ba domse 'they haven't pushed'

bố bá címsé 'they haven't pacified'

Since the P_0 does not use a tense auxiliary, the negative marker bá stands between the subject and the verb. The tone pattern is C in all non-progressive negative forms. Notice, finally, that ke takes the place of 15 in completing the above sentences.

3.2. [+NEG], P_1 . This tense follows the above rule except that the negative marker is realized with a LH tone:

```
bổ bể bằ kfun ke 'they didn't hit' (earlier today)
bổ bể bằ tổn ke 'they didn't refuse'
bổ bể bằ dồmsẽ ke 'they didn't push'
bổ bể bằ címsẽ ke 'they didn't pacify'
```

Because of the unpredictable tonal alternation, we assume that there is a L tone preceding the negative marker ba in just this tense. This L tone should be compared with the L which was postulated after the main verb in the [-FOC] affirmative P_1 .

3.3. [+NEG], P_2 . This tense follows the normal pattern C with ba:

```
bổ cí bá kfun kể 'they didn't hit' (yesterday, some days ago)
bổ cí bá tổn kể 'they didn't refuse'
bổ cí bá domsě kể 'they didn't push'
bổ cí bá címsé kể 'they didn't pacify'
```

3.4. [+NEG], P_3 . This tense is also regular. The negative marker is realized LH because of the preceding L tone of the tense auxiliary the

```
bổ tổ bằ kfun ke 'they didn't hit' (a long time ago)
bổ tổ bằ tổn ke 'they didn't refuse'
bổ tổ bằ dồmsẽ ke 'they didn't push'
bổ tổ bằ címsẽ ke 'they didn't pacify'
```

- 4. [-PROG, +FUT, -NEG]. All future tenses are built on the future marker nú (variant núú, which may mark emphasis). This marker is derived from the copula verb 'be' of the same form. There are no focus variants in the future tenses, nor is 13 required to make the utterances complete.
- 4.1. [-NEG], F_O . This tense uses the future marker nu without any further tense auxiliary. In this and all other non-progressive future verb forms, the tone pattern is C:

```
bố nú kfun
bố nú tốn
bố nú đồmsế 'they are about to refuse'
bố nú đồmsế 'they are about to push'
bố nú címsế 'they are about to pacify'
```

4.2. [-NEG], F_1 . This tense uses the auxiliary géé (probably from the main verb ge 'make, do').

```
bố nú giế kfun 'they will hit' (later today; indefinite)
bố nú giế tốn 'they will refuse'
bố nú giế domse 'they will push'
bố nú giế címse 'they will pacify'
```

While the ${\rm F_1}$ expresses future actions later in the same day, it expresses a less definite future than the ${\rm F_0}$ tense. In general the time distinctions blur somewhat more in the future than in the past tenses.

4.3. [-NEG], F_2 . This tense uses the auxiliary boy (from the main verb boy 'exit, reach'), and the normal tone pattern C:

```
bố nú bốy kfun
bố nú bốy tốn
bố nú bốy đồmsẽ 'they will hit' (tomorrow, some days from now)
'they will refuse'
'they will push'
'they will pacify'
```

4.4. [-NEG], F_3 . This tense uses the auxiliary yúú (from the main verb yúú 'dawn, be clear'):

```
bố nú yấu kfun
bố nú yấu tốn
bố nú yấu đỡmsẽ 'they will hit' (a long time from now)
'they will refuse'
'they will push'
'they will pacify'
```

- 5. [-PROG, +FUT, +NEG]. All future and/or progressive verb tenses form their negatives by placing ké at the beginning of the clause and ke at the end of the clause. In addition, the marker yà can optionally appear immediately after the subject.
- 5.1. [+NEG], F_O . This tense is formed as indicated above. In all non-progressive future negatives the tonal pattern is C:

```
kế bố (yà) nữ kfun ke
kế bố (yà) nữ tổn ke
kế bố (yà) nữ dốmsẽ ke
kế bố (yà) nữ címsẽ ke
they are not about to push'
they are not about to push'
they are not about to push'
```

Although the negative marker γa is optional, as indicated by the parentheses, the tone on the future marker αa is LH even when γa is not present. This would seem to indicate that it is present underlyingly and is deleted optionally after it has changed αa to αa .

5.2. [+NEG], F_1 . This tense follows the general pattern.

```
kế bố (yà) nữ giế kfun ke
kế bố (yà) nữ giế tổn ke
kế bố (yà) nữ giế đoms ke
kế bố (yà) nữ giế đoms ke
kế bố (yà) nữ giế címs ke
'they will not push'
'they will not pacify'
```

5.3. [+NEG], F_2 . This tense follows the general pattern.

```
kế bố (yà) nử bốy kfun kẽ 'they will not hit' (tomorrow, some days from kế bố (yà) nử bốy domsě kẽ 'they will not push' kế bố (yà) nử bốy címsé kẽ 'they will not pacify'
```

5.4. [+NEG], F_3 . This tense follows the same general pattern, again with pre-clause k£, post-clause k£, and optional ya with tone pattern C:

```
kế bố (yà) nữ yứu kfun ke
kế bố (yà) nữ yứu tổn ke
kế bố (yà) nữ yứu đồmsẽ ke
kế bố (yà) nữ yứu cổmsể ke
they will not push'
'they will not push'
```

6. [+PROG, -FUT, -NEG]. Progressive verb forms utilize the PROG marking described in Chapter 10. In addition to the suffixal modifications that were enumerated, the tones of [+PROG] forms can be quite different from those of their [-PROG] counterparts.

6.1. [-NEG], θ . The [+PROG] P_0 tense has a present tense progressive meaning and will be indicated as the "O" tense. As seen in the following examples, it requires the same tone pattern B as its [-PROG] counterpart and the same 15 to make the utterance complete:

```
bó kfùně lò 'they are hitting'
bó toné lò 'they are refusing'
bó dòmsě lò 'they are pushing'
bó cimsé lò 'they are pacifying'
```

6.2. [-NEG], P_1 . This tense uses tone pattern A exactly as its [-PROG, -FOC] variant:

```
bố bế kfùnề lò 'they were (just) hitting' (earlier today)
bố bế tonề lò 'they were (just) refusing'
bố bế dòmsề lò 'they were (just) pushing'
bố bế cịmsề lò 'they were (just) pacifying'
```

6.3. [-NEG], P2. Again, the only difference between this and its [-PROG, -FOC] variant is the presence of the PROG suffix on 'hitting' and 'refusing'. Thus, tone pattern B is used:

```
bổ cí khủng là 'they were (just) hitting' (yesterday, some days ago)
bổ cí tong là 'they were (just) refusing'
bổ cí dòmsg là 'they were (just) pushing'
bổ cí cimsg là 'they were (just) pacifying'
```

Because domse 'push' and cimse 'pacify' do not change in the PROG, the above sentences are ambiguous with [+PROG] and [-PROG] readings.

6.4. [-NEG], P_3 . Again, tone pattern B is used exactly as in the [-PROG, -FOC] instances of the P_3 tense:

```
bố tổ khung lỏ 'they were (just) hitting' (a long time ago) bố tổ tong lỏ 'they were (just) refusing' bố tổ dồmsẽ lỏ 'they were (just) pushing' bố tổ cimsẽ lỏ 'they were (just) pacifying'
```

- 7. [+PROG, -FUT, +NEG]. The negative of progressive non-future tenses is exactly the same as non-progressive futures: ké at the beginning of the clause, ke at the end of the clause, and ya optionally just after the subject.
- 7.1. [+NEG], θ . Since there is no auxiliary in the 0 tense, the optional yà appears directly before the verb:

```
kế bố (yà) khunh ke 'they are not hitting' kế bố (yà) tonh ke 'they are not refusing' kế bố (yà) dòmsh ke 'they are not pushing' kế bố (yà) cimsh ke 'they are not pacifying'
```

The tone pattern for all progressive negatives is A.

7.2. [+NEG], P_1 . The forms for this tense are as expected, except that the tense marker is pronounced on a M tone, i.e. be. This probably reflects the change from be to be, which then simplified to M. The same simplification will be noted in the P_2 forms.

```
kế bố (yà) be kfùnh kể 'they were not hitting' (earlier today)
kế bố (yà) be tonh kể 'they were not refusing'
kế bố (yà) be cimsè kể 'they were not pushing'
'they were not pacifying'
```

7.3. [+NEG], P_2 . The forms again are regular except that the tense marker carries a M tone:

```
ké bó (yà) ci kfùnè kɛ
ké bó (yà) ci tonè kɛ
ké bó (yà) ci dòmsè kɛ
ké bó (yà) ci cimsè kɛ
ké bó (yà) ci cimsè kɛ
'they were not hitting' (yesterday, some days ago)
'they were not refusing'
'they were not pushing'
'they were not pacifying'
```

7.4. [+NEG], P_3 . This tense is realized as expected, i.e. with initial kE, final kE, and tone pattern A:

```
ké bɔ (yà) tò kfùnè kɛ 'they were not hitting' (a long time ago)
ké bɔ (yà) tò tonè kɛ 'they were not refusing'
ké bɔ (yà) tò cimsè kɛ 'they were not pushing'
ké bɔ (yà) tò cimsè kɛ 'they were not pacifying'
```

8. [+PROG, +FUT, -NEG]. Future progressives also take tone pattern A in all cases. As seen in the following paradigms, everything remains identical to the non-PROG forms except for tone. The verb, of course, takes the expected PROG form as predicted by Chapter 10.

```
8.1. [-NEG], F_O.
```

```
bố nú kfùnề 'they are about to be hitting'
bố nú tonề 'they are about to be refusing'
bố nú dòmsề 'they are about to be pushing'
bố nú cimsề 'they are about to be pacifying'
```

8.2. [-NEG], F7.

```
bố nú giếc khunh 'they will be hitting' (later today; indefinite) bố nú giếc tonh 'they will be refusing' 'they will be pushing' 'they will be pacifying'
```

8.3. [-NEG], F_2 .

```
bố nú bốy kfùne 'they will be hitting' (tomorrow, some days from now)
bố nú bốy tone 'they will be refusing'
bố nú bốy dòmse 'they will be pacifying'
```

8.4. [-NEG], F_3 .

```
bố nú yúú kfùnề 'they will be hitting' (a long time from now)
bố nú yúú tonề 'they will be refusing'
bố nú yúú dòmsè 'they will be pushing'
'they will be pacifying'
```

It is recalled that the PROG and non-PROG forms of 'push' and 'pacify' are identical except for tone.

9. [+PROG, +FUT, +NEG]. Like the non-PROG future tenses, [+PROG] F_0 - F_3 form their negative by placing ké at the beginning of the clause, ke at the end, and (optionally) ya immediately after the subject. Unlike the [-PROG] negative future tenses, which take tone pattern C, these tenses take tone pattern A.

9.1. [+NEG], F_{Q} .

```
kể bố (yà) nữ kfùnề kẽ 'they are not about to be hitting' kể bố (yà) nữ tonề kẽ 'they are not about to be refusing' kể bố (yà) nữ cịmsề kẽ 'they are not about to be pacifying' 'they are not about to be pacifying'
```

9.2. [+NEG], F_1 .

```
kế bố (yà) nữ giếc khung kể 'they will not be hitting' (later today; in-
kế bố (yà) nữ giếc tong kể 'they will not be refusing' definite)
kế bố (yà) nữ giếc dòms kể 'they will not be pushing'
kế bố (yà) nữ giếc cims kể 'they will not be pacifying'
```

9.3. [+NEG], F2.

```
kể bố (yà) nữ bốy kfùnề kể 'they will not be hitting' (tomorrow, some kể bố (yà) nữ bốy tonề kể 'they will not be refusing' days from now) kể bố (yà) nữ bốy cimsề kể 'they will not be pushing' 'they will not be pacifying'
```

9.4. [+NEG], F3.

```
kế bố (yà) nữ yửú kfùnề kẽ 'they will not be hitting' (a long time from kế bố (yà) nữ yửú tonề kẽ 'they will not be refusing' now)
kế bố (yà) nữ yửú cìmsè kẽ 'they will not be pushing'
kế bố (yà) nữ yửú cìmsè kẽ 'they will not be pacifying'
```

In all of the above forms the future marker nú has a LH tone because it is preceded by underlying yà, which optionally deletes by rule.

10. Compound P_3 and F_3 . The feature system introduced in Chapter II treats both the P3 and F3 as [-TOD, -NEAR]. The claim inherent in this feature system is that these two distant tenses constitute a natural class, i.e. they share semantic properties. Significant confirmation of this claim is derived from the fact that the P_3 marker to can be added to the F_3 and the F_3 marker yúú can be added to the P_3 :

```
bố tổ nổô tổn 'they refused' (a long time ago; definite time)
bố tổ nổô yùu tổn 'they refused' (a long time ago; indefinite time)
```

bố nú yúú tốn 'they will refuse' (a long time from now; definite time) bố nú tố yúú tốn 'they will refuse' (a long time from now; indefinite time)

(Although the [+FOC] variant of the P3 is seen in the first pair of sentences, the [-FOC] is obtained by deleting n55 and adding 15 to make the utterances complete.) The P3 marker precedes the F3 marker in both tenses. By combining the two a less definite time reference is interpreted: in the simple P3 and F3 the speaker probably knows when or approximately when the event took place

or will take place. In the compound forms, the speaker probably does not know when or if the event did or will take place. We shall see in Chapter 18 that Noni can allow several auxiliaries to occur in sequence before the main verb.

- 11. Irregular verbs. There are a number of verbs which are irregular in one or the other verb paradigm. Since we have addressed the question or irregular PROG forms in Chapter 10, the irregular forms which require discussion here are those occurring in [-PROG] tenses.
- 11.1. CVV radicals. CVV radicals undergo obligatory vowel shortening in [-PROG, -FUT, -FOC] tenses. The most common pattern is for the CVV syllable to shorten to CV and add | 6. 18 such verbs were elicited with this pattern, of which 7 are cited for illustrative purposes:

```
bố shìle là
             'they have just come down' (shii 'come down from')
bố belé là
             'they have just cried'
                                          (béé 'cry')
bố calé là
             'they have just pierced'
                                           (céé 'pierce')
bố tulé là
             'they have just carried'
                                          (túú 'carry on head')
bố solé là
             'they have just knotted'
                                          (soo 'knot')
bố fàiế là
             'they have just borrowed'
                                          (foo 'borrow')
bố sàlě là
             'they have just torn'
                                          (saa 'tear')
```

The above forms are in the [-FOC] Po. It is tempting to relate this lé suffix to the lé found in PROG forms of CVV radicals, e.g. bó shìlië lò 'they are coming down'. However, this could not account in any general way for the following exceptions to the above pattern:

(a) Four verbs have been found which shorten their vowel but do not take $l\,\acute{e}$:

```
bố cổ lò 'they have just dragged' (cii 'drag')
bố kổ lò 'they have just called/hooted' (kổ 'call by hooting')
bố kổ lò 's.o. has just tired of them' (fổ 'to tire s.o. of sth.')
bố kỗ là 'they have just ground' (kaa 'to grind')
```

(b) Six verbs shorten their vowel and take a suffix ke:

```
bố mề kế là
             'they have just carried (child)'
                                                (mee 'carry a child')
bố gèkế là
             'they have just put'
                                                (gee 'put')
bố đề kế là
            'they have just cooked'
                                                (dee 'cook')
bố fukể là
            'they have just become satiated'
                                                (fúú 'become satiated')
bá baké là
            'they have just borne a child'
                                                (bɔɔ́ɔ 'bear a child')
bố yakế là
            'they have just built'
                                                (vɔɔ́ɔ 'build')
```

The variant bo dele lo 'they have just cooked' has also been recorded.

(c) Three verbs have been found which shorten their vowel and take -ne:

```
bó bené lò 'they have just accepted/they are accepting' (béé 'accept') bó sené lò 'they have just told/they are telling' (séé 'tell') bó tené lò 'they have just called/they are calling' (téé 'call')
```

As seen in section 4.2 of Chapter 10, the forms with $-\eta e$ are also found in the PROG. This means that the above sentences are ambiguous, as indicated, between P_0 and 0 readings. Finally, there is the following exceptional verb:

```
bố kontế là 'they have just caught' (kốổ 'catch')
```

It is significant to note that H tone verbs of groups (b) and (c) are also tonally exceptional. Instead of having LM tone in tone pattern A, they show a ML pattern:

```
bố nổo boò 'they have borne a child' bố nổo beè 'they have built' bố nổo beè 'they have accepted' 'they have told'
```

What this reveals is that b55 'bear a child', y55 'build', béé 'accept', séé 'tell', etc. are bisyllabic, i.e. they consist of a CV syllable followed by a V syllable. The velar consonants k and n probably were once part of the verb stem, which may thus have been either CVC or, more likely, CVCV. By treating verbs in (b) and (c) as bisyllabic, we can assign their tones as we do to verbs such as cimsé 'pacify', e.g. b5 n55 cimsè 'they have pacified'.

11.2. Tonal exceptions. We have accounted for the tonal irregularities of the verbs in (b) and (c) in a principled way. A few additional verbs show other tonal irregularities.

First, the following two verbs have LH instead of LM in tone pattern A:

```
bố nốc diể 'they have eaten' (diể 'eat')
bố nốc fi 'they have helped' (fi 'help, take')
```

No explanation for this irregular tonal behavior comes to mind.

The second set of tonal irregularities concerns L tone verbs having the -m 'positional' extension (Chapter 9, section 3.3). Where other verbs show tone pattern C, these verbs show tone pattern B:

```
kèé-jìím 'to lie down'
kèé-shìím 'to hang oneself'
kèé-ñìím 'to hide oneself'
```

It may be that verbs of the form CVVm are actually inherently progressive, i.e. 'to be lying down', etc. The following recorded pair of sentences shows that at least for the verb ngvuum 'kneel', a vowel shortening process accounts for the observed [-PROG] reading:

```
bố ngyunmé là 'they are kneeling' (as a group)
bố ngyumé là 'they have just knelt'
```

These verbs require further investigation in the different tense/aspects.

VERB CONJUGATION III: RELATIVE CLAUSES

Some tenses are realized differently according to whether they occur in a main vs. relative clause. For this reason, relative clause tenses are treated in a separate chapter. The differences which are found may be tonal and/or segmental. Although the sentences cited below are all object relativized, relative clause tenses are uniform whatever the type of relative clause. (As seen in Chapter 19, subject relatives involve a LM tone relative pronoun, while all other relative constructions utilize a L tone relative pronoun.) In describing the various tenses as they occur in relative clauses, we shall follow the same outline as in the preceding chapter.

- 1. [-PROG, -FUT, -NEG, -FOC]. The [-FOC] variants of these tenses are much more frequently attested in relative clauses than their [+FOC] variants. Significantly, they do not require 13 to complete the clause.
- 1.1. [-FOC], P_O . This tense does not differ from its realization in main clauses. Thus, the following examples show tone pattern B:

```
wè wvù bố kfun 'the person they have hit'
wè wvù bố tổn 'the person they have refused'
wè wvù bố dòmsě 'the person they have pushed'
wè wvù bố cimsé 'the person they have pacified'
```

1.2. [-FOC], P_1 . This tense also is identical to its counterpart in main clauses, i.e. it takes tone pattern A and a L tone suffixed to the verb:

```
wè wvù bố bế kfùn 'the person they hit' (earlier today)
wè wvù bố bế tồn 'the person they refused'
wè wvù bố bế címsè 'they person they pacified'
```

1.3. [-FOC], P_2 . This tense differs from its realization in main clauses in two ways: (a) it takes tone pattern A instead of B, and (b) it requires the suffix $-l \grave{\epsilon}$ (realized $-\grave{\epsilon}$ after a consonant and optionally after a short vowel):

```
wè wvù bố cí kfùnề 'the person they hit' (yesterday, some days ago)
wè wvù bố cí tonề 'the person they refused'
wè wvù bố cí dòmsèè 'the person they pushed'
wè wvù bố cí címsèè 'the person they pacified'
```

It is noted in the last two examples that se+ ϵ is realized -s $\epsilon\epsilon$.

1.4. [-FOC], P_3 . This tense also has tone pattern A (rather than the B found in main clauses) and takes the same suffix as the [-FOC] P_2 :

```
wè wvù bố tò kfùnề 'the person they hit' (a long time ago)
wè wvù bố tò tonề 'the person they refused'
wè wvù bố tò dòmsèè 'the person they pushed'
wè wvù bố tò cimsèè 'the person they pacified'
```

The P3 marker to may be deleted in affirmative relative clauses, e.g. we wvù

bố khung 'the person they hit' (a long time ago).

2. [-PROG, -FUT, -NEG, +FOC]. These tenses do not differ from their counterparts in main clauses. Thus, as seen below,

```
wè wvù bố nổô kfùn 'the person they have hit' (Po) wè wvù bố bếc kfùn 'the person they hit' (P1) wè wvù bố tổ nổô kfùn 'the person they hit' (P2) 'the person they hit' (P3)
```

they take tone pattern A and do not acquire the -Iè suffix seen in [-PROG, -FUT, -NEG, -FOC] P_2 and P_3 relatives. Their specific function in relative clauses is discussed in Chapter 19.

3. [-PROG, -FUT, +NEG]. These tenses are identical with their main clause counterparts. As seen in the following examples,

```
wè wvù bổ bấ kfun ke 'the person they haven't hit' (P<sub>0</sub>)
wè wvù bổ bế bắ kfun ke 'the person they didn't hit' (P<sub>1</sub>)
wè wvù bổ cổ bấ kfun ke 'the person they didn't hit' (P<sub>2</sub>)
wè wvù bổ tổ bấ kfun ke 'the person they didn't hit' (P<sub>3</sub>)
```

they thus take bá in the P_0 , P_2 and P_3 or bá in the P_1 . (The LH tone on the underlying bá in the P_3 is conditioned by the preceding L of tà.) They all have clause-final $k\epsilon$ and tone pattern C, as in main clauses.

4. [-PROG, +FUT, -NEG]. The future tenses F_0 - F_3 appear exactly as in main clauses, i.e. with tone pattern C:

```
wè wvù bố nữ kfun 'the person they are about to hit' (F<sub>0</sub>)
wè wvù bố nữ géể kfun 'the person they will hit' (F<sub>1</sub>)
wè wvù bố nữ bốy kfun 'the person they will hit' (F<sub>2</sub>)
wè wvù bố nữ yữữ kfun 'the person they will hit' (F<sub>3</sub>)
```

5. [-PROG, +FUT, +NEG]. There are two patterns of negation in relative future tenses. One is identical to the main clause except (a) clause-initial $k\dot{\epsilon}$ is not present, and (b) $y\dot{a}$ is obligatory:

```
wè wvù bố yà nữ kfun ke 'the person they are not about to hit' (F_0) wè wvù bố yà nữ gếể kfun ke 'the person they will not hit' (F_1) wè wvù bố yà nữ yữu kfun ke 'the person they will not hit' (F_2) wè wvù bố yà nữ yữu kfun ke 'the person they will not hit' (F_3)
```

As in the main clause, the tone pattern is C and the clause ends with $k\epsilon$.

The second pattern of negativizing relative future tenses is by placing \hat{f} \hat{g} between \hat{n} and the future auxiliary (if there is one). Again, the tone pattern is C:

```
wè wvù bố nú fố ya kfun wan ke 'the person they are not about to hit' (F_0) wè wvù bố nú fố ya gốc kfun wan ke 'the person they will not hit' (F_1) 'the person they will not hit' (F_2) 'the person they will not hit' (F_3)
```

Recall the negative imperative markers fò yà (fò ya ?) seen in section 3 of Chapter 11, which also took tone pattern C. The marker fô apparently comes from the verb fáŋ 'refuse' and intensifies the negation.

6. [+PROG, -FUT, -NEG]. These tenses are equivalent to their main clause counterparts. As seen in the following examples,

```
wè wvù bố kfùně 'the person they are hitting' (0) 'the person they were hitting' (P1) wè wvù bố cổ kfùně 'the person they were hitting' (P2) 'the person they were hitting' (P3)
```

the P_1 continues to be distinguished from the 0, P_2 and P_3 tenses in taking tone pattern A instead of tone pattern B. This same pattern was observed in Chapter 12 in the affirmative [-PROG, -FOC] and [+PROG] tenses in sections 1 and 6, respectively (see also the summary in Chapter 15).

7. [+PROG, -FUT, +NEG]. These tenses cause some confusion. In main clauses the negatives of non-future progressive tenses all took tone pattern A. The main clause construction consisted of initial ké, final ke, and optional yà. This same construction can be used in relative clauses except, as seen above in section 5, the clause-initial ké is not present and the marker yà is obligatory. However, two tone patterns have been noted. First, in object relative constructions pattern A has been recorded (with some uncertainty in the 0 tense, whose B tone pattern should be verified):

```
wè wvù bố yè kfùně ke 'the person they are not hitting' (0)
wè wvù bố yà bế kfùnè ke 'the person they were not hitting' (P1)
'the person they were not hitting' (P2)
wè wvù bố yà tổ kfùnè ke 'the person they were not hitting' (P3)
```

In subject relative constructions, however, the marker ya has H tone and the overall tone pattern is A for P_1 and B for O, P_2 and P_3 :

```
wè w'u ya kfùně ke 'the person who is not hitting' (0)
wè w'u bé yà kfùně ke 'the person who was not hitting' (P1)
wè w'u cí ya kfùně ke 'the person who was not hitting' (P2)
wè w'u tò ya kfùně ke 'the person who was not hitting' (P3)
```

As a further complication, an alternative negative construction is available using the ba negative marker seen in section 3:

```
wè wvù bố bấ kfùnè ke 'the person they are not hitting' (0) wè wvù bố bế bắ kfùnè ke 'the person they were not hitting' (P1) 'the person they were not hitting' (P2) wè wvù bố tổ bắ kfùnè ke 'the person they were not hitting' (P3)
```

As observed, this construction imposes tone pattern A on all four tenses.

8. [+PROG, +FUT, -NEG]. As in main clauses, future progressives take tone pattern A:

```
wè wvù bố nữ kfùnè 'the person they are about to be hitting' (F_0) wè wvù bố nữ bốy kfùnè 'the person they will be hitting' (F_1) 'the person they will be hitting' (F_2) 'the person they will be hitting' (F_3)
```

9. [+PROG, +FUT, +NEG]. As in the non-future progressives, there are two negative relative constructions. The first utilizes ya and assigns the expected tone pattern A:

wè wvù bố yà nữ kfùnè ke 'the person they are not about to be hitting' (F_0) wè wvù bố yà nữ giế kfùnè ke 'the person they will not be hitting' (F_1) wè wvù bố yà nữ yữ kfùnè ke 'the person they will not be hitting' (F_2) 'the person they will not be hitting' (F_3)

The second negative construction utilizes fô ya and, again, tone pattern A:

wè wvù bố nú fố ya kfùnè ke wè wvù bố nú fố ya giế kfùnè ke wè wvù bố nú fố ya bốy kfùnè ke wè wvù bố nú fố ya yúủ kfùnè ke (F1,F2,F3, respectively)

Since Noni speakers in villages such as Mbinon and Lassin are said to pronounce $f\hat{s}$ $y\bar{s}$ as $f\hat{s}\eta\hat{e}$ $y\bar{s}$, we see the etymology of the first of the two negative markers from the main verb $f\hat{s}\eta$ 'refuse' (cf. Chapter 14, section 10).

VERB CONJUGATION IV: CONSECUTIVE CLAUSES

The term "consecutive" refers to verb conjunction without an overtly expressed subject. As seen in the following examples,

```
bố cí nóỏ yèm / bố n 'they sang and danced'
bố cí nóỏ mèsè / báck bènkfù 'they finished cooking yams'
bố cí nóỏ lewtè / èghảw me 'they ran to market'
bố cí nóỏ lewtè / èghảw me 'they ran faster than I'
```

this construction has a number of different, though related functions in Noni. In the above examples, all in the [+FOC] P2 tense, the consecutivized verb is separated from the first main verb by a slash (/). The first sentence has a consecutive verb form Ebin '& danced' used simply to represent the coordination of two actions. (The second normally, but not necessarily, follows the first.) In the second sentence it is observed that the consecutive verb form èdeè '& cooked' provides a complement to the verb mesé 'finish'. In the third sentence, $eg\overline{eh}$ '& went' is a directional verb which is required to indicate motion towards the market. The sentence literally means 'they ran and went to market'. Were ${\rm eg}\overline{\rm e}{\rm h}$ not present, the meaning would be 'they ran at the market', i.e. there would be no direction, only movement within a locality. Finally, the last sentence involves the consecutive verb form eghaw '& overcame, & was superior to'. It thus literally means 'they ran and overcame me', one of the most frequent ways to express a comparative in the language. It can be noted that Noni does not use constructions with 'take' + a consecutive verb form, as has been reported in Eastern Grassfields Bantu, e.g. Bamileke-Fe'fe' (Hyman 1971).

In the following sections we shall simply assume the context 'to sing' followed by a consecutive verb form, e.g. 'they sang and hit', 'they are singing and hitting'. Since this does not sound complete, we may add the object boom 'children' to complete the sense. We shall follow the same outline for the presentation of verb forms as in previous chapters. However, it is important to note that the binary verbal features refer to the tense, aspect and polarity of the main verb that precedes the consecutive form. In the last sections of this chapter we shall see examples where the feature specification [+NEG] refers exclusively to the consecutive clause.

- 1. [-PROG, -FUT, -NEG, -FOC]. These tenses all take a special consecutive verb form carrying tone pattern ${\tt D}.$
- 1.1. [-FOC], $P_{\rm O}$. As seen in the following examples, a consecutive verb following a $P_{\rm O}$ tense takes a prefix è with its D tonal pattern:

```
bó yém lò kkfuh boom
bó yém lò któn boom
bó yém lò kdomse boom
bó yém lò kcímse boom
bó yém lò kcímse boom
bó yém lò kcímse boom
```

The L-HL-L of '& pacified' is also heard as L-H-L, i.e. ècímsè boom '& pacified children'. We shall continue to transcribe it with the contour tone.

1.2. [-FOC], P_1 . The consecutive forms following a P_1 verb are identical to those just seen for the P_0 :

```
bố bế yem kkfun boom 'they sang and hit children' (earlier today)
bố bế yem ktổn boom 'they sang and refused children'
bố bế yem kcẩmse boom 'they sang and pushed children'
bố bế vem kcẩmse boom 'they sang and pacified children'
```

1.3. [-FOC], P_2 . Again, the forms are identical to the P_0 :

```
bố cí yếm kkfuh boom 'they sang and hit children' (yesterday, some days bố cí yếm ktổn boom 'they sang and refused children' ago)
bố cí yếm kcîmsè boom 'they sang and pushed children'
'they sang and pacified children'
```

1.4. [-FOC], P_3 . In this case the tones are the same as the above, but the prefix $\hat{\epsilon}$ is not present:

```
bố tổ yếm kfun boom
bố tổ yếm tổn boom
bố tổ yếm domse boom
bố tổ yếm cầmse boom
bố tổ yếm cầmse boom
bố tổ yém kfun boom
'they sang and hit children' (a long time ago)
'they sang and refused children'
```

In the second sentence we note that the consecutive verb form ton '& refused' does not have the rising tone seen in the preceding examples. From this we safely conclude that the L of the prefix &- causes the H of '& refuse' to become a LH rising tone.

2. [-PROG, -FUT, -NEG, +FOC]. The [+FOC] variants of the P_0 - P_3 tenses take exactly the same consecutive verb forms as their [-FOC] counterparts, as seen in the following representative sentences:

```
bố nổo yếm kkfun boom 'they have sung and hit children' (Po) bố bốt yếm kkfun boom 'they sang and hit children' (P1) bố tổ nổo yếm kfun boom 'they sang and hit children' (P2) bốt tổ nổo yếm kfun boom 'they sang and hit children' (P3)
```

3. [-PROG, -FUT, +NEG]. After a negative (and [-PROG]) P_0 , P_1 , P_2 or P_3 , the consecutive form consists of the bare verb with tone pattern C:

```
bố bấ yếm kfun boom ke 'they haven't sung and hit children' (P<sub>0</sub>) bố bế bấ yếm kfun boom ke bố cổ bấ yếm kfun boom ke bố tổ bấ yếm kfun boom ke bố tổ bấ yếm kfun boom ke 'they didn't sing and hit children' (P<sub>2</sub>) 'they didn't sing and hit children' (P<sub>3</sub>)
```

In this case the consecutive verb forms take the same tonal pattern as the main verb. The negative marker ke comes at the end of the clause.

4. [-PROG, +FUT, -NEG]. After a [-PROG] affirmative future tense, the consecutive verb again takes a tone pattern C, like the main verb:

```
bố nú yếm kfun boom 'they are about to sing and hit children' (F_0) bố nú bốy yếm kfun boom 'they will sing and hit children' (F_1) bố nú yốu yếm kfun boom 'they will sing and hit children' (F_2) 'they will sing and hit children' (F_3)
```

5. [-PROG, +FUT, +NEG]. The same consecutive forms with tone pattern C are found after [-PROG] future negatives as were seen after the corresponding affirmatives in section 4:

```
kế bổ nữ yếm kfun boom ke 'they are not about to sing and hit children' (F<sub>c</sub> kế bổ nữ bốy yếm kfun boom ke 'they will not sing and hit children' (F<sub>2</sub>) kế bổ nữ yữu yếm kfun boom ke 'they will not sing and hit children' (F<sub>3</sub>)
```

6. [+PROG, -FUT, -NEG]. AZZ [+PROG] tenses take a [+PROG] consecutive verb form with tone pattern A. Since there are no divergences from this pattern, we shall continue to illustrate these tenses with the verb 'to hit' only.

```
bố yemtế khung boom
bố bế yemtế khung boom
bố cổ yemtế khung boom
bố tổ yemtế khung boom
```

7. [+PROG, -FUT, +NEG]. In the following sentences we have left the optional negative marker ya out:

```
kế bố yemtè kfùnè boom kẽ 'they are not singing and hitting children'(0) kế bố be yemtè kfùnè boom kẽ 'they were not singing and hitting children' kế bố ci yemtè kfùnè boom kẽ '' '' (P_2) (P_1)
```

8. [+PROG, +FUT, -NEG].

```
bố nú yemtè khunè boom
bố nú giế yemtè khunè boom
bố nú bốy yemtè khunè boom
bố nú yuấu yemtè khunè boom
bố nú yuấu yemtè khunè boom
```

[+PROG, +FUT, +NEG].

```
kế bố nử yemtè kfùnè ke 'they are not about to be singing and hitting kế bố nử giế yemtè kfùnè ke kể bố nử bốy yemtè kfùnè ke 'they will not be singing and hitting children' kế bố nử yúủ yemtè kfùnè ke (F_1, F_2, F_3, respectively)
```

10. Since the above forms show one subject held constant for two (or potentially more) verbs, the consecutive construction is strictly speaking not a clause in its own right. (The sequential construction, treated in Chapter 20, involves a change of subject and therefore does constitute a clause.) However, because it can have certain independent properties, we will continue to refer to it as a clause. In reality we have a succession of verb phrases.

One such independent property is negation. In all of the above examples involving negation, the negative marking was conditioned by the first or main verb, but carried through the sentence. In the following subsections we shall consider cases where only the consecutive verb phrase is negative.

10.1. [-PROG, -FUT, +NEG]. The following sets of sentences illustrate the construction used when the main verb is affirmative, but the consecutive verb is negative. Again we see that the form (è) fwo or its variant (è) fáné appears as a negative marker, although derived from the main verb 'refuse'.

```
bố nóỏ yèm thuổ ya khun boom ke 'they have sung and not hit children' (0) bố nóỏ yèm thuổ ya domst boom ke bố nóỏ yèm thuổ yà címse boom ke bố bết yèm thuổ ya khun boom ke etc. 'they have sung and not pushed children' 'they have sung and not pacified children' 'they have sung and not pacified children' 'they sang and didn't hit children' (P1)
```

bố cí nốờ yếm thườ yà khun boom ke 'they sang and didn't hit children' (P_2) etc.

```
bố tổ nốc yếm fwố ya kfun boom ke 'they sang and didn't hit children' (P3) bố tổ nốc yếm fwố ya tổn boom ke 'they sang and didn't refuse children' 'they sang and didn't push children' bố tổ nốc yếm fwố ya címsé boom ke 'they sang and didn't pacify children'
```

As seen in section 1, consecutive verbs following [-PROG, -FUT, -NEG] tenses take tone pattern D and in all but the P3 a prefix $\hat{\epsilon}$. The negative marker fw5/fáné shows this same alternation between $\hat{\epsilon}$ fw5 in P0, P1 and P2, and fw6 in P3. It therefore is clearly a verb to take this verb prefix. Perhaps these sentences once meant 'they sang and refused to hit children'. Note that the consecutive verbs appear in tonal pattern C.

10.2. [-PROG, +FUT, +NEG]. If the non-progressive tense is future, a sequential construction must be used, e.g.

bố nú yếm bố sàá kfun boom kε 'they are about to sing and not hit children' (F₀)

The above is not a consecutive construction because the subject bo 'they' is repeated in the second clause. The marker saa, note, probably derives from se 'before' + ya 'negative'.

10.3. [+PROG, +NEG]. Since very little material was collected of this type, our discussion will have to be restricted to the 0 tense, for which forms like the following were noted:

bố yemté yà khung boom ke 'they are singing and not hitting children'

This construction with ya and PROG verb forms taking tone pattern A can probably be generalized to any of the [+PROG] tenses.

15

SUMMARY AND RULES

1. The tonal patterns seen in the preceding chapters are summarized in the following table:

	A ('R')	B ('R')	C ('R')	D ('R')
[-REL, -CNS, -PROG, -NEG]	P _O -P ₃ [+FOC] P ₁ [-FOC]	P ₀ , P ₂ , P ₃ [-FOC], Imp	Fo-F3	
[-REL, -CNS, -PROG, +NEG]			P ₀ -P ₃ , F ₀ - F ₃ , Imp	
[-REL, -CNS, +PROG, -NEG]	P ₁ , F ₀ -F ₃ Imp	0, P ₂ , P ₃	3, 1	
[-REL, -CNS, +PROG, +NEG]	0, P ₁ -P ₃ , F ₀ - F ₃ , Imp			
[+REL, -CNS, -PROG, -NEG]	P ₁ -P ₃ [-FOC] P ₀ -P ₃ [+FOC]	Po [-FOC]	Fo-F3	
[+REL, -CNS, -PROG, +NEG]			P ₀ -P ₃ , F ₀ -F ₃	
[+REL, -CNS, +PROG, -NEG]	$P_1, F_0 - F_3$	0, P ₂ , P ₃	Ü	
[+REL, -CNS, +PROG, +NEG]	P ₁ -P ₃ , F _o -F ₃	0		
[-REL, +CNS, -PROG, -NEG]			F _O -F ₃ , Imp	Po-P3
[-REL, +CNS, -PROG, +NEG]	. E		P _O -P ₃ , F _O - F ₃ , Imp	
[-REL, +CNS, +PROG, -NEG]	0, P ₁ -P ₃ , F ₀ - F ₃ , Imp			
[-REL, +CNS, +PROG, +NEG]	0, P ₁ -P ₃ , F ₀ - F ₃ , Imp			

In the above table the imperative (Imp) has been included, although we have not exemplified all of its properties (e.g. imperative + consecutive verb). We have not included the ba negatives found in progressive relatives.

Looking over this distribution, a number of recurrent patterns become evident. It is not clear how these patterns should be accounted for. The following questions arise:

- (1) Should we assign patterns, e.g. A-D, or pre- and post-posed floating tones, e.g. 'R', 'R', and 'R'?
- (2) Should these patterns and/or floating tones be grouped into semantic classes (e.g. realis vs. irrealis)?

(3) Should the rules be conditioned directly by the above tense, aspect and polarity features, or by the constructions in which the different feature combinations occur?

This last question is particularly interesting. Most tenses in Noni involve auxiliary verbs which of course combine with a following main verb. Now the question arises as to the nature of the construction holding between each auxiliary and main verb combination. Consider, for example, the fact that [-PROG] future consecutives are characterized by tone pattern C. At the same time, however, [-PROG] futures have this tone pattern in main clauses. Since the future tenses employ verbal auxiliaries, one might argue that the main verb is in a consecutivized form, and therefore, that tone pattern C is, basically, a consecutive tone pattern.

We shall not attempt to resolve the difficulties arising in these different approaches. Rather, we shall present one set of rules assigning patterns on the basis of the features REL, CNS, PROG, NEG and the tense features. These may serve as a first attempt and perhaps a basis for further investigation.

- 2. The rules. The above table indicates several general patterns. It also has exceptions. In presenting the following set of ordered rules, it has been decided that exceptional tone assignments should come earliest, followed by the more general rules. It is not claimed that these rules are definitive, especially since some tenses are left out (e.g. tones in sequential clauses). It is interesting, however, that once the exceptions are accounted for by specific rules, so much can be predicted by referring to the class features.
- 2.1. P2 and P3 - $|\epsilon|$. The first rule assigns tone pattern A and - $|\epsilon|$ to the P2 and P3 as follows:

$$\begin{bmatrix} -FUT \\ -TOD \end{bmatrix} \longrightarrow A-1\epsilon / \begin{bmatrix} \hline +REL \\ -PROG \\ -NEG \\ -FOC \end{bmatrix}$$

The tenses P_2 and P_3 are referred to with the feature specifications [-FUT, -TOD]. The tone pattern and -1 ϵ suffix are assigned only in relative clauses where the two tenses are non-progressive, affirmative and non-focused. Note that the L tone found on the suffix is part of the tone pattern, rather than belonging directly to the suffix.

 $2.2.\ P_1.$ A glance at the table reveals that the P_1 tense usually takes tone pattern A, whatever the cooccurring feature specifications. This fact is captured by the following rule:

$$\begin{bmatrix} -FUT \\ +TOD \\ -NEAR \end{bmatrix} \rightarrow A \quad (except [-PROG, +NEG])$$

2.3. P_O - P_3 consecutives. The rare tone pattern D is assigned to these consecutives when they are non-progressive and affirmative:

It should be noted that the feature specifications [-FUT, -PROG, -NEG] are obtained from the main verb. Also, an additional statement will be needed to spell out the prefix $\hat{\epsilon}$ which characterizes the P_O, P₁ and P₂ consecutive.

2.4. [+FOC]. As seen in the table, all [+FOC] verb forms receive tone pattern A:

2.5. Tone pattern A. At this point three rules remain to assign the general cases of tone patterns A, C and B. These will be given in this slightly reordered way.

The first general rule assigns tone pattern A whenever the feature specification [+PROG] combines with any of the following: [+CNS], [+FUT] or [+NEG]. This rule is presented informally as follows:

[+PROG] +
$$\{[+CNS], [+FUT], [+NEG]\} \rightarrow A$$

Thus, a consecutive progressive, a future progressive or a negative progressive tense will automatically receive tone pattern A. Some feature combinations will, of course, have more than one of the required features, e.g. the verb tenses which are [+CNS, +PROG, +FUT, +NEG], which have all four!

2.6. Tone pattern C. The general rule for the assignment of tone pattern C is as follows:

$$[-PROG] + \{ [+FUT], [+NEG] \} \rightarrow G$$

Any tense which is either [-PROG, +FUT] or [-PROG, +NEG] will receive tone pattern C at this point.

2.7. Tone pattern B. Since this is the last rule, we have the possibility of stating this last tone assignment as applying "elsewhere". However, because the remaining forms constitute a natural class, namely [-FUT], we shall state this rule as follows without further comment:

3. Exceptions. Further refinements may be needed as more data are incorporated. Most of the potential exceptions have been accounted for by the specific rules which precede the last three rules. However, we have not account for either the relative progressive negatives with bá, or the 0 relative progressive negative which has B rather than the expected A tone. This latter tense may thus have to be considered an exception. Finally, recall from page 64 that there are some tonal irregularities that have either to do with specific verb forms (e.g. dié 'eat', fi 'help, take') or with a definable class of verbs, those taking the -m 'positional' extension, e.g. jim 'lie down'. These will have to be incorporated within the analysis somehow.

PART IV: BASIC SENTENCE STRUCTURE

16

WORD ORDER AND GRAMMATICAL RELATIONS

1. Basic word order. Noni is a typical SVO language whose unmarked word order is:

S AUX V DO IO X

In a Noni sentence with neutral focus (see below), the subject (S) will precede the auxiliary (AUX), which in turn precedes the main verb (V). After the verb will come the direct object (DO) and indirect object (IO) followed by any additional adverbials or prepositional phrases (e.g. locatives, temporals). An example of a neutral sentence with all of the above elements is the sentence

bś cỉ ñá kèngòm ϵ boom fogò 'they gave plantains to they P_2 gave plantains to children yesterday children yesterday' S AUX V DO IO X

In the above example the [-FOC] $\rm P_2$ tense is used and the X of the formula S AUX V DO IO X is represented by a temporal adverb.

Noni sentences have been observed to diverge from this pattern in two different situations. First, a focused subject is postposed immediately after the verb:

 $\hat{\epsilon}$ cĩ ná bố kèngòm ϵ boom fogò 'they gave plantains DS P_2 gave they plantains to children yesterday to children yesterday'

In this sentence the subject b5 'they' occurs immediately after the verb, indicating that it is focused (as, for example, in response to the question 'who gave plantains to children yesterday', or as a contrast to someone else giving plantains to children yesterday). The initial subject position remains basic, however, as seen by the obligatory dummy subject (DS) marker & which appears in this position. Everything else remains the same (i.e. the DO precedes the IO which precedes the time adverbial).

The second divergence concerns the indirect object relation. A limited number of Noni verbs, e.g. 'to give', allow an alternative construction as seen in the sentence

bố cí ná boom bố kếngôm fogò 'they gave plantains to they P_2 gave children with plantains yesterday children yesterday'

A different preposition is used (b5 'with' instead of ϵ 'to') and the word order of 'plantains' and 'children' is reversed. One could maintain, however, that in this last sentence the DO is 'children' and the indirect (or "oblique") object is 'plantains' (cf. section 4).

2. The subject relation. As seen in the preceding examples, the subject is expected to be initial in its clause, although it may be postposed immediately after the verb if it is to be focused. The subject pronouns are exactly as given at the beginning of Chapter 4. The six primary human subject pronouns are illustrated with the intransitive verb gwe 'fall' below:

```
me ngwě lò 'I have just fallen'
wò gwě lò 'you [sg] have just fallen'
wvù gwě lò 'he/she has just fallen'
bèsèn gwèkěn lò 'we [excl] have just fallen'
bò gwèkěn lò 'they have just fallen'
'they have just fallen'
```

These examples illustrate several important points about the subject-verb relation in Noni. First, the sentence me ngwě là 'I have just fallen' shows a homorganic nasal on the verb. This nasal must appear on αll verbs (including auxiliaries) when the subject is me 'I', e.g.

```
me nú <u>n</u>tố <u>n</u>yúú <u>n</u>kfun wan 'I will hit a child' [a long time from now, F_3] me <u>ncí nốờ nyêm knkfuh wàn</u> 'I sang and hit a child' [yesterday, P_2]
```

These sentences show the nasal prefixes on various tense auxiliaries. The auxiliaries not '[+FOC]' and nú '[+FUT]' do not take a nasal prefix, although they begin with a nasal. Otherwise all verbs take 1st person singular subject-verb agreement and all auxiliaries taking this agreement are verbs. The second example above shows that this agreement process even extends into consecutive clauses. The example

me <u>nyemté nyà mbinè ke</u> 'I am singing and not dancing'

shows nasal agreement not only on the main verbs 'sing' and 'dance', but also on the negative morpheme yà. This suggests that the latter form is also a verb, at least when used consecutively as in the example.

The second interesting point about the examples 'fall' above is that a plural subject conditions a verbal suffix -ken. This particular suffix was discussed and further exemplified in section 2.3 of Chapter 9, where it was identified as the "iterative" extension. If -ken were absent in the examples with plural subjects, this would mean that the affected persons all fell at once, i.e. as a group. Thus, a sentence such as bố gwế lồ 'they have just fallen' would be appropriate if the falling people had been tied together and then fell.

Finally, [-PROG, -FUT, -FOC] verb forms require a verbal complement. A sentence such as *bɔ́ gwe is ungrammatical. For this reason the verbal focus marker lò is required in the above examples. In the [+FOC] variants, however, the verb can occur finally, e.g. bɔ́ nɔ́ɔ̀ gwè 'they have fallen'.

3. The direct object relation. The direct object (DO) occurs immediately after the verb and is not marked by any preposition, e.g.

```
bố nổô khủn wàn 'they have hit a child' (Po) yunsè cỉ nổô dèè kèngôm 'Yungse cooked plantains' (P2) me nú mbổy wố mbvum 'I will drink wine' (F2)
```

The object pronouns are identical to the subject pronouns as enumerated in

Chapter 4, e.g.

```
bổ gồmtế me 'they are looking for me'
bổ gồmtế wò 'they are looking for you [sg]'
bổ gồmtế bèsen bổ gồmtế bèn 'they are looking for us [excl]'
bổ gồmtế bèn 'they are looking for you [pl]'
bổ gồmtế bổ 'they are looking for them'
```

It was seen in Chapter 9, section 3.4, that an extension of the shape -Cen is sometimes used to express reciprocal actions, e.g.

```
bố tenéné lò 'they are calling each other' (cf. téé 'call')
bố kồněné lò 'they love each other' (cf. kon 'love')
```

This extension is more rarely used for reflexive actions. Instead, the nouns ye 'body' and ϵ kol ϵ 'head' (pl. bikoo) appear in object position for this purpose:

```
bố yếnế ye bố lễ 'they see themselves' (lit. they see their bodies, cl. 10)
bố kwààkể bikoo bibốốiể 'they are thinking of themselves (lit. they are thinking of their heads, cl. 8)
```

Thus, there is no reflexive pronoun in Noni.

4. Indirect objects. There are a number of ways in which Noni can express recipient and benefactive relations to the verb. Although these are typically referred to as "indirect objects" in English, as in "John bought Mary a new car" (where "Mary" would be an indirect object), we shall use this term to refer only to such objects which, as in the English example, do not take a preposition. To continue the parallel, "Mary" would not be an indirect object, but rather an "oblique" object in the sentence "John bought a new car for Mary".

Defining indirect objects in this way, there are relatively few instances of a verb taking two noun complements, neither of which requires a preposition. Some examples are as follows:

```
me nóò ngòy wan bènkfu me nóò ngèsè wan bènkfu me nóò ngèsè wan bènkfu me nóò na wan bènkfu me nóò ncìinsè wan bènkfu me nóò ncìinsè wan bènkfu me nóò nlèm wan wé 'I have sent yams for the child ['s benefit]' me nóò nlèm wan wé 'I have sent yams for the child [to eat]' il have sent yams for the child [to eat]' il have sent yams for the child [to eat]' il have sent yams for the child [to eat]' il have sent yams for the child [to eat]' il have sent yams for the child [to eat]'
```

As can be seen in the translations, indirect object $\widetilde{\text{wan}}$ 'child' in these examples represents the person for whose benefit an action is carried out. This notion of "benefactee" can occasionally be realized also by the preposition ε , e.g.

```
me nóò ndèè wàn bènkfǔ 'I have cooked yams for the child [to eat]' me nóò ndèè bènkfǔ ɛ wan (= same)
```

These two sentences both indicate that the cooking of yams took place for the child's benefit. Normally, however, ϵ indicates a directional recipient. Thus, contrast the sentence 'I have sent yams for the child [to eat]', given above, with the following:

me nɔɔ́ɔ ncììnsè bènkfǔ ε wan 'I have sent yams to the child'

Also, the meaning of 'for someone's benefit' is expressable through a possessive construction, e.g.

me nóò nlèm wế nfòn 'I have cultivated the field for the chief ['s benefit]' (lit. 'the field of the chief')

As we shall see in the next chapter, the postposition | & unambiguously indicates 'for' in the sense of 'in the place of', e.g.

me nổờ ndềt bènkfu wan é 'I have cooked yams for the child' (i.e. in his place)

Thus, to summarize, we find the following:

 $V \quad O_2 \quad O_1$: ' O_1 for O_2 's benefit'

V 0_1 ϵ 0_2 : ' 0_1 to/from 0_2 ' (occasionally 'for 0_2 's benefit')

 $V \quad O_1 \quad O_2 \quad \text{i.e.} \quad : \quad O_1 \text{ for (in the place of) } O_2'$

 $V O_1$ 'of' O_2 : 'O₁ for O_2 's benefit'

We have omitted from this summary the function of b5 'with', which was seen with the verb 'give' at the beginning of this chapter. The different uses of this preposition will be seen in Chapter 17.

OBLIQUE AND ADVERBIAL COMPLEMENTS

1. Definitions. In the preceding chapter we distinguished between direct and indirect objects. However, so-called indirect objects were seen to be of at least two types: (a) a noun or noun phrase appearing directly after the verb without a preposition or postposition; or (b) a noun or noun phrase appearing after a potentially present direct object and marked by either the preposition ϵ or the postposition $|\epsilon|$. Because of the similar semantic interpretation of 'child' in such sentence pairs as,

me nóò ndèè wàn bèŋfkǔ me nóò ndèè bèŋkfǔ ε wan

both of which mean 'I have cooked yams for the child's benefit', it is tempting to identify both instances of 'child' as "indirect objects". In this semantic approach the difference between an unmarked and a prepositionally marked noun (phrase) is ignored.

In this chapter we shall consider only the latter type of noun phrase, i.e. one marked with either a preposition such as ϵ , or by a postposition. The two interpretations of the first sentence are that it is a sentence with the structure S AUX V IO DO, or a sentence with the structure S AUX V DO DO. In the first analysis the word order given as basic at the beginning of Chapter 16 is violated; in the second analysis, two undifferentiated DO's are postulated. While we tend toward the second analysis, what will be critical for the present discussion is simply whether a noun phrase is obtique or not. It will be considered oblique in just the cases where a preposition or postposition is required. After exemplifying the major oblique relations recognized by Noni, we shall consider the various adverbials and particles which occur in post-verbal position in the sentence.

2. The preposition ϵ . We have already seen a number of instances of this preposition. Most of the examples translate ϵ as 'to'. Depending on the semantics of the verb, however, this preposition can mean 'from':

me ncoonké kèngòm ɛ kwoòn 'I am stealing plantains from the woman' 'they are hiding plantains from the woman' bố fối kèngòm ɛ kwoòn 'they are receiving plantains from the woman' 'they are buying yams from the woman'

The last sentence is ambiguous and can also mean 'they are buying yams for the woman's benefit'. Although a can indicate 'to' and 'from' it is, strictly speaking, not a directional preposition. Thus, a sentence such as the first above does not necessarily mean that you are stealing plantains from the place where the woman is, or from her person. For this purpose the preposition for is used (see below).

3. The postposition |é. This postposition has the meaning 'for' in the sense of 'in the place of' or 'in someone's stead'. It is realized |é after a vowel, é after a consonant. It is most definitely related to locative |é (see p. 14 and below), which may appear with other prepositions or alone. In addition to the 'for' interpretation of this postposition, the

verbs yén 'see', bi 'follow', gay 'greet', fil 'be tired of', kom 'touch' and ciné 'leave' require lé after their complement:

```
me nóà nyến wè lẻ 'I have seen someone'
me nóà mbì wàn é 'I have followed the child'
me nóà ngày kwoàn ẻ 'I have greeted the woman'
me nóà nhòm fysè lẻ 'I have touched something'
me nóà ncinè wàn é 'I have left the child'
```

But for this postposition the nouns we 'person', wan 'child', kwoon 'woman' and fyee 'thing' would be direct objects in the above sentences. Perhaps the lé marking indicates a kind of indirect object that is lexically conditioned by the verb. Thus, one 'sees to someone', 'touches to something' etc.

4. The preposition b5. This preposition was already seen in the sentence b5 c1 nd boom b5 kengem fogo 'they gave plantains to the children yesterday', on p.76. Its literal meaning is 'with'. Thus, in this sentence, one 'gives the children with plantains' (cf. English 'provide someone with something'). In the following sentences it is noted that b5 is used with directional verbs (e.g. t5 'come') to mean 'take/bring something to':

```
me nóò ntỏ bỏ bènkfǔ ε wò 'I have brought yams to you'
me nóò ntỏ wò bỏ bènkfǔ 'I have brought you yams' (for your benefit)
me nóò ntỏ bỏ bènkfǔ wò lễ 'I have brought yams for you' (in your place)
```

The prepositional phrase bố bènkfủ 'with yams' co-occurs with the three-way opposition we saw on p.79.

This use of bo may be termed 'comitative'. In conjoining two humans, the following distinct possibilities have been noted:

```
me ntổ bố wàn 'I am bringing the child' me ntổ beè wàn tổ là 'I am coming with the child' the child and I are coming'
```

The first sentence indicates that I am carrying the child, i.e. that he is not walking along side of me. The second sentence indicates that I am leading, but that the child is walking on his own. Finally, the last sentence represents a true coordination of two noun phrases: the child and I are coming as equals. It is noted that the second and third sentences employ compound pronouns as described in Chapter 4. Thus, perhaps bo 'with' comes from bo 'they'.

The other major use of $b\delta$ is in an instrumental sense, as seen in the following sentence:

me nóò nsès ñàm bố fèno 'I have cut meat with a knife'

When the comitative or instrumental phrase involves a potential pronoun, the form $d \lor u$ substitutes for $b \circlearrowleft d + a$ noun:

```
me nóò nsès ñàm dvú 'I have cut meat with it'
me nóò ñà wvù dvǔ 'I have given him it'
```

There is no corresponding form for human referents. The form dvú is of course related to locative class 18.

- 5. Locative complements. We examined the locative classes 16-18 on pp. 13-14. These and other locative constructions are further illustrated in the following subsections.
- 5.1. Class 16. This locative class involves the locative prefix f3-'to, at', which occasions the appropriate noun class agreements seen below:

```
fòway fome 'at my market'
fòway fɛn 'at this market'
fòyewe fo wan 'at the compound/house of the child'
fòfa fo kwoon 'on the woman's head'
```

The first two examples, fòway 'at the market' and fòyewé 'at the compound, at the house', involve locativized nouns which exist in other noun classes, cf. way 'market' 3/13, yew 'house' 9/10. The form fòfa 'on the head', on the other hand, exists only in class 16 (cf. also fòfalaan 'at the summit [of a tree, mountain etc.]', for which no corresponding non-locative noun has been found).

As seen, the normal meaning of class 16 fb- is 'to, at'. With this meaning, these locatives often occur as complements:

wvù tooté ken fàko 'he is throwing firewood to/on the ground' bố nổờ tồm nwà fò wàn é 'they have sent the book to the child'

It is conceivable in the second sentence that the locativized noun could mean 'for the child'. Compare the following synonymous sentences, one of which uses $f\dot{b}$ -, the other $\dot{\epsilon}$:

bố nổờ tồm wan fố byudyểu bwêw ể 'they have sent the child in his bố nốờ tồm wan ế byudyểu bwêw ể place'

Both prepositional phrases (written here with the preposition fò and postposition lé as separate words) indicate 'for him' in the sense of 'in his stead'. It is interesting to note that while fò is used to indicate direction toward a person, it may not required with a place name, e.g.

bố nổờ tồm wan ŋkờ 'they have sent the child to Nkor'

Finally, while the preposition usually indicates direction toward something or someone, it acquires the meaning 'from' after certain verbs:

me nóò nè fò wàn é 'I have left the child' wvù nóò lewtè fòwày 'he has rum from in the market'

5.2. Class 17. Class 17 locatives usually begin with ϵ - (with either L or M tone). Some examples with modifiers are as follows:

```
ɛjè jan 'in this road' (jè 'road' 9/10)
ɛ̀ntě jɔ fìɛ̀w 'inside the rat' (keté 'stomach' 7/8)
ɛ̂jɔ́ɔ̀ jɔme 'in my stream' (jɔ̀ɔ̀ 'water, stream' 9/10)
```

These examples all show a preposition ϵ 'in' and, as indicated, derive from nouns found in other classes. In some cases no corresponding noun is known:

```
èkǔ 'under'
èmuù 'in the mouth'
```

In other cases the base noun is known, but the derivation is irregular. Thus, in addition to Entě 'inside', which derives from the class 7 variant ϵ té of keté 'stomach' (see p.10), we have the following:

```
èkê 'in the bag' (εκειὲ 'bag' 7/8)
èkôw 'in the bush' (εκόwε 'bush' 7/8)
```

These forms both lose their final $-\epsilon$ in forming a class 17 locative (as well as undergoing tonal changes, which mark class 17 formation in other cases we have seen).

Examples of class 17 locatives appearing as verb complements are as follows:

```
bố nổờ nà bvulem ềmuủ jo wan 'they have given (put) fufu in the child's mouth'
bố nổờ bốy wâ ềwấy jan 'they have already come out of this market'
```

5.3. Class 18. Only three locative class 18 forms have been found in the language. These are illustrated below with modifiers:

```
yếw dvume 'in my house' (yếw 'house' 9/10)
con dvun 'in this hut' (còn 'hut' 9/10)
cin dvu kwoòn 'around the woman's waist'
```

Two of the three class 18 forms have corresponding non-locative counterparts, as indicated.

While class 18 locatives are only three in number, their pronominal form $dv\acute{u}$ is used widely as a replacive for $b\acute{o}$ 'with' + a noun, e.g.

wan wvù me ñanyî bố kènkfǔ 'the child to whom I am feeding with yams' kènkfǔ kè me ñanyî wan dvú 'the yams I am feeding the child (with)'

The meaning of the second relative clause is literally 'the yams that I am feeding the child with them' (cf. section 4).

5.4. Other locatives. The two most frequent locative constructions are `c...le' in' and yi...le' ion'. We have seen instances of the former in section 5.2. Not all such prepositional phrases are instances of class 17, however. Instead, an option is often observed as in the following:

```
è gweè jo wvù 'in his ceiling' (gweè 'ceiling' 3/4)
è gweè wvù lě """"
```

Either $\grave{\epsilon}$ is used alone and in class 17, or it is used in combination with the postposition $l\acute{e}$, and the locativized noun remains in its original class. Additional examples of locative complements with $\grave{\epsilon}...l\acute{e}$ include:

```
me mbék mbìsè nkàw k mbàsè lẻ 'I added salt to the soup' (P1)
yoù yin nú k nkàn é 'these tomatoes are about to be fried'
yoù yin cí nú k ndèk lẻ 'these tomatoes were to be cooked'
```

The last two examples show that verbs can be placed in the £...lé construction to derive a stative verbal form (lit. 'these tomatoes are in frying', etc.). As seen in the following additional examples, a nasal prefix is also required:

```
(ngáy 'break')
           'to be broken'
lycpná
                                (sho 'clean')
           'to be cleaned'
Enshale
                                (foo 'borrow')
           'to be borrowed'
ènfààlě
                                (gaw 'divide')
           'to be divided'
èngàw!ě
                                (yoy 'twist')
           'to be twisted'
Ènvovi
                                (bom 'mould')
           'to be moulded'
emcdmá
                                (ton 'refuse')
           'to be refused'
èntoné
                                (veně 'open')
ènyènèlě
           'to be opened'
                                (hile 'hide')
            'to be hidden'
èñilèlě
                                (kawyé 'chop')
            'to be chopped'
ènkawyèlě
                                (tfúsé 'retaliate')
            'to be retaliated'
èntfusèlě
            'to be given birth' (boo 'give birth')
èmboole
```

As indicated, it seems most appropriate to write these forms as one word.

The second common locative is yi...lé which has the functions indicated in the following examples:

```
me nú ntó yi kèma kin é
wvù nóò gwè yi ste lé
wvù felé me yi bvudétè lé
me nóò ntòy yi wan é
"I will come during this week"
he has fallen from the tree'
he is taller than I'
I have accused the child'
```

The basic meaning of yi... 'e is 'on' as seen in the simple examples:

```
yí yèw é 'on the house'
yí εbɔ́ le 'on the hand'
```

However, the preceding sentences show that this construction can be used temporally (to indicate duration) as in 'I will come during this week' (cf. yi bilom le 'for years'). The second sentence can be interpreted as 'he has fallen from on the tree', while the third sentence literally means 'he surpasses me on height', cf. the following:

```
wvù felé me yi bvufánê le 'he is fatter than I' (bvufánê 'fatness') wvù felé me yi bvujítê le 'he is heavier than I' (bvujítê 'weight')
```

Finally, the verb toy 'accuse' appears to require yi... le and must be so marked in the lexicon.

Other locatives are usually derived from nouns or other locatives, e.g. Entě lén 'between' (from Entě 'inside' [from Eté 'stomach'] + lén '?'), èjim 'behind' (from jim 'back'; but cf. èjîm 'at the back'), etc.

- 6. Adverbs. There are a number of adverbs in Noni which occur after the verb, hence in complement position. We shall divide these into three categories: aspectual, temporal and manner adverbs.
- 6.1. Aspectual adverbs. The three adverbs considered to be in this category are wâ 'already', 'ò 'just, instead', and kê 'only'. These adverbs all occur between the verb and the first post-verbal complement (e.g. the direct object, if there is one). The following sentences illustrate the use of wâ 'already' with a variety of tenses:

```
ntòn nóò kwè wâ
yuŋsè cí bǐ wâ èjǐm fòwǎy
me ncí nóò mèsè wâ èndeè bènkfǔ
me ndèèlě wâ bènkfǔ
```

'the pot has (already) broken' (lit. 'died')
'Yungse followed (alr.) behind to market'
'I finished (alr.) cooking yams'
'I am (already) cooking yams'

These sentences are in the P_0 [+FOC], P_2 [-FOC], P_2 [+FOC] and 0 tenses, respectively, indicating that wa has a more general use than English 'already'.

The second aspectual adverb is the 13 we have seen in preceding chapters. If a [-FUT, -FOC] verb is not followed by a complement of some type, 13 is required, e.g.

```
bèsèn gwèé lò 'we are falling' [as a group]
me ntóó lò 'I am coming'
bèsèn gwě lò 'we have just fallen'
me ncí tó lò 'I just came, I came'
yuŋsè lewté wâ lò 'Yungse is already running'
```

The tense is 0 is the first two sentences, P_{O} [-FOC] in the third, and P_{O} [-FOC] in the fourth. The last sentence demonstrates that the adverb wâ does not count as a verbal complement, since this sentence would not be acceptable if 'à were not present. (The wâ is thus an expansion of the verb, rather than of the verb phrase, as we have also discovered for corresponding adverbs in Aghem.)

In the above sentences, we cannot delete 15 or else the remaining forms are ungrammatical. The 15 which is present thus either completes the [-FUT, -FOC] verb, or, places additional focus on the verb itself. In case there is a complement, e.g. an object, 15 is not required, e.g.

```
me nkfùně wan 'I am hitting a child' (0) me nkfùn wan 'I have hit a child' (P_0, [-FOC]) me ncî nkfùn wan 'I hit a child' (P_2, [-FOC])
```

In the 0 tense, a normal present tense sentence is obtained. In the [-FOC] P_0 , P_1 and P_2 , however, utterances without $|\dot{a}\rangle$ can only be in answer to a question, e.g. 'whom did you hit?'. They are made more independent, with focus on the verb, by including $|\dot{a}\rangle$, i.e.

```
me nkfun là wan 'I have just hit a child'
me nci nkfun là wan 'I hit a child'
```

If, on the other hand, 13 is placed in a sentence which could have stood alone without it, 13 adds the meaning 'just, instead', e.g.

```
me nkfùně là wan 'I am just hitting a child' (0)
me nú mbóy ngwe là wan 'I will instead hit a child' (F<sub>2</sub>)
```

It should be noted that where 13 is required to complete a sentence, this occurs only in main clause affirmatives, e.g. not in relative clauses.

The final adverb presented in this section is ké, which is used to mean 'only', occurring before a focused post-verbal constituent, e.g.

```
È nử kể me
'it is only I'
È lòó kể bố
'only they are weaving'
wvù cĩ ñá kể ɛ me màn
'he gave it but to me alone'
me nú ndiế kể ɛ nùù kèngòm màn 'I will eat but plantains alone'
```

The first two sentences have the subject postposed to the verb. The result is that this focused subject falls under the scope of ke 'only'. The third and fourth sentences also have man 'alone, only' ('one') which acts with ke.

As seen, ké can be translated as English 'but', e.g. 'it is but I', in many cases. This marker is clearly related to the ke found in negative constructions, e.g.

wvù cỉ ñá ŋwà ε nùù ε me, kế nùù ε wò kε 'he gave the book to me, not to you'

(The constructions just since with ϵ nùù... involve constituents which are focused; see Chapter 22.)

6.2. Temporal adverbs. Many of the referential time expressions resemble nouns, e.g. keséen 'now', nfèw '(in a) short time', ncèke 'a long time ago', ntfuu 'tomorrow'. Others resemble prepositional phrases of the locative type seen in earlier sections of this chapter, e.g. fogò 'yesterday', fò she ntfùu 'day after tomorrow' (lit. at spend-the-day tomorrow), efonefòn 'often'. The class I noun kan is used to indicate the number of times something has taken place, e.g. kàn mwàn 'once', bòkàn bofée 'twice', etc. All of these temporal expressions appear after the verb (and object, if any):

me mbέὲ ŋkfùncὲ wàn bòkàŋ bɔfέε me ntò nóò ntòn wan lòm béè vunsè nữ tố nfèw cààn vunsè tà nóò lewtè wâ ncèke

'I hit the child twice' 'I refused the child last year' 'Yungse will come in a short time' 'Yungse ran (already) a long time ago'

Very few, if any, temporal expressions appear to belong to a separate class of adverbs.

6.3. Manner adverbs. The argument can also be made for manner adverbs that most resemble nouns. Typical manner adverbs are illustrated below:

yunsè ci nóò yèm ntay vunsè ci nóò yèm fó ntèèn vunsè cĭ nóò lewtè weè yunsè cĭ nóò lewtè caàn yunsè ci nóò lewtè cèkèy já wě

'Yungse sang well' 'Yungse sang much' 'Yungse ran slowly' 'Yungse ran a little' 'Yungse ran fast' 'stand up!'

Of the manner adverbs in the above sentences, only we's 'slowly' shows a relationship to an existing non-adverb, namely wee 'be weak'. Other forms which may belong to an adverb class include ban 'please', nen 'thus', and $t\overline{\epsilon}h$ 'also', as seen below:

jîcê wan é ban me nge nèè ban bố nổo diekèn tến

'hold the child, please!'
'what can I do!' (exclamation) 'they have eaten too' wò dèèle lá fì wò jimténé nen 'what are you cooking that you are perspiring like this/thus?'

AUXILIARY VERBS

1. Verbal criteria. The Noni language is rich in auxiliaries which precede the main verb. A particularly complex sentence containing several auxiliaries is the following:

me ntò nge mbéè mbvů nyúú nkfun wan ke 'I had still not ever I P_3 do still again P_3 hit child NEG hit the child'

In this sentence there are five auxiliaries (with first person singular nasal agreement): ntò, nge, mbéè, mbvů and nyúú. These auxiliaries are verbs, since they meet the following criteria for verbal status:

- (a) They have the same segmental and tonal structure as main verbs, i.e. they begin with a consonant and have at least one vowel, and they take the tones typical of one or another tense.
- (b) They take first person singular subject agreement, as seen in the nasal prefix on each of the five auxiliaries in the preceding example.
- (c) They appear in second position (after a main verb), e.g. as a consecutive.

As an example of criterion (c), consider the following sentences:

me ngòmtě ngéé nyến wò lễ bbbn 'I want to see you today'
me ngòmtě mbóy nyến wò lễ ntfùú 'I want to see you tomorrow'

The first clause me ngòmtě...'I want' is in 0 tense. The second clause begins with the F_1 auxiliary géé in the first sentence and the F_2 auxiliary bóy in the second. The difference in meaning is as indicated: the seeing will take place today vs. tomorrow. If these forms were not verbs, we would not expect them to appear in this position.

In this chapter we shall look at the various types of auxiliary verbs found in Noni. We shall demonstrate that most of them are related to still existing main verbs.

2. Tense auxiliaries. The following tense auxiliaries have been observed in preceding chapters (see especially Chapter 12):

form	auxiliary meaning	main verb meaning (if any)
bé	P ₁	?
cí tà	P ₂ P ₃	('cé 'spend the day' ?)
nú	[+FUT]	(to 'come' ?)
g éé bóv	F ₁	(<ge 'do'="" ?)<="" td=""></ge>
bốy yửů nóò	F ₂ F ₇	'exit, reach' 'to dawn'
nóò	F_ [+F0c]	nú 'be' + ε

The future tense markers have a more transparent derivation than their past tense counterparts. (This is a general fact for the whole Grassfields Bantu region.) It should be recalled that P3 to can also be used in the F3 and

that F₃ yuu can also be used in the P₃ (see p.62).

- 3. The copula 'be'. The copula verb 'be' has a number of main verb and auxiliary functions. It occurs in infinitive form as kèé-nùù 'to be', although in many contexts it acts like a H tone verb and has a short vowel. (The form kèé-nùù is, of course, [+PROG].) We shall cite the copula as nú, since this is the most common form it takes.
- 3.1. 'Be' as a main verb. As seen in the following examples, nú can be followed (a) by a noun; (b) by an adjective; (c) by a locative; and (d) by a prepositional verb form:

è nǔ me
yunsè nǔ cowè
nwà wvun nú fwɛ
yunsè nǔ yèw
yon yin cí nú è nkan é
mú nú muféém èkê
'it is I'
'Yungse is an adult'
'this book is new'
'Yungse is in the house'
'these tomatoes were to be fried'
'they are two [e.g. kolanuts] in the bag'

Thus, unlike other African languages, Noni has only one copula which is used in all of the above contexts.

3.2. 'Be' as an auxiliary. There are at least three distinct functions of nú as an auxiliary. The first of these, the marking of [+FUT] tenses, has been illustrated several times. Similarly, its appearance in the nób marker in [-PROG, -FUT, +FOC] tenses (except the P_1) has been noted, as has its variant nú+ ϵ , e.g.

me nóò ngwe = me nú ε ngwè 'I have fallen' [+FOC]

The copula is much more involved in focus marking in Noni than the above [+FOC] P_0 tense would indicate (cf. Chapter 22). Thus, to mark verb complements as focused, they are preceded by the markers ϵ nùù, e.g.

me nyến ε nùù wàn é εjè 'I have seen a *child* in the road' wvù cĩ ñá bìnkfǔ ε nùù ε me' 'he gave the yams to me'

In this construction the long vowel (=[+PROG]) variant of nú is used. Because of the marker nú, one might wonder if this is a sequential construction (Chapter 20), i.e. 'he gave the yams and it is to me'. The following sentences would then be the [-PROG] sequential counterpart to the above:

me ncí nóò nkèw wâ ɛ nú ndèèlè bìnkfǔ 'I started cooking yams'
me ncí nóò mèsè wâ ɛ nú ndèèlè bìnkfǔ 'I finished (sth.) and was cooking
bònwà boo nú bēm = bònwà bo ɛ nú bēm 'the books which are mine' yams'

These usages may have something to do with focus, since emphasis is on the clause with ϵ nú (or on the bem in the relative clause example, i.e. the books which are mine, my books).

The sentences with 'start' and 'finish' above hint to the last major function of $n\hat{u}$ as an auxiliary verb: it frequently marks progressive actions:

wvù nóò nử jewtè 'he has been becoming mad'
wvù nóò nử yemtè binè 'he has been singing and dancing'
me nứ wâ ntố lé 'I am already about to come'
me mbé nù wâ èntố lé 'I was already coming [when]...'

In the first two sentences nu appears in the P_0 [+FOC] tense, acquiring a perfect progressive meaning. The third sentence resembles the F_0 sense of nu, except for the postposition le and the tone pattern A on the verb. Finally, the last example has the $\grave{\epsilon}$ + verb + le construction seen on p.84. The sense of this last sentence thus is 'I was in coming [when something happened]'.

4. Habitual aspect. The habitual aspect is formed with the auxiliary geε (cf. geε 'put, let, store'), as seen in the following sentences:

```
me ngiế ngwề 'I fall'
yunsè cỉ giế nhù wan
bố tổ giế dèlè kèngòm
bố nữ yữu giế khunề boom 'they will always be hitting children'
```

As seen in these sentences, the main verb is always in a [+PROG] form. It is interesting to note in the first sentence that the focus adverbial 13 is not required to make the sentence complete. Although this sentence is unambiguously habitual in its meaning, the corresponding sentence with the 0 tense, i.e. me ngwèé 13, can either meaning 'I am falling' or 'I fall'.

5. Other time and manner auxiliaries. Although we shall not set up each one as a separate tense or aspect, there are many additional auxiliary verbs which refer to the time or manner in which an action takes place. A summary is provided below, followed by examples:

main verb meaning

```
shéé
                    'later'
                                          'remain'
      cé
                 'in the day'
                                       'spend the day'
     shé
                'in the night'
                                      'spend the night'
   bóó/bvú
                    'again'
                                      ( bvu 'interrupt'?)
    yā̀nsen
                   'quickly'
                                       'to be early'
                 'off and on'
      .já
                                       'leave, stand'
    moòcĕ
                    'often'
                                      'stay for short while'
    baynέn
                   'rarely'
                                          'be rare'
     yaw
                  'be first'
                                         'be first'
e.g. me nú nshéé ntá
                                 'I will come later'
     me nú ncé nkfun wvù
                                 'I will hit him tonight'
     me mbé nshe nyemtè mbinè
                                 'I was singing and dancing all day'
     yunsè bòó lewté là
                                 'Yungse is running again'
     kế yunsè nùủ yaŋ̀sĕn ke
                                 'Yungse will not come quickly'
     yunsè ci nóò jà èlewtè
                                 'Yungse ran off and on'
     yunsè mòòci kế tɔ̈
                                 'Yungse often comes' (stays short but comes)
     yunsè gé baynèn là số tá
                                 'Yungse rarely comes' (does rarely before coming)
     è cí yàw yunsè ègēn fòway
                                 'Yungse was the first to go to the market'
```

It will be noted that the main verb is in one or another consecutive verb form (with tone pattern A, C or D). Some auxiliaries, on the other hand, either can or must take an infinitive, e.g.

```
wvù gòmtě (kèé) tó 'he wants to come'
wvù fàànkě (kèé) tó 'he refuses to come'
me nkòònkě (kèé) tó 'I like to come'
me nkenké kèé ne 'I have to leave'
```

auxiliary meaning

form

The last auxiliary verbs to be considered may ultimately be considered part of the aspectual and mood systems. In the following sentences it is seen that the auxiliary be(ϵ) means 'still':

yunse béé lewté là 'Yungse is still running' bó béé djekéné là 'they are still eating'

The negative is formed with bá(á), e.g.

yunsè bàá lewtè wâ kɛ 'Yungse is not running yet' bá báá diekènè wâ kɛ 'they are not eating yet'

The presence of wâ 'already' is noted in these negatives. The form bá(á) was seen to be the [-PROG, -FUT] negative marker in Chapter 12. The fact that $b\hat{e}(\hat{\epsilon})$ exists (with the same vowel length alternation) suggests that $b\hat{a}(\hat{a})$ is a contraction of $b\hat{e}(\hat{\epsilon})$ + ya.

In addition to b $\ell(\ell)$, the auxiliary ge (from ge 'do') can be used as part of the verbal focus system. Representative uses of ge follow:

me mbá ngé ngwe ka 'I have not gone and fallen' wò bế gể khữn wvù njẽ là 'why did you go and hit him?'

The first sentence asserts that I have not fallen, while someone apparently had thought I did. As seen in the translations, this use of ge seems aptly captured by the colloquial 'to go and do something'.

It is interesting to note that bé(é) 'still' closely resembles the P1 [+FOC] auxiliary béè, while gé 'go and do something' closely resembles the F1 future auxiliary géé. These should not be confused, however.

The last auxiliaries to be considered here are the conditional markers bee and nee:

me mbèé (nèé) ηgwe ε wò (nɛn) bé baa wám 'I would have fallen if you had not screamed' (P1) ε me ncí mbèé ηkεηκὲ begēw to me ncí nóò 'if I had had the money, I would have gone to market' (P2)

By themselves, sentences like me mbbé ngwe can mean 'I would have fallen', 'I almost fell' and 'I am about to fall' (i.e. I am almost falling). The additional marker nee is optional in the above sentences, although it adds greater doubt to the reality of the condition. The parenthesized word nen in the second clause of the first sentence means 'thus'.

PART V: COMPLEX SENTENCES

19

RELATIVE CLAUSES

1. Subject relatives. All relative clauses are formed in Noni by placing the relativized noun at the head of the clause followed immediately by a relative pronoun. This pronoun is segmentally identical to the pronouns described in Chapter 4. When the relativized noun is subject of its clause, the relative pronoun takes LM no matter what the noun class of the head noun, e.g.

wan wvu gwě 'the child who has fallen' bom bo gwèkěné 'the children who have fallen' bwě yl conké 'a dog that steals'

As we saw in Chapter 13, there are restrictions on the use of main clause tenses in relative clauses, and in some cases, the relative tenses differ from their main clause counterparts (see section 5). For example, in the example 'a dog that steals', which uses the 0 tense, the verbal focus marker lò is not required to complete the clause as it was in the main clause.

The relative construction and relative pronoun remain identical when the verb has a complement and when it occurs in the negative:

kwoòn wyu cí kfùnè wan 'the woman who hit the child' boñil bogéé goytè bìnkfu' 'the people who buy yams' ñam yl nú fô ya boy lom boom ke 'the animals that will not bite children'

The resulting clause can, of course, be used as subject of a larger clause (e.g. 'the woman who hit the child fell') or as object (e.g. 'I saw the woman who hit the child that fell'), and also as an oblique (e.g. 'I gave the yams to the woman who hit the child').

2. Object relatives. All non-subject relativized nouns take a L tone relative pronoun instead of a LM one. This is illustrated with the following clauses where the object has been relativized:

kwoòn wvù bèsèn cỉ kfùnè 'the woman that we hit' bìnkfǔ bì boñil géé goytè 'the yams that people buy' boom bò nam nú fô ya bóy lóm ke 'the children that the animals will not bite'

As seen, there is no resumptive object pronoun in the relative clause (i.e. one does not say 'the woman that we hit her').

3. Oblique relatives. In all other cases the relative pronoun remains L in tone, but a resumptive pronoun is required in the relative clause. Thus, consider the following relative clauses involving, respectively, the preposition ϵ 'to', the preposition bó 'with' and a possessive:

kwoòn wvù me ntomté wan ɛ wvù 'the woman that I am sending a child to' boom bò me ngèně fòway beèbɔ 'the children that I am going to market feñii fè me nci nyɛnè kebɛbè kêw è 'the bird whose wing I saw' with'

These relative clauses have the following structure: 'the woman that I am sending a child to her', 'the children that I am going to the market with them', and 'the bird that I saw its wing'. (Note in this last sentence that the possessive pronoun can refer to non-human referent, although it must be animate.)

The locative classes have two possible relative clause structures. First, they can follow the direct object pattern:

fàway fà me ngắn 'at the market where I went' bbả jà me nci ngèèlè fèmbì 'in the hand where I put the kolanut' yèw dvù me nci ncelè 'in the house where I spent the night'

The relative pronouns fò (class 16), jò (class 17) and dvù (class 18) are used exactly as any other non-subject relative pronoun, and there is no resumptive pronoun in the relative clause. On the other hand, the following is possible:

wấy wvù me ngiến fố 'the market that I went to'
ɛbố kè me nơi ngiêlê fèmbì jố 'the hand that I put the kolanut in'
yèw yì me nơi ncelê dvů 'the house that I spent the night in'

In this case the basic noun is relativized (with class 3, 7 and 9 concords, respectively) and the locative class pronoun appears resumptively in the relative clause. Thus, f5, j5 and dv4 mean 'to it', 'in it', etc.

This recalls the use of dvú as a replacive for b5 prepositional phrases, where the head noun is inanimate (see Chapter 17, section 4). As seen in the following examples, dvú appears in relative clauses as a resumptive pronoun meaning 'with it':

kènkfǔ kè me ñanyí wan dvú 'the yams I am feeding the child' (cf. me ñanyí wan bố kènkfǔ 'I am feeding the child yams')
fènò fè me nsɛɛlé ñàm dvú 'the knife I am cutting meat with' (cf. me nsɛɛlé ñàm bố fènò 'I am cutting meat with a knife')

The relative clauses thus literally mean 'the yams that I am feeding the child with them' and 'the knife that I am cutting meat with it'.

4. Uses of relative clauses. The primary purpose of a relative clause is to identify a referent. What is included in the relative clause is usually shared knowledge between the speaker and hearer. Thus, when one says 'where is the child who fell?', one assumes that the hearer knows that there is a child who fell. (but see section 5). As seen in the following example, where the head of the relative clause is a proper name,

yunsè, wvū cí gwèlè, nu wâ síin fò yèw é 'Yungse, who fell, is already now at home'

As seen from the pauses marked by commas, this kind of relative clause adds information that the hearer may not have shared with the speaker. Noni apparently can therefore allow such "non-restrictive" relative clauses.

It will be recalled from constructions seen in Chapter 7 that relative clauses take the place of what would be adjectives is English:

```
wan wvu fàn 'a fat child' (fàn 'become fat')
boom bo fànkèn
nkèm wvu jitè 'fat children'
'a heavy load' (jitè 'be heavy')
fèmbì fè yeelé 'a red kolanut' (yeelè 'be red')
kwoon wvu dèfě 'a tall woman' (dèfè 'be tall')
```

As seen, the LM subject relative pronouns are used.

Another use of the relative construction is in expressing indirect questions, e.g. $% \left(1\right) =\left(1\right) \left(1\right) +\left(1\right) \left(1\right) \left(1\right) +\left(1\right) \left(1\right) \left($

```
kế me ŋkeè wè wvù wò bế yẽn ke 'I don't know whom you saw'
kế me ŋkeè fysè fè wò bế yẽn ke 'I don't know what you saw'
kế me ŋkeè bvudvủu bvù wò gèně fó ke 'I don't know where you are going'
kế me ŋkeè wan wvu gèně ke 'I don't know which child is going'
```

Literally, these sentences mean 'I don't know the person you saw', 'I don't know the thing you saw', 'I don't know the place you are going to it' and 'I don't know the child who is going'. The above sentences thus have the other interpretation that a specific person/thing/place is referred to.

In addition, the following indirect questions have also been recorded:

```
kế me ŋkeè fò wò gèně ke 'I don't know where you are going' kế me ŋkeè sêkè wò gèně ke 'I don't know when you are going' kế me ŋkeè nù wò nữ gen ke 'I don't know how you will go'
```

In the first sentence fò is of course the object relative pronoun for class 16. This sentence thus means 'I don't know to where you are going'. The forms sékè and nù seen in the following two sentences are probably also nouns, although they were not observed with a relative pronoun. (sékè also occurs in the corresponding direct question; see Chapter 22).

- 5. Relative tenses. The tense structure of relative clauses was presented in Chapter 13. There it was seen that a number of tenses have a different morphology in relative clauses from that observed in main clauses. These differences can be seen also in the table on p.73 and are summarized again as follows:
- (a) In the [-PROG, -NEG, -FOC] $\text{P}_1,$ the relative form takes an additional L tone suffix - $\dot{\text{V}}.$
- (b) In the [-PROG, -NEG, -FOC] P_2 and P_3 , the relative form takes an additional suffix -1 $\hat{\epsilon}$, thereby changing from tone pattern B to tone pattern A.
- (c) In the [+PROG, +NEG] O tense, the tone pattern is A in main clauses, but B in relative clauses.
- (d) As indicated above, 13 is not required in relative clauses to complete the affirmative 0 and [-FOC] $P_0\!-\!P_3$ tenses.

Another interesting tense restriction concerns the [+FOC] variants of the P_0 - P_3 tenses. When the head of the relative clause is specific (i.e. nongeneric) in meaning, i.e. the speaker has a specific referent in mind, the [+FOC] variants of these tenses cannot be used in a relative clause. Consider, however, the following pairs of sentences:

```
ta wan wv\overline{u} cí gwèlè 'look for the child who fell' (P2, [-FOC]) ta wan wv\overline{u} cí nóò gwè 'look for a(ny) child who fell' (P2, [+FOC])
```

kế me nkeè wàn wyù gwẻ kẽ 'I don't know the child who has fallen' kế me nkeè wàn wyù nóò gwè kẽ 'I don't know any child who has fallen'

(The second pair involves the P_0 [-FOC] and [+FOC] tenses, respectively.)

The expected or most frequent relative construction is with the first, or [-FOC] variant. In the first sentence the speaker, who assumes that the hearer knows that a child fell, tells the latter to look for this child. In the second sentence, however, the speaker tells the hearer to look for any child who may have fallen. The speaker and hearer do not presuppose that any child fell at all. In this second sentence the relative clause contains information not previously assumed by the hearer. It is for this reason that the [+FOC] variant is used. The second set of sentences works similarly: in the [-FOC] variant it is mutually accepted by speaker and hearer that a child has fallen. In the [+FOC] variant, on the other hand, it has not been assumed that any child has fallen.

The same distinction is found even when the head of the relative clause is a proper noun. Thus, compare the first sentence in section 4 with the following:

yunsè wvù cí nóò gwè nu wâ síin fò yèw é 'any Yungse who fell/did fall is not at home'

This sentence requires a very special context in order for it to be acceptable. The preceding context must be that someone claims there is a person named Yungse, that he is at home, and that he fell. The above sentence denies the entirety of this context: there is no Yungse such that he fell and is now at home.

The use of [+FOC] tenses in relative clauses is important in determining the exact meaning of the feature specification [+FOC] itself. In related Grassfields Bantu languages (e.g. Aghem; see Watters 1979), the [+FOC] tenses cannot appear in relative clauses. Presumably their function is slightly different from that found in Noni (see also Chapter 22).

SEQUENTIAL AND SUBJUNCTIVE CLAUSES

1. Conjunction in Noni. While noun phrases can be conjoined with bo 'and, with', verb phrases and entire sentences are not conjoinable by a simple conjunction 'and'. As seen in the following two sentences,

me ncí nóò ngèn fòwăy, gèènen e me ncí mba ngoy bènkfű ke'I went to market, but I didn't buy yams'

me ngòmtě kèé ngoy bènkfű, gèènan ké me nkankè begew ka 'I want to buy yams, but I don't have any money'

the conjunction geens 'but' can introduce a second main clause--or even an imperative, as in the sentence:

gènế fòway, gèènen fò wố goy fysè ke 'go to market, but don't buy anything'

In most cases, however, the second clause must be in a special, "sequential" form if the subjects of the two clauses are not identical. (If they are identical, the second verb will take a consecutive form; see Chapter 14.) This even includes many instances of complement clauses. Thus, compare the following two sentences:

wvù dòó le me ncí nóò ngèn fòwáy 'he says that I went to market' me mbéè nyèn wan é e wvù (e) gwè 'I saw a child fall'

Verbs of reporting and knowing use the complementizer | E 'that' followed by a main clause, as in the first example. Other types of complementation, such as that following a sensory verb used to report some perception rather than knowledge (e.g. 'I saw a child fall' vs. 'I saw he was right') employ sequential forms. Thus, the Ø form of the verb gwe 'fall' in the second example has up to now not been seen with tone pattern A except after [+FOC] noo. As will be seen in the following section, this is the form of a sequential verb when it is [-PROG, -NEG] and follows a Po, Pl or P2 in the main clause.

2. [-PROG] sequentials. We shall consider affirmative [-FUT] sequentials first. As seen in the following forms,

me nóò nyem, wan ε kfùn wè 'I have sung, and a child has hit someone' me nóò nyem, wan ε ton wè 'I have sung, and a child has refused someone' me nóò nyem, wan ε dòmsè wè 'I have sung, and a child has pushed someone' me nóò nyem, wan ε cimsè wè 'I have sung, and a child has pacified someone'

the P_O sequential takes a verb form with tone pattern A and a marker ϵ between the subject and the verb. This ϵ is obligatory with a noun subject, but only optional with a pronominal subject (see 'I saw a child fall' above). The same sequential forms appear after [+FOC] and [-FOC] tenses—and after P_1 and P_2 , as seen in the following representative sentences:

me mbéè nyèm, wàn ɛ kfùn wè 'I sang and a child hit someone' (P1) me nci nóò nyèm, wàn ɛ kfùn wè 'I sang and a child hit someone' (P2)

As seen in the following sentences, however, the sequential forms are different when following a P3 tense:

```
me ntò nóò nyēm, ɛ wan kfun wè 'I sang and a child hit someone'
me ntò nóò nyēm, ɛ wan dōmsè wè 'I sang and a child refused someone'
me ntò nóò nyēm, ɛ wan dōmsè wè 'I sang and a child pacified someone'
""" sang and a child pacified someone'
```

In this case the ϵ precedes the subject and the tone pattern of the sequential verb is D, rather than A (cf. the consecutive D pattern in Chapter 14).

As seen in the following sentences, there is no negative sequential form when following a [-PROG, -FUT] tense in the main clause:

```
me nóò nyem, wvù bàa kfun wè 'I have sung and he has not hit anyone' me mbéè nyem, wvù bà baa kfun wè 'I sang and he did not hit anyone' (P1) me ntò nóò nyem, wvù ci bàa kfun wè 'I sang and he did not hit anyone' (P2) me ntò nóò nyem, wvù tò bàa kfun wè 'I sang and he did not hit anyone' (P3)
```

The negative second clauses are identical in form to the main clause negative. Thus, these sentences literally mean 'I sang, he did not hit anyone'.

It has been pointed out that the [+FOC] auxiliary no comes from the copula nú be' and a marker ϵ . Indeed, nú ϵ is an alternative pronunciation for nóò, e.g. wvù nóò gwè ~ wvù nú ϵ gwè 'he has fallen'. We are now in a position to appreciate the source of this ϵ : it is in fact the sequential marker found after P_0 , P_1 and P_2 main clauses. This explains why the [+FOC] P_0 - P_2 tenses have a main verb with tone pattern A. (The P_3 remains anomalous.)

In the following attested sequential we note that the \mathbf{P}_2 auxiliary verb cf may occur:

me ncí mbóy, wvù céè sheè èkfuh wan 'when I went out he remained and hit the child'

The form cee derives from c1 + ϵ , and the verb 'hit' is in consecutive form with tone pattern D.

Now, both the cee form for the P2 and the zero form for the P0 can occur as a main clause, especially as an interjection:

```
wan \epsilon gwè 'the child has fallen' wan \epsilon céè gwe 'the child fell'
```

(The tone pattern D in the second example is not readily explainable.) These sentences seem to function as interjections. They do not require no (to create no) or io to be complete. In order to appreciate the nuance distinguishing these "main clause sequentials" from regular $P_{\rm O}$ constructions, contrast the following:

```
me nóò nkfùn wàn 'I have hit the child' (report, out of the blue)
me ε ηκfùn lò wàn 'oh! I've just hit the child' (exclamation, probably an
i'I have hit the child' accident/unintentional)
(that's why he's crying)
```

These sequentials are most closely related to their corresponding noo construction, although they seem to function as exclamations rather than neutral reports.

3. [-PROG, +FUT]. The [+FUT] sequentials corresponding to the [-PROG, -FUT] constructions we have just seen are illustrated below:

```
me nú nyém, \epsilon wvùú kfun wè "I am about to sing, and he is about to hit me nú ngéé nyém, \epsilon wvùú kfun wè "I will sing, and he (F_1) someone' (F_0) me nú nyúú nyém, \epsilon wvùú kfun wè " " " (F_3)
```

The verb is always in tone class C. As in the case of the P_3 , the subject is preceded by a marker ε . This time, however, there is an additional H tone vowel (-V) following and fusing with the subject. We shall see that these forms are used also in certain subjunctive clauses.

When the future sequential clause does not have a tense auxiliary, it is normally assumed that the action of this clause will take place during the same time span as the action of the main clause. It is possible, however, to put a future auxiliary in the sequential clause, e.g.

```
me nú nyếm, \epsilon wvùú bốy kfun wề 'I will sing, and he will (tomorrow) F_0 hit someone'
```

The negative future sequential is obtained by placing se ya before the verb (and ke at the end of the clause). (se ya often contracts to sad.) This form has been found at all future tenses:

```
me nú nyếm, wvù sàá kfun wè kɛ 'I'm about to sing, and he is not about me nú ngéé nyém, wvù sàá kfun wè kɛ to hit anyone' (F_0) me nú nyúú nyém, wvù sàá kfun wè kɛ 'I'll sing, and he will not hit anyone' (F_1, F_2, F_3)
```

The marker sé in other contexts means 'before' (see Chapter 21). As in the affirmative, one can put a future tense auxiliary in the negative sequential clause (cf. me nú nyúú nyém wvù sẽ yùú kfun wan kɛ).

4. [+PROG] sequentials. The affirmative, progressive sequential is formed by using the [+PROG] verb form with tone pattern A. Examples follow:

```
yungsè dié bènkfǔ ε me ngènè fòwày 'Yungse is eating yams, and I am going me nyemté ε wvù kfùnè wan 'I am singing and he to market' is hitting a child'
```

The following sentence illustrates a complement clause which is a [+PROG] sequential (cf. section 1):

me mbέè nyow ε wan beelè 'I heard a child crying' (over time)

(The [-PROG] is also possible, i.e. me mb $\dot{\epsilon}\dot{\epsilon}$ nyow ϵ wan ϵ bee 'I heard a child cry' (punctual, shorttime).)

The negative progressive sequential is formed as follows:

me nyemté ϵ wvù yà kfùnè wàn k ϵ 'I am singing and not hitting a child'

Again it is tone pattern A, this time preceded by $\gamma \grave{a}$ and having $k\epsilon$ at the end of the clause.

- 5. Subjunctive clauses. As a type of sequential, subjunctive clauses are considered here. There are two types:
 - (a) those which cause a H tone prolongation on the subject of the clause;
 - (b) those which do not cause a H tone prolongation.

We saw examples of (a) in section 3, since this construction is used as the normal sequential after a future verb in a main clause. It is not easy to differentiate the two constructions semantically, and more work will be necessary to do so satisfactorily. A hint of the problem is seen in the fact that the two appear to be interchangeable in certain contexts, e.g.

me ntốố lẽ wvùu kfun wan 'I am coming in order for him to hit the wvù bếc tố lẽ me nyến wvù lễ 'he came in order that I see him' child'

These sentences show that both subjunctives are used after a verb such as 'come' to indicate purpose. The following sentences show that both are found in the consequent clause after an 'if'-clause:

ε yunsè bě yem, ε me mbín 'if Yungse sang, I will dance' ε yunsè yèm lò, ε me mbín 'if Yungse sings, I will dance'

It may be that the H tone subjunctive expresses greater unreality or doubt, since it is the normal hortative, e.g.

me ńgèn éè 'may I go?' bèń gen 'you pl. go!' gee é me ńgwe 'let me fall!'

It also has been found to express an indirect command, e.g.

wvù béè gày le me ngènè 'he told me to leave'
" wòó gènè you sg.
" wvùú gènè him/her
" bèsèn gènè us
" bèn gènè you pl.
" bó gènè them
" wan gènè the child

(The last line shows the irregular creation of a MH rising, which is normally not permitted in the language). As seen in all of the above examples, the tone pattern is C in the non-progressive, A in the progressive.

As mentioned, the \emptyset subjunctive has been found to express purpose. It may be less of a hortative in nature, and more of a consequential or resultative, e.g.

me nóà nfòn fèño le wò séé ñàm dvů 'I have sharpened a knife for you to cut meat with it'

As seen, there are two complementizers (IE and &), with & indicating greater purpose or force (i.e. 'I sharpened a knife so that...' vs. 'I sharpened a knife in order that...'). Compare also the following:

me ncèyté la wan diékén 'I am waiting for the child to eat' me ncèyté é wan diékén 'I am waiting in order for the child to eat' Interestingly, the causative, formed with the verb ge 'make, do', takes a sequential clause after a [-FUT] tense, but a \emptyset subjunctive after a [+FUT] tense, e.g.

```
me mbét ngè wan s bee 'I made the child cry'
me ngèé s wan beelè 'I am making the child cry'
me nú nge s wan béé 'I will make the child cry'
```

This may be because there must be an agreement in [FUT] marking between the two clauses. It will be noted that the \emptyset subjunctive is identical in tone to the future tenses (C, if [-PROG], A if [+PROG]). In fact, the future tenses are built on this pattern following the copula nú. Take out the nú, and you have the \emptyset subjunctive.

```
wvù béè gày le fò wan gen ke 'he told the child not to go' wvù kàně le fò me ntó ke 'he is refusing that I come'
```

The same negative construction is observed here as was seen earlier in the imperative (p.53). More examples of both types of subjunctives will be seen in the discussion of condition clauses in the following chapter.

TEMPORAL AND CONDITION CLAUSES

- 1. Temporal clauses. Temporal clauses express events which take place either simultaneous or before or after some other event, which is reported in the main clause. A number of different constructions have been encountered. Since a lot more work remains to be done in this area, the findings presented in this chapter are intended mostly to pinpoint topics for further research.
- 1.1. 'While' and 'as' clauses. The first kind of temporal clause to be considered involves actions which take place over time, but which are simultaneous with some other action. Such clauses are usually expressed with the conjunctions 'while' and 'as' in English. Note, first, in the following sentences that a simple progressive consecutive is used if the subject of the second verb is identical with the subject of the first:

me mbéè ngèn fòwăy nyemtè 'I went to market singing'
me nci nóò nfĭ wvù bó kwen ngènè fòwăy 'I helped him with firewood
going to market'

When the two subjects are different, a separate clause introduced by $\boldsymbol{\epsilon}$ is required:

bố nú mɛ̃sě ε wò diekènè 'they will finish while you are eating' bố nú mɛ̃sě ε wò bεε diekènè 'they will finish while you are eating' wvù bế mɛ̃sè ε wò bεε diekènè 'he finished while you were eating'

The first sentence has only a single [+PROG] verb, with tone pattern A. The other two sentences have $b \dot{\epsilon}(\dot{\epsilon})$ 'still', which does not alter the overall meaning. As seen in these sentences, the form of the second clause stays the same after past or future main clause verbs.

If the verb is [+PROG], an alternative way of expressing such durative simultaneity is with the form seke 'when', e.g.

me mbé nyemtè sékè wvù bě kfùnè wan 'I was singing when he was hitting the child'

1.2. 'When' clauses. As just seen, sékè can introduce a 'when' clause. Other examples are as follows:

me mbé nyem sékè wvù bě kfùn wan 'I sang when he hit the child' sékè me nyemté fò wò bín kɛ 'when I sing, don't dance!'

As seen on p.93, sékè is also used in indirect questions. It will be seen in Chapter 22 that it also is used in direct questions, e.g.

bố yemtế sékế lấ 'when are they singing?'

This form probably once was a noun meaning 'time'.

Other means of expressing non-progressive simultaneity in a 'when' clause

are seen in the following sentences:

wvù nử diờ tổ ể me ndee bvulem 'when he will come, I will be cooking fufu' diờ yem fò wò bín ke 'when you sing, don't dance!' 'when you sing, don't dance!'

The first two sentences involve the auxiliary dlò which immediately precedes the main verb. The last sentence uses the locative marker fò (class 16) in a new temporal sense. It may thus be interpreted as meaning 'where you sing, don't dance!'.

1.3. 'After' clauses. 'After' clauses have been recorded with either seke (='when') or the auxiliary diò, e.g.

bố nú mềs số sốk bố diek n yun số nử dið dốc bìnk fử ể njì sốc số lá 'they will finish after they eat' 'after Yungse cooks yams, Nji will sweep the compound'

The distinction between 'when' and 'after' is minimized when the verb form in the former case is [-PROG]. Thus, the first sentence above could also be translated 'they will finish when they have eaten'.

1.4. 'Before' clauses. While 'after' clauses are technically not distinct from 'when' clauses, 'before' clauses are. First note the following sentences, where the subject of the 'before' clause is identical to the subject of the main clause:

me mbé ngènè fòway nyemtè ènsẽ nyen wò lễ 'I was going to market singing,
when (=before) seeing you'
they will finish before coming'
bèsèn nử tổ số èdiekèn 've will come before eating'

The marker sé means 'before' but as seen in the first sentence, where it takes a nasal because the subject is first person singular, it is a verb. As seen in the following sentences, sé also means 'without':

me mbé ndiè bvuism số ñàm 'he ate fufu without meat'

The same marker is used if the subjects of the two clauses are not identical:

bố nú mẽ sẽ ε wò sế dịckèn
me mbết nyềm wvù sẽ kfun wan
wvù bế kfun wan me nsẽ nyềm
'they will finish before you eat'
'I sang before he hit the child'
'he hit the child before I sang'

Because there is no distinct 'after' construction, the examples in section 1.3 being built on 'when' clauses, 'before' clauses often are used to express 'after' by reversing the order of the actions. Thus, the last sentence can also mean 'I sang after he hit the child'. Finally, it will be noted in the following examples that the 'before' construction is sometimes translated other than with 'before' in English, e.g.

me mbék nkfun wâ wan wò số to 'I had already hit the child, when you me nú nyém me nsố ya mbín (kc) 'I will sing and not dance' arrived'

While the first sentence could have been translated 'I had already hit the child before you arrived', the second can not mean 'I will sing before I

dance'. (The latter meaning would be expressed by me nu nyém nsé mbin). Consider, on the other hand, the following sentence, which does not involve sé at all:

wà nữ gế toà ϵ me ndè $\hat{\epsilon}$ wâ bìnkfữ 'I will have cooked yams by the time you are arriving'

The literal meaning of this sentence is 'you will be (doing and) coming that I have already cooked yams'. Thus, the cooking of yams takes place before the arriving.

- 2. Condition clauses. Condition clauses show some variation and are most adequately described in terms of an 'if' (or 'condition') clause preceded by ϵ , followed by a 'then' (or 'consequent') clause. The consequent clause is introduced either by $\dot{\epsilon}$ or to, and the condition clause apparently can take any tense, circumstances permitting, e.g.
 - ε yunsè dèèle bìnkfu é me nyèèsè la 'if Yungse is cooking yams, I will sweep the compound'
 - ε yunse nóò dèè bìnkfǔ to njì nóò yèse lá 'if Yungse has cooked the yams, Nji has swept the compound'
 - ε yunse nử gó dèèle bìnktử to me nú ngó lif Yungse will always cook yams, I will sweep the compound' yèèsè lá

The above sentences represent "simple conditions". The condition clause states a proposition which may or may not be true, but on which the consequent clause depends. The following sentences show that the [-FOC] variant of [-PROG, -FUT, -NEG] tenses is expected in condition clauses:

- ε yunsè dèkě bìnkfů to njì nóò yèsè lá ε yunse be dêke binkfu to nji béê yèse lá
- e yunse cí děkě bìnkfů to nji cí nóo yèse lá
- ε yunsè tò dèkě bìnkfǔ to njì tò nóò yèsè lá

'if Yungse has cooked/cooked yams, Nji has swept/swept the compound'

(Po, P1, P2, P3)

There is a slight difference in meaning when the [+FOC] variants are used (cf. their presence in relative clauses, pp.93-4), e.g.

- 'if Yungse has cooked/did ε yuŋsè nóò dèè bìŋkfǔ to njì nóò yèsè lá cook yams, then Nji has e yunse bee dee binkfu to nji bee yese la swept/swept the compound' ε yuŋsè cí nόὸ dὲὲ bìŋkfù, etc.
- ε vunsè tò nóò dèè bìnkfu, etc.

 (P_0, P_1, P_2, P_3)

The emphasis is on the reality or truth of the proposition included within the condition clause. It is thus as a result of Yungse's cooking yams that Nji sweeps the compound. It is interesting to note that the ϵ which introduces the condition clause can be deleted if the tense within it is [-PROG, -FUT, -NEG, -FOCl, e.g.

'if Yungse cooks yams([-FOC]), I will eat* (ε) yuŋsè dèkě bìŋkfǔ έ me ndíé

In this example we also observe that the [-FOC] Po tense is used with future time reference. For this reason its [+FOC] variant can be used to make it clear that a Po reading is intended (i.e. if Yungse has cooked yams'). Similarly, as seen in the following example, the two P1 tenses may be used with a hypothetical or unreal sense:

(ɛ) yuŋsè bẻ dɛɛ bìŋkfǔ to me mbéè ndìé ɛ yuŋsè béè dèè bìŋkfǔ to njì béè yèsè lá

'if Yungse had cooked yams, I would have eaten (them)' 'if Yungse had cooked yams, Nji would have swept the compound'

Note in the first sentence that the verb dee has M tone and does not take its irregular dèkè form seen in the P_0 example. Since dèkè is expected with the [-FOC] P_1 tense (see p.63), and since the P_1 should assign 'cook' a L tone as per tone pattern A (see p.74), it is likely that the bé is the conditional auxiliary seen on p.90. In any case, there is probably a relationship between P_1 bé and the conditional auxiliary. Below it is seen again to replace the P_1 marker (cf. the P_2 condition clause which follows):

me mbèé (nèé) ηgwe ε wò (nɛn) bé baa wám

"I would have fallen if you
had not screamed' (P1)
"I would have fallen if you
had not screamed' (P2)
had not screamed' (P2)

Another feature of the 'if-then' construction is that the subject of the consequent clause is often deleted if it is coreferential with the subject of the condition clause. Thus, compare the following two sentences:

ε yunsè dèkě bìnkfů ε wvùú yèsě lá 'if Yungse cooks yams, he [#Yungse]
ε yunsè dèkě bìnkfů to yèsě lá 'if Yungse cooks yams, he [=Yungse]
will sweep the compound'

Finally, it will be noted in different consequent clauses above that both subjunctives seen in Chapter 20 are utilized in this construction (i.e. the \emptyset subjunctive and the $-\hat{V}$ subjunctive).

INTERROGATIVES AND FOCUS

1. Yes-no questions. Yes-no questions have all of the properties of declarative sentences, i.e. the same word order, verb morphology, tone etc. These interrogatives are distinguished from their declarative counterparts by having one of two particles at the end of the sentence, e.g.

```
yunsê ci nóò dèè kèngòm fogò lễ 'did Yungse cook plantains yesterday?'
yunsè ci nóò dèè kèngòm fogò lê '' '' '' '' (P2)
yunsè béè dèè kèngòm ễ 'did Yungse cook plantains?' (P1)
yunsè béè dèè kèngòm ê '' '' '' '' ''
```

The first sentence of each pair uses the question marker le/e, while the second sentence of each pair uses lê/ê. The first marker has also been recorded with L tone, e.g.

```
wvù nóò gwè lè 'has he fallen?'
cf. wvù nóò gwè ê """
```

As seen in these last sentences, the variation between $|\xi|$ and $|\xi|$ vs. ξ and $|\xi|$ is slightly different for the two markers: in general, the first marker is realized $|\xi|$ after a vowel and ξ after a consonant; the second marker is realized in most cases $|\xi|$ after a double vowel and $\hat{\xi}$ after a short vowel or consonant. (No explanation is offered for the form $|\xi|$ in the second sentence above.)

The difference in meaning between the two question markers is subtle, but appears to rely on the expectations of the speaker. In utilizing |ê, the speaker expresses surprise. Thus, in the second sentences we would have the meanings 'did Yungse cook plantains?' (understood: he wasn't supposed to, I didn't expect him to), 'has he fallen?' (understood: I'm surprised, etc.). This additional indication of surprise or unexpectedness is not present in the questions formed with |ě.

Finally, it is possible to add the marker lê to a single word, e.g. a noun, to work a question: wan ê 'a child?'.

2. Wh-questions. Noni forms "WH" questions by placing one of several question words in the position immediately following the verb. When the question word is the subject of the sentence, the dummy subject marker è is found in the initial subject position, e.g.

```
    dèèlĕ yee kèngòm
    kfùnĕ yee wan
    iyho is cooking plantains?'
    iyho is hitting the child?'
    yené yee wan é
    iyho sees the child?'
    iyhat has fallen?'
```

As seen, there is no question marker at the end of the sentence. (The é seen in the third example is the lé particle required for the verb 'see' and a few other verbs; see p.81.) However, lè can be added in an echo question, e.g.

```
le è gwè la lè 'what fell?' ('you said that what fell?')
```

The initial marker 1ϵ is the complementizer used, among other places, in reported speech; cf. 1ϵ no 'I say' (to one person) vs. noone (to many people) = the name of the Noni language. Thus, the above sentence means '...that what fell?'

The question word yes 'who(m)' has a plural form bàà. Both of these are now illustrated in object position:

```
wò kfùně yee 'who (sg.) are you hitting?'
wò kfùně bàà 'who (pl.) are you hitting?'
```

Similarly, the question word Iá 'what' has a plural form mùlă, as illustrated again in object position in the following sentences:

```
wò kaanké lá 'what (sg.) are you frying?'
wò kaanké mulă 'what (pl.) are you frying?'
```

Again there is no final question marker other than the WH word. The marker lè can be added if the interrogative is an echo question, e.g

```
wvù kfùně ε nùù yεε lè 'he is hitting who?' (='you said he is hitting who?')
```

This time, however, because the question word is an object, it is placed in great focus by the preceding focus markers ϵ nùù (see section 3). The normal or non-contrastive way of asking a question where the WH word is other than the subject of the sentence is by placing the WH word immediately after the verb, as seen below:

```
wà gèně feè 'where are you going?'
wà gèně séké lá 'when are you going?'
wà gèně ni lě 'how are you going?'
wà gèně nje lă 'why are you going?'
wà yené wàn wú lá lé 'which child do you see?'
```

The word lá 'what' is seen in three of the above questions which are thus interpreted as 'what time are you going?', 'what reason are you going?' and 'what child do you see?'. The lé of the 'how' question is probably the postposition seen in Chapter 17, section 3.

Questions can be made on indirect objects, objects of prepositions etc., e.g. wò ñàá kènkfǔ ε yεε or (more polite) wò ñàá kènkfǔ ε nùù ε yεε 'to whom are you giving the yams?'

Indirect questions consist of a generic head noun + a relative clause (see p.93). The following example of a WH word being used as an indefinite universal quantifier was recorded:

```
è tổ yεε ε wò gāỳ 'whoever comes, greet him!'
```

3. Focus. In discussing the [+FOC] vs. [-FOC] variants of certain tenses, it was pointed out that the former are used in the Po-P3 when there is "even focus". That is, no particular element within the sentence is singled out for prominence. Such sentences have the basic subject-auxiliary-verb-object-oblique-adverbial order and are particularly appropriate for utterances made out of the blue, e.g.

bwēm, yunsè nóò dié bvulem bwēm 'Mother, Yungse has eaten my fufu!'

This particular sentence is intended to represent how a child might report to his mother that Yungse has eaten his fufu, i.e. running to her without any previous linguistic context. Had there been any previous context such as 'why are you crying?' or 'what has happened?', the child might answer with the [-FOC] Po as follows:

vunse dié byulem bwem 'Yungse has eaten my fufu'

In this particular case, the object 'my fufu' receives prominence, as indicated in the translation. In general, a sentence such as this last one can answer any number of questions, e.g. 'what has happened?', 'what has Yungse done?', 'what has Yungse eaten?', 'whose yams has Yungse eaten?', 'what of yours has Yungse eaten?'. In answer to each question the scope of the assertion is different. Thus, in answer to the question 'what has happened?', the assertion would be the whole proposition ('what has happened is that Yungse has eaten my fufu'). In answer to the question 'what has Yungse eaten?', the assertion would be the noun phrase object ('what Yungse has eaten is my yams').

Although all of these interpretations are possible, the constituent found immediately after the verb will be considered the marked focus (cf. Watters 1979) unless one of the following obtains:

(a) the [+FOC] tense is used, if there is one;(b) the verbal focus marker 15 immediately follows the verb;

(c) a constituent other than the one immediately following the verb is marked for focus (by the markers ε nùù).

The alternative in (a) was seen at the bottom of p.105. The other two alternatives are seen below:

'Yungse has just eaten my fufu' vunsè dié là bvulem bwem yunse be die byulem bwem e nuù eben 'Yungse ate my fufu today' (i.e. not yesterday)

The marker 13 in the first sentence can either indicate 'just' (as indicated) or 'instead', or it can focus on the lexical meaning of the verb, e.g. 'Yungse has eaten my fufu' (he hasn't just cooked it, as you said). The second sentence places ϵ nùù before the adverb èbèn 'today', contrasting it with, e.g. 'vesterday', as indicated.

A sentence with a [-FOC] Po-P3 tense is thus incomplete unless it is an answer to a question, as seen, or unless it is changed according to (a), (b) or (c) above. This is particularly striking in the case of intransitive verbs which do not have any complement. As was seen in Chapter 12, the marker 15 is required to complete these tenses as well as the 0 tense, e.g.

bèsèn yemté là 'we are singing' 'the child is falling' wan gwèé là

Take off the 15 and the above sentences are ungrammatical. Thus, in the 0 tense, or in [-FOC] Po-P3, if there is no element to the right of the verb, the citation form of the sentence will necessarily involve 15. If there is an object (or other complement), 15 is not required in the 0 tense, e.g.

'Yungse is cooking plantains' vunsè dèèlě kèngòm yunse deele lo kengom 'Yungse is cooking plantains instead' However, in the P_0-P_3 [-FOC] tenses, even such sentences involving a complement seem incomplete--because they place focus on the complement (and seem to be answering a question):

```
yunsè c' dèké kèngòm 'Yungse cooked plantains'
yunsè c' nóò dèè kèngòm 'Yungse cooked plantains'
yunsè c' dèké lò kèngòm 'Yungse cooked plantains'
'Yungse cooked plantains'
'Yungse cooked plantains'
```

The citation form of a P_0 - P_3 tense is always the [+FOC] one, because it is this form alone which does not assume a previous linguistic context.

It should be noted with respect to the above four sentences that only one focus operation can take place per sentence. Thus, if the [+FOC] variant of a tense is chosen, neither 1δ nor ϵ nùù can co-occur with it; if 1δ is present, ϵ nùù cannot co-occur with it, and as we shall see, if the subject is postposed after the verb, we get none of the three morphological focus markers ([+FOC] tense, 1δ , or ϵ nùù).

- 4. Constituent focus. Different constituents can be overtly marked as focused by at least three different means.
- 4.1. Subject focus. Whenever a subject is focused (e.g. in answer to a subject-WH-question, or in order to contrast the subject of a sentence, etc.), it is placed immediately after the verb. A dummy subject marker è remains in the original subject position, e.g.

```
è gwě njì 'Nji has fallen'
è bě dèkè yuŋsè 'Yungse cooked (it)' (P<sub>1</sub>)
è yemté wan εjè 'a child is singing on the road'
```

These sentences can answer the questions 'who has fallen?', 'who cooked (it)?' and 'who is singing on the road?', or they can contrast with another person's prior assertions 'John has fallen', 'Mary cooked (it)' and 'a woman is singing on the road'. When the postposed subject is first person singular, there no longer is any nasal agreement, cf.

```
me nyemté là 'I am singing' 
è yemté me 'I am singing'
```

What is interesting about subject-postposing is that the object may optionally be preposed to the verb, pre-empting the dummy subject marker:

```
è dèèlě njì kèngòm 'Nji is cooking plantains'
kèngòm dèèlě njì 'Nji is cooking plantains'
```

The second sentence has the grammatical form 'plantains are cooking Nji', and yet it is clear from the semantics that it is Nji who is cooking plantains. Both utterances can be used to answer the question 'who is cooking plantains?', although the second variant, with object preposing, implies that the question was angry (i.e. 'Nji was cooking plantains, that's who was cooking plantains'). While the first sentence is more neutral, with 'plantains' simply repeated as previously established information, the second sentence sounds more counterassertive. The second sentence would be particularly appropriate if contradicting someone's assertion 'Yungse is cooking plantains'. The following exchange demonstrates how the interpretation of such a sentence might depend on context:

k kfùně yee wan 'who is hitting the child'!
wan kfùně njì 'Nji is hitting the child'

The second sentence would with no context specified mean 'the child is hitting Nji'. However, when following the question 'who is hitting the child?', it takes on the opposite reading: 'Nji is hitting the child'. While a number of instances have been recorded with subject-object inversion in sentences with a 0 or P_0-P_3 [-FOC] tense, inversion apparently cannot take place if the object to be preposed is a 1st or 2nd person pronoun. We thus find the following question-answer exchange:

È kfùnế yea me 'who is hitting me?' È kfùnế njì (wò) 'Nji is hitting (you)'

The sentence wò kfùně njì could not answer this question, since its only meaning is 'you are hitting Nji'.

No examples were recorded where a postposed subject was preceded by ϵ nùù, the focus marking system employed for non-subjects.

4.2. Verb focus. As indicated above, a verb is placed in focus by following it by the marker 13. The meaning taken on is 'just', 'instead', or lexical meaning of verb in focus. Thus, in answer to the following question, the verb focus is marked as indicated:

wò bế gè là 'what did you do?' (P₁, [-FOC]) me mbé yem lò 'I sang'

4.3. Object focus. The object is found directly after the verb and therefore, with the right [-FOC] verb form, and without a preceding 13, a sentence can always be interpreted with the object in focus, e.g.

wà cỉ dèkế lấ 'what did you cook?' me ncỉ ndèkế bìnkfǔ 'I cooked yams'

In order to make the object contrastive or counter-assertive, it is preceded by ϵ nuù:

wà cĩ dèkế ε nùù là 'what did you really cook?'
wà cĩ dèkế kèngòm 'you cooked plantains'
me ncĩ ndèkế ε nùù bìnkfǔ '(no,) I cooked (only) yams'

The last sentence can either be in answer to a WH-question which itself uses ϵ nùù, or it can follow the second sentence, correcting 'plantains' with 'yams' (and preceding the answer with $\delta\gamma$? 'no!'). As seen, the markers mark the following constituent as being the only appropriate filler of that slot (cf. ké 'only' which was seen before a focused element on p.85).

4.4. Oblique and adverbial focus. Other elements of the sentence are focused by placing ϵ nùù before them:

bố ñáá kèngòm ε nùù ε wan 'they are giving plantains to the *child*' me ncí gwě ε nùù fɔgò 'I fell *yesterday*' (P2)

The form ϵ nùù is a [+PROG] consecutive of the copula nú 'be'. The meaning of the last sentence thus is 'I fell and being yesterday'.

REFERENCES

- Anderson, Stephen C. 1979. "Verb structure". In Hyman (1979b), Part II, 73-136.
- Hombert, Jean-Marie. 1980. "Noun classes of the Beboid languages". In L. M. Hyman (ed.), Noun Classes in the Graesfields Bantu Borderland, 83-98. Southern California Occasional Papers in Linguistics 8. Los Angeles: University of Southern California.
- Hyman, Larry M. 1971. "Consecutivization in Fe'fe'". Journal of African Languages 10.29-43.
- Hyman, Larry M. 1975. "Noni (Misaje group)". Appeared (1980) in L. M. Hyman and J. Voorhoeve (eds.), Les Classes Nominales dans le Bantou des Grassfields, 259-274. L'Expansion Bantoue, Vol. I. Paris: Société d'Etudes Linguistiques et Anthropologiques de France.
- Hyman, Larry M. 1979a. "Phonology and noun structure". In Hyman (1979b),
 Part I, 1-72.
- Hyman, Larry M. (ed.). 1979b. Aghem Grammatical Structure. Southern California Occasional Papers in Linguistics 7. Los Angeles: University of Southern California.
- Hyman, Larry M. 1980. "Relative time reference in the Bamileke tense system". Studies in African Linguistics 11.227-237.
- Richards, Russell M. 1981. "Les langues des GRASSFIELDS du Cameroum: Bibliographie critique avec notes sur la langue noni". Mémoire pour le D.E.A. Université de Nice.
- Watters, John Robert. 1979. "Focus in Aghem: a study of its formal correlates and typology". In Hyman (1979b), Part III, 137-197.

ENGLISH-NONI WORD LIST

			7/0
accept	béé	bark	keban 7/8
accompany	bilé	basket,	ŋkàà/ŋkaà 9/10
accuse	tổy	large	kèncakè 7/8
add to	kwếnsế, bTsě	kind of	gvume 1/2
adult	cowè	bat	kelim 7/8
advance	sa fwě	bathe	cokě
advance	ntefe 1/2	be	ոմ ոնն
	téfé	bead, v.	500
advise		bean	όwe/ε όw 5/6
after	sékè, diò	bear fruit	yóm
afternoon	ε mùnshion		low
again	bốổ, bvú	bananas	kedéw 7
age-group	keghan 7/8	beard	
agree	béé	beat, v.	son
agreement	nsèysè 1/2	drum	kom ncòm
ahead	fwe	because	njè
all	-tfu/-cii	bed	kentaw 7/8
already	wâ	bed, in the	èntaw 17
also	tĒĥ	bedbug	kènguùn 7/8
angle	bόό/εbów 5/6	bee	ลักได้เ 9/10
angry, be	tónyế shem	beehive	ŋwaw ñi 1/2
angry, oc animal	ñàm/ñām	before	SE
		beg	LÉKÉ
ant	Egonle	behind	έj Y m
antelope	jòm/jōm	be1ch	bew
anus	ntòn		
arm, sg.	keyi 7	bell	mbyene/mbyene 9/10
p1.	nkenken 4	bell (hunting)	Kecolo //o
armpit	kènkamkam 7/8	belly	tři/etew 5/6
arrange	seyse	belt	kècàw 7/8
arrow	mǔn/bòmûn 1/2	bend, tr.	ngvůsé .
ascend	ben	bent, become	lem
ashes	taan 5	between	èntě lến 17
ask (for, Q)	bífé	bile	blén 9
assemble	bááncé	bird	feñii/muñim 11/12
at	fà	birth, give	b 5 5
automobile	mutù 1	bite	lóm
avoid	lew	bitter, become	
•	j i	black, become	
a person		black dye	njiw
go w/out	jim	blacken, tr.	ylise; intr. (n)jiw
axe	temé/stem 5/6	· ·	byuyilè 14
back	j)m/j/m 9/10	blackness	
backbone	kfum 4	blood	εlεmé/bilεεm 7/8
bad, be	béfé	blow w/mouth	féfé
bag	εkεlè 7/8	lips, bello	
bag, in the	ὲkê 17	horn	tấn
bake in ashes	fume	nose	mTsě
ball, make	táy	boa constricto	r kom/kom 9/10
out of fufu		body	ye 4
bamboo	gwájn/gájn 3/4	bone	εkufε 7/8
banana leaf	ncòò	book	ŋwà 1/2
		borrow	sám, foo
(unopened,	yerrowy	0011011	

bow, n.	kènèw 7/8	carving	nsem 1/2
box	nkòw 1/2	castigate	son
bracelet	kèshaw 7/8	catch	kổổ
brain	bòn 1	from air	ka
branch	třew/třew 9/10	caterpillar	kèmbenmbén 7/8
of vine	twéŋ/téŋ 3/4	edible typ	pe mfumè/mfumèn 9/10
break	san	cave	fwen/fen 3/4
snap	ng5y	cease	Ið
shatter, tr	. yo (kill)	of rains	sểmể
f.c.	kwe (die)	ceiling	gweè/geè 3/4
fufu	gosě	chair (=bed)	kèntāŵ 7/8
open breast	tënë	chalk	fwem 3
breath	mbenè/emben 5/6	change	kómén
breathe	keyoy 7/8	into	béy
bridge	yoy duoà 7/17	charcoal	kèkâ 7/8
brief, v.	dwεὲ 3/13	chase	kốŋ
broom	tfum	chat	fε
bubble, v.	keyesè 7/8	cheek	kediàw 7/8
build	yo yoʻo, d i ʻo	chest	gyùw/gv ú w 9/10
bunch (guinea		chew	cám
	kekám 7/8	chief	nfòn 1/2
corn) bundle	for 112/m	child	wan/boom 1/2
=dirt	feyilè/munyilém 11/12	chin	kenan 7/8
burn, tr.	yile 5/13 t3n	chisel, n.	kèncofe 7/8
intr.	kộộ (catch)	V.	cófé
burnt, become	fié	choose	cáw
burst	telž	chop	k <u>á</u> w, gwo
open	bvúyế, tếnế	church	c5
bury		claw	kawè∕εkāẁ 5/6
with hand	dyey, doy fom	clay	mbšm 1/2
bush (=forest)	εκόνε 7/8	clean, v.	che
bush country	ŋkwanté/ŋkanté 3/4	clear, be	yůů
bushcow	mbòn 1/2	grass	kāsě
bust	to	011	kwaa
but	gèènen	throat	ka
buttock	kèñì 7/8	click, make a	
button	mba 1/2	cloth	ndvu/ndvu 9/10
buy	goy	cloud cock	kèmbèw, keyekè 7
buzz (bees)	ko		nyem shèè/nyem shéé 9/10
cadaver	gvűne/εgvűn 5/6	cocoyam cold, be	kekoò 7/8
calabash	kebεn 7/8	coldness	18è
small mimbo			kèntòn 7/8
calf (of foot)	nsoo 4	color, v.	con
call	téé, bến	come, v.	saa t3
from distance	e kii	command, v.	vámén
cane	kèmbàn 7/8	compound	4 .
canerat	ntòn/ntōn 9/10(~1/2)	contradict	lá/εla 5/6 kTcέn
carry (on head)	tűű	cook, v.	
without hands	ទ ខៀម	fufu	dee
a child	mee	cornbeer	key
on a pole	shumě	cough	ŋkǎŋ 1 kểy
heavy weight		count	tág
carve	sém	country	ketom 7/8, wɔŋ̀ 3
			Veron 1/02 Moil 2

			- III -
courtyard	lå kfulen 4	drink, v.	wó
cover, v.	bấŋ	drip	cém
cowhorn	ket5ŋ 7/8	drive away	kon
cowry (=tooth)	kege 7/8	drown, intr.	1ε + jòò ο/10
crab	εkamε 7/8	drum	ncòm/ncōm 9/10
craw1	ŋāŋlĕ	drunk, be	jew
cricket	ndèèŋkůù 1/2	dry, become	yom
cross (road, pit	c) dan	dust	kebvu 7/8
river	lentě	ear	keto: 7, kentonton 7/8
crossing, foot-	kelfincεὲ 7/8	ear, in the	è <u>n</u> togton
crunch	ngélé	early, be	yanjsen
cry	béé tốn kwe		ngwénsé/ngénsé 3/4
cut (w/knife)	séé	eat	die
e.g. rope	sóm	egg	gée/ɛgé 5/6
in two	san	eight	ñàŋ
tree, flower	kfuy	eitheror	nòònòò
loose	kwan	elephant	klyóð 7/8
to measure	céy	embroider	tóm
out	kwêy	enter	léy
cutlass	ກິວັ/ກິວ 9/10	entertain	téw
cup	fek⊃/muŋkŚm 11/12	equal, be	kwenten
cure, v.	fε	evening	èk∣egùù~kegùù 17
curse, v.	wcl	exchange	komen
dance, n.	biné 5/13	excrement	còo/coo 9/10
ν.	bin	exit, v.	pol
dark, get	jffm	explain	nelė
dash, v.	mawcĕ	extinguish	jimse
dawn, v.	yůů	eye	jisé/ejise 5/6
day	dľw 5/6, bvutfu 14	eyebrows	monkonè 15
death	kwe /kwe 9/10	face	byushiw/monshim 14/15
debt	fìò/fiò 9/10	faint, v.	gwe kèngwè
deceive	i é é	fall	gwe
deep, become	sllyčn	fan, v.	fűlé
defeat	ghấw	farewell, say	k <u>e</u> m
defecate	ñi	farm, at the	wene
delay, v.	dvuy	fart, n.	ñ(i 5/13
depart	g€n	fat, n.	efom 6
descend	bow, shii	fat, become	fan, kow
dew	ngwi 4	father	cee 1/2
die	kwé	fatness	byufanê 14
dig	cow sow cim	fatty, be	f5m
dip	COW	fault, n.	nje
dirt	yile 5/13	fear, n.	jàn 9
disperse, intr.	sankě	ν.	fan
dispute, n.	nsaw 1/2	feather	gvunè/εgvữn 5/6
divert	léwsé	feed	ñáné.
divide	gaw	fence	ketāŋ̀ 7/8
dodge	ĬĔw	fence in	gem
dog	bwě/bwé 9/10	field	we/ne 3/4
door	diow fwésé	fight, n.	bvudiòw/mondiòm 11/12
dove	kembembε 7/8	ν.	dio (+bvudiòw)
down	kècinè	figtree	kengvum 7/8
drag	cii	find	b5ŋ
draw (water)	goòfě	fine, n.	kej5w 7/8
*			

21. (1. 3.4)			
finish	mēsĕ, ká	grass, grazi	ng gáÿ/εgáy 5/6
fire	ŋgốy 3	thatching	yun/yun 9/10
fire, set	keně		type kekwakwa 7/8
fireplace	kètaŋtε 7/8	white	kèsewsè 7/8
firewood	kwen/ken 3/4	green	
firm, be	cffm	grasslands	mbwesem/mbesem 3/4
first, be	yaw	grate	ngèy 1/2
fish	blè/blé 9/10	grave	kwaa
fishscale	ngaw/ngaw 9/10	greet	jėm/jėm 9/10
fishtrap	kènt fùm 7/8	each other	gayié
five	tTh	grind	3-7
flap (wings)	son	groan	gow, kaa
flash (lighten	ing) law	ground	dvum, kέγ
flatness	bvùyĕw 14		nshε/nshε 9/10
flea	fembam 11/12	ground, on th	
float	léy	groundnuts	bilèn 1
flour	mum 15	grow up	taw
fly, n.	byum/byum 9/10	grow (of root	
fly, v.	bf	grown, be	kó
fold	bon	guinea corn	gij/egin 5/6
follow	bllé	gun	ngóy/jingóci 5/13
	eat) byse bi dien 8	habitually	g č č
fool, v.	léésén	hair	yuw/yuw 9/10
foot	εkalε/bikaa 7/8	hand	kebo 7/8
forehead	fòjisé 16	handle, make	téy
forest	ε κό ψε 7/8	hang up	téy
forge	tfűy	on shoulder	
forget	daye	oneself	shlim
four	ne	hard, become	témé
friend	nsaan/nsaan 9/10	harm, do to	lốw kèn∣òw
frightening, be	1120011/1120011 3/10	harness, v.	sóó
frog	kèshàm 7/8	harvest, v.	gwelé
fruit		hat	kef5 7/8
fry	shem/shem 9/10	hawk	ndòn 1/2
fufu	kấŋ	he	w∨ù
full, become	bvulem 14	head	εkolε/bikoo 7/8
	yinsen	head, on the	fòfå 16
gall bladder	bábá 4	headdress	kènsèé 7/8
gather, fetch	kew ko	headpad	can/can 9/10
up get		headrest	mbwaàn 1/2
gift	kénké	heap, v.	kóóncé
give	ñå 1/2	in mounds	maw
	ñá.	hear	yów
as dash	mawce	heart	fetelè/muntelèm 11/12
sth. spoiled		heat, v.	kfum
to dancers	feně	heavy, become	ſſtĚ
back	tfűsé	hee1	nci /bòncí 1/2
go	gen	height	byudéfè 14
to farm	la	help, v.	ff
goat	bie/bie 9/10	hen	shèè/shéé 9/10
god	ñò 1/2	here	fen
good, be	jeè	hiccup	shiewshiew 9/10
gorilla	febow/mumbow 11/12	hide sth.	ñTiĕ
	cá? 1/2	oneself	ñim
square	ŋwaw 1/2	hip	kèncemte 7/8
		222 2 3	

hit (hammer)	kfun	knife	fèmò /mùmòm 11/12
things togeth	er kwáw	knot, v.	sóó
hoe, n.	jìw/jfw 9/10	know	kèè~kèy
hold	ji, kếŋkể	kolanut	fèmbi/mumbim 11/12
	kebōw 7/8	labor, be in	I 6m
large, tree	fwen/fen 3/4	lack	tan
honey	b∨utōw 14	ladder	bvuków/monkóm 14/15
honor, n.	ŋgvùm 1/2	ladle	kènsòmè 7/8
ν.	w∨ům	1ake	mámásé
hoof	kek o ỳ 7/8	lame person	ntènè 1/2
hook	fèncaw 11/12	language	diew
house	yew/yew 9/10	laugh, v.	kľe
how	nè, nù	law	nc 1/2
hundred	gwéé/géé 3/4	lay sth. down	jīkě tőfé
hunger	jèn 9	eggs	nfổơn 5/13
hunt, v.	la byugóm	leaf	jów
hunting	bvugóm 14	leak, v.	yľém
hurt	ton	lean against	ja, nε
husband	doy/bòle 1/2	leave	cíné, cínélé
hut	còn/con 9/10	sth.	gkàgkà
hut, in the	con 18	left (side)	of
hyena	kűlá 1/2	legislate lend	sámsé, foòsě
I	me	leopard	bìa/bia 9/10
if	kèncim 7/8	let	93p
illness	benen, few	lie, tell a	mbyay (+mbyay)
imitate	È, yilé	lie down	j) im
in incite	bέysé	light, be (wt.	
inclined, be	tfun	like, as	diew Élé, kedie
inform	tfum	1ick	lá
inheritance	bvushew 14	lion	byulè 1/2
injured, be	lêm	listen	yéké
inside	Èntě 17	little	(wan) cààn
instead	15	far	ncè njoolé
intelligence	byùfèé 14	liver	shem/shem 9/10
intelligent, be		1oad	ŋkĕm 1/2
interrupt	bvu	lock up	fen~fien
intestines	bvul e w 14	locust	cùma 9/10
invalid	ŋkàwlè 1/2	long, become	defě
investigate	S⊃₩	look	táá
invite	kaw	after	táá fó
iron	ketfu 7/8	at	ͺͺϳΤ <u>ὴ</u> ĕ ၞ
itch, v.	fő	loosen (ground	1) cewse
jealous, be	ghew	lose (to lepre	
jigger (=salt)	ŋkaw∕ŋkaw 9/10	lost, get	la 11/12
journey	ghané 5/13	louse	fensεsè/munsεsèm 11/12
judge, v.	sáw	love, v.	kən
juju (kind)	ŋkòw 1/2	lower, tr.	shTkě
jum <mark>p</mark> over	dan	lying (falseh	
just, adv.	13	mad, be	jew yin/yin 9/10
kidney	fTw 4	maggot	nfùne~nfòne/bònfune 1/2
kill	уб	maize	
knee	núu/enú 5/6	make	ge dièmsĕn/bolemsè 1/2
kneel	ŋg∨ůům	man	G CHISCH/ DO LONG 7/4

many	ntèèn	nine	bvùùkε
market	way/jiwaci 3/13	no	åv?
Nkor day 1	bvűt fù į	nose	nd ié 5/13
day 2	bvûnkaà	not	kέ… kε
day 3	bvúzhí	now	keséen
day 4	bvűkèmà	numerous, be	doo
day 5	bvậŋkaàdēh	oath	Jēw~n≀èw 1/2
day 6	bvűzhilden	often	êfanêfan
day 7	b∨ū́sɔ̃w	oil	meèm 15
day 8	byúmbón	o1d	ngày
marriage	kèntaashé 7/8	old, become	Ion
masculinity	bvulemsè 14	on	yllé
mat	ghām 4, ķejinē 7/8	once	kàn mwàn
matter	sáka 1	one	man
mausoleum	fum 4	on1y	kέ, maŋ
me .	me	open, v.	veně
measure, v.	few	open, break	san
medicine	εfowέ/bifoo 7/8	or	noo
medicine man	ngàn 1/2	outdo	ghẩw
meet	tásén	ow1	kènkfù 7/8
message	ntom 1/2	pacify	
milk, n.	emben 6	pack	lówcε, címse
milk, v.	kám		báánce
millipede	kèngວ່າຣະ 7/8	pain, n. pain, v.	nsasè, nlòcê 1
mistake	jây 1/2		lōċš
mistake, make	jáy	palace	ntow 1/2
mix	fyey, fyeyskn	pale, become	yom ,
money	kwaw	palm-kernel	cèe/cee 9/10
monkey	can/can 9/10	palmnut	bián/bíán 9/10
moon	kwee/kee 3/4	palmtree	kebán 7/8
morning		wild	duun 5
mortar	entán entán 17 ndvůu 5/13	partridge	gwè/gwé 9/10
mosquito	bem 4	pass, v.	f ő y
mother	bwee 1/2	pay	law
mould, v.	bwee 1/2 bóm	pebble	cow/cow 9/10
mountain		peck, v.	jów
mouth	kom/jikomè 3/13	peel	yun
	dlow/dew	penis	jòw/jów 9/10
mouth, in the much	èmû 17	pepper	nsose l
	ntèèn	person	wè/bɔñil(m) 1/2
mud	kencaà	perspiration	njimtèn 1
nashroom (kinas)	kèmfulàn, kèjèmtê 7/8	perspire	jŤm̀těn
nail, n.	kenkantê 7/8	pestle	ketfűw 7/8
nail, v.	tá	pick	kəy
nakedness	kècimte 7	up	bắn
name, n.	die 5/6	piece, v.	kổ
name, v.	co .	pierce	bon, céé
narrow, be	doosè	pig	ŋkonyan 1/2
	cem 9/10	pinch	nám
near, put	b <u>e</u> nc <u>ě</u>	pipe	kèlăn 7/8
neck	kèndòn 7/8	pit	kèntokè 7/8
net	kènsàn 7/8	place	bvudvůu/mondvůům 14/15
	-fε	plait	lo
news	sáka	plant, v.	jếwế
night	entaຶ່ງ 17	by suckers	yie
provide the State of the State	754577 2 5477	o) auckers	y i e

```
kf f 5/13
                                             relative
                 kèngòm 7/8
plantain
                                                             shéé
                                             remain
                 véfé
plaster, v.
                                                             mεy
                 ketáálé 7/8
                                                behind
platform
                                                             kwawce, komé
                                             remember
                 dvům (+ndvùm)
play, v.
                                                             bvůsé
                                             remove
please, adv.
                  ban
                                                             fásé
                                                from fire
                  hélé
pluck
                                                             cófé
                                                from heap
                  nitw 9
poison, n.
                                                             shámsé, fewcě
                                             report, v.
                  jíw
   ٧.
                                                             fíésén
                                             resemble
                  fofé
poor, be
                                                              Téétén
                                             rest. v.
                  ntòn 1/2, nshaàn 9/10
pot
                                                              lew
                                             restrict
                  kfun
pound, v.
                                                             cf
                                                 by fon
                  dom fo
pour
                                                              tfűsé
                                             retaliate
                  kefofè, kètân 7
poverty
                                                              kásé
                                             return
                  kfum, ców
powder, v.
                                                              kamsě
                                             revive
                  ncoonken 1/2
praise, n.
                                                              koné
                                             reward, v.
                  coònkěn
    ν.
                                                              jìtá 13
                                             ribs
                  tew
prefer
                                                              ati/eain 5/6
                                             rice
                  fwe 5
pregnancy
                                                              kwen/ken 3/4
                                             ridge
press, v.
                  ycm
                                             right (not wrong) len
                  fádà
priest
                                                              f èncaw
                                             ring, n.
                  domsě
prod
                                                              iεm
                                                 ν.
                  SEW
profit
                                                              ncàgùgù 1/2
                                              ringworm
                  nkaw
promise, n.
                                                              νέέ
                                              ripe, become
                  kaw
    ٧.
                                                              ben
                                              rise
protect
                  CEY
                                                              jè/jé 9/10
                                              road
                  goôfě
pul1
                  leè/εleη 5/6
                                              roast, v.
pumpkin
                                                               kembaàn 7/8
                                              rock
                  ñò wù jisé
pupil (of eye)
                                                               bwem/bem 3/4
                                              rod
                  monkfunye 15
pus
                                                               kέwé
                                              roll up
                  dāmsě
push
                                                               kebow 7/8
                                              room
                  dɔ̃mè, bāsè
    down
                                                              gwen/gen 3/4
                                              root
                  gεε, tóm
put.
                                                               kfu 1/2
                                              rope
    sth. away
                  cakě
                                              rotten, become fo
    in mounds
                  maw
                                                               dien
                                              round, become
    on stick
                  500
                                                 and smooth
                                                               cinsé
 quarrel, v.
                  wáá
                                                               vété
                                              rub
                  γāὴsἔn
 quick, be
                                                               sáw
                                              rule over
                  fwew/few 3/4
 quill
                                                               dvumen
                                              rumble
                  .wanga/bowanga 1/2
 rabbit
                                                               1éwté
                                              run
                  jàn 9
 rain
                                                               ncèse 1/2
                                              sacrifice, n.
                   keyáw 7/8
 ram
                                                               ntián 4
                                              saliva
                   bìòw/bíów 9/10
 raphia palm
                                                               nkaw 9
                                              salt
                  baynen
 rare, be
 rare, be in- baynen rat (small edible) fènkâw 11/12
                                                               monshéém 15
                                               sand
                                                               fűű
    edible fieldrat shon/shon 9/10, fyèw/
                                              satiated, be
                                                               ngèy 1/2
                                               savannah
                   fyεw 9/10, bwaw 3/13,
        (types)
                                               savannah, in the Entfun 17
                   fèngùu/mùngùum 11/12
                                                               dó
                                               say
    houserat
                   ncênkôn 1/2
                                                               kelalásé 7/8
                                               scar
    giant
                   cèe/cee 9/10
                                               scarce, become sey
                   bóy
 reach
                                                               káy
                                               scare away
                   yέξ
 red, become
                                                                      tέmsé
                                                               táwsé
                   bvuyelê 14
                                               scatter
 redness
                                                               kán
                                               scrape
                   ton, kan, fán
 refuse
                                                               kámé
                                               scratch self
                   kómé
 regret
```

```
scratch s.o.
                    shwaw
                                               slave
                                                               nfwàà/bonfaà 1/2
  scream
                    wám
                                               sleep, n.
                                                               ñi 5
  search for
                    mcp
                                                               léfé
                                                  ν.
     among things tem
                                               sleeping place (chief) kècéè 7/8
  season; hunting nyèm
                                               sleepy, be
                                                               yew.
     planting
                   modvuùm
                                               slide, slip
                                                               1 em
     rainy
                   byudaàm
                                               slow1v
                                                               wèè
     clearing
                   fwε
                                               smal1
                                                               tếlè, waà/boom
  see
                   yến...lể
                                               small, be
                                                               doosè
  seed
                   ŋgàw/ŋgɔ̈ẁ 9/10
                                               smell, intr.
                                                               sám
  seize
                   15
                                                  tr.
                                                               ηὖὖηkέη
  sel1
                   gesě
                                              smoke, n.
                                                               n.iTm 4
 send
                   tóm, clinsé
                                                  ν.
                                                               Ьſ
 separate (fibers) son
                                                 meat
                                                               tám
 set trap
                   TALE
                                              snail
                                                               kèmbà 7/8
     gum trap
                   láw
                                              snake
                                                               y5/yo 9/10
 settle dispute
                   sómtén
                                                  boa
                                                               kom/kom 9/10
 seven
                   sooshwî
                                                  short, fat
                                                               kefé 7/8
 sew
                   tásé
                                              snap, v.
                                                               naśv
 shake, intr.
                   shiw, cin
                                              snore
                                                               ko
    tr.
                   ka, jεm
                                              snot
                                                              kemfemtèn 7/8
 shame
                   kènwùměn 7
                                              soak
                                                              COW
 sharp, be
                  k33
                                              soft, make
                                                              yey
 sharpen
                  tásến, fon
                                              soil
                                                              nshε /nshε 9/10
    to a point
                  cáántén
                                              song
                                                              nkèè 1/2
 she
                  wvù
                                              sore
                                                              kènkòkεn 7/8
 sheep
                  njèè/njeè-bonjeè 1/2~9/10 sorrow
                                                              shen
 shield
                  kèbâw 7/8
                                              soul
                                                              keyɔ̃v 7/8
 shirt
                  kekóm 7/8
                                              soup
                                                              nshièw
 shiver
                  cin
                                              sour, become
                                                              ŋgwáy
 shoot
                  tá. tóm
                                              speak
                                                              jemě
 short, be
                  kfűfé
                                              spear
                                                              g35/εg3η 5/6
 shortness
                  bvukfűfè 14
                                              spend the day
                                                              shé
 shoulder
                  gwêw 9
                                             spend the night ce
shout at
                  káγ
                                             spit
                                                              tổy
show, v.
                  don
                                             split
                                                              wáy
shut
                  fáá
                                                 firewood
                                                              saně
sick, become
                  CÉM
                                             spoil
                                                              béfé
sides, take
                  bεyέn
                                             spoon
                                                              kèntàsè 7/8
sift
                  yeken
                                             spread to dry
                                                             téné
   sth. moist
                  kfu
                                             sprinkle
                                                             miàw (+miáw)
sigh w/disgust
                  shốm
                                             spy, v.
                                                             kee, niàn kến
silent, become
                 cífé
                                             squeeze
                                                             kám
sing
                  yểm (+ŋkèè)
                                                together, intr. facé
sink
                  16 (+jòò)
                                             squirrel
                                                             fetene/munteném 11/12
sip
                 fów
                                             stab
sister
                 jèmè/jémè 9/10
                                             stammer
                                                             bέw
sit
                 shii
                                             stamp, v.
                                                             cin
six
                 spacan
                                             stand, v.
                                                             111m
skin, n.
                 jèw/jēw
                                                up, intr.
                                                             iá
                 YOW
                                                             léké
                                                sth. up
   put on drum
                 daa
                                             star
                                                             jšn/jšn 9/10
slander
                 kon
                                             start
                                                             kεw
slap
                 son, tám
                                                a tune
                                                             yaw
```

```
then, therefore to &
start to cultivate ban
                                                             fèfê 16
                                            there
startled, be
                 já
                                                             cວັງ/bວcວັງ 1/2
                                            thief
stay for short time modce
                                                             ven/yen 9/10
                                             thigh
                 cắn
                                                             fyeè/mweèm 11/12
                                             thing
stem of sth.
                nson/nson 9/10
                                                             tam, kwaw
                 fete/muntem 11/12
                                             think
stick
                                                             fwew/few 3/4
stick poles in ground tom
                                             thorn
                                                porcupine
                                                             ngaw/ngaw 9/10
sticky, be (meat) fomten
                                                             ntfukè
                                             thousand
still, yet
                                             t.hree
                                                             tee
                 keté 7/8
stomach
                                                             sèw~sìw/séw 9/10~1/2
                                             throat
                 te/eta 5/6
stone
                                                             kàwlà
                                             throne
                 nsoy//ensoy 5/6
   cooking
                                                             tfűmé
                                             throw (lance)
                 ກgວິw 1/2
   grinding
                                                away
                                                             lan
                 kekowte 7/8
stool
                                                             tôfé
                 kèndvunsh€ 7/8
                                                down
   carved
                                                             bãỳkἕn
                                             thunder, v.
                 gεε
store, v.
                                             thus
                                                             nεn
straight, become nelé
                                                             ngèmiè 1/2
                  fencanca/muncancam 11/12 tick
                                                             YTIE
                                             tie together
                 témé, táw
strong, become
                                                              káγ
                  kènkfùytè
                                                up
stump
                                                             kefew
                                             time
                  keyon 7
stupidity
                                                short while nfèw
                  vám.
suck
                                                             ncèkε
                                                long ago
                  ngaw 1/2
suffering
                                                one time, etc. kan
                  kwen
sufficient, be
                                                              dásé
                                             tired, be
                  nwanwá 1/2
sugarcane
                                                              fff...lé
                                                of sth.
summit, at the
                  fòfáláán 16
                                             to (recipient) ε
                  díw
                                                 (direction) fò 16
                  fév
surpass
                                                              nàmà 1/2
                                             tobacco
                  m i
swallow
                                                              èbèn
                                             today
swear (truth)
                  kan
                                                              ntaashé
                                             together
swear word
                  kàn
                                             tomato (garden egg) yòn/yɔn 9/10
                  vεsě
sweep
                                             tomorrow
                                                              ntfùű
                  vekè
sweet, be
                                                              fà shế ntfùữ
                                                 day after
                  ben
swell.
                                                              kefásé 7/8
                                              tong
swim
                  sam
                                                              lemé/εlem~jilemé 5/6~13
                  n.jey
                                              tongue
swing, v.
                                                              entán èbèn
                  ketáálé 7/8
                                              tonight
table
                                                              tē'n, dvű
                  wvún/jiwánte 3/13
                                              too
tail
                                              tooth
                                                              kege 7/8
                  kesán 7/8
    of cow
                                                              kèbogàlě 7/8
                  fi, jo
yeyi
                                              tortoise
take
                                                              kom...lé
                                              touch
teach
                                                              Ιξηςε, cin.
                                              trample
                  mTsèm 4
tears
                                                              ketám 7/8
                                              trap
                  saa
tear, v.
                                                              kèntfùm 7/8
                                                 fish
    bamboo
                  tan
                                                              lεn
                                              travel
                  séé, gay...lé
tell
                                                              kete 7/8
                                              tree
                  fε
    story
                                                              ciŋ
                                              tremble
    riddle
                  ma
                                                              ketom 7/8
                                              tribe
ten
                  voofè
                                                              ngÈw
                  ŋgɔ̈w/ŋgɔw 9/10
                                              trouble
termite
                                                              bānkě
                                              turn
 testify
                  shómsé
                                                              bay
                                                 over
                  bɛlě
 thank
                                                              y5y
f€€
                                              twist
                   1ε, έ, ε
 that, conj.
                  sấŋ
                                              two
 thatch, v.
                                                              kèbâw 7/8
                  b3
                                              umbrella
 them, they
```

uncle (mater under(neath) untie up uplift uproot urinate urine us (excl) us (incl) vagina vegetable vein village waist, aroun wake s.o. up walk wander about want war wash (self) weter we (excl) we (incl) weak, be wealthy, become week weight well, adv. well, get wet, become what	fay wě bay báá jay mɔnjam 15 bèsèn beènè kené/ɛken 5/6 mbàsè/mbásé 9/10 gwɛŋ/gɛŋ 3/4 ntèw 1/2 d cln kamsě lɛn la njim gɔm jèm 9/10 cökɛ jòò~dìò/jɔò 9/10 bèsèn beènè	when where which white, be White man why who whole widen wife wildcat wind wine wing wipe wire witch with with vithout woman work, n. v. wound wrap up wring write yam year yesterday day before you sg. you pl. young	sékè, diò, séké feè lå bå bàlà nje lå yee -tfu/-cii koòsè kwe/bokenè 1/2 kejinè 7/8 fwefwe 3 mbvūm 15 kebebè 7/8 sho fèncaw 11/12 twě/bocéw 1/2 bó dvů sé kwoòn/bokenè 1/2 lemè 5/13 lem elemte 7/8 kólé~kócé kám, táy sán kènkfů 7/8 kelöm~kelön 7/8 fogò kefwé wò bèn (wan) cààn	۱ ڏ (Q)
--	--	---	--	----------------

INDEX

abstract nouns, 13	grammatical relations, 76-79		
adjectives, 26-27, 92-93	imperatives, 52-53, 98		
adverbs, 84-86	infinitives, 52, 89		
articles, 24-25	instrumentals, 81, 92		
aspect, 35-38, 41-50, 90	interrogatives, 25, 30, 104-105		
habitual, 89	indirect questions, 93		
progressive, 41-50, 52, 59-62, 67,	labialization, 8, 26, 28		
71, 72, 97, 100	labiodentalization, 1, 3		
auxiliaries, 87-90, 101	locative classes, 13-14, 16, 19, 21,		
tense, ch. 12, 87, 89, 96, 97	82-83, 92		
causatives, 38-39, 44, 99	locative pre- and postpositions, 14, 82-84		
comparatives, 69, 84	nasal-consonant sequences, 3, 7, 8, 11,		
complementizers, 95, 98-99, 105	77, 87		
condition clauses, 90, 98, 102-103	negation, 53, 57-58, 59, 60-61, 62, 66,		
conditional mood, 90, 103	67-68, 70, 71-72, 90, 97, 99		
conjunction	noun class concords, chs. 4-8; summary,		
nouns, 81, 101	noun class forms, 7-14		
pronouns, 17-18	irregular, 8, 9, 10, 11		
sentences, 95	numerals, 28-29		
consonant system, 1-2	objects		
copula, 88	direct, 76, 77, 80, 91		
demonstratives, 23-24	indirect, 76, 78, 80 oblique, 16, 76, 80, 81, 83, 91-92		
determiners, 23-25			
diminutives, 11	partitive, 16		
directional verbs, 69, 82	palatalization, 26, 28, 29		
focus	possessives		
adverbs, 108	nouns, 19, 79		
auxiliary ([+FOC]), 55, 57, 66, 70, 88, 90, 93-94, 102, 105-106	pronouns, 19-22, 92		
direct object, 85	pronouns		
(within) noun phrase, 32	compound, 17-18		
oblique objects, 85, 108	independent, 15-18, 78, 98		
subject, 76, 85, 104, 107	logophoric, 15, 17-18, 21		
verb, 56, 63, 77, 85, 90, 91, 93, 106, 108	quantifiers, 30, 86		

reciprocals, 13, 39, 43, 78 references, 109 reflexives, 18, 39, 78 relative clauses, 26, 27, 32, 65-68, 83, reported speech, 15, 17-18, 21, 95, 98, sequential clauses, 88, 95-97 statives, 84 subjects, 76, 77 subject-verb (nasal) agreement, 3, 77, subjunctives, 98, 103 syllable structure, 2-4 temporal clauses, 100-102 tone patterns (verbs), 51-52 irregular, 64, 75 rules, 74-75 summary, 73 tone spreading, 6, 19 tone system, 5-6, 51-52, 73, 74-75 verb extensions, 35-40, ch. 10 attenuative, 35-36, 43 bifurcative, 37-38 causative, 38-39, 44 distributive, 36, 43 frequentative, 38, 44-45 iterative, 37, 77 positional, 39, 44, 64 reciprocal, 18, 39-40 transitive, 39, 48 verb forms, 34-40 (and following chapters) irregular, 45-48, 63-64, 75 verb radicals, 34-35 verb reduplication, 38, 44-45 verb tenses

compound, 62

consecutive clauses, 69-72
main clauses, 56-64, 96
overview, 53-55, 73
relative clauses, 65-68, 93-94
vowel system, 1
word list, 110-119
word order
noun phrase, 31-32
sentence, 76, 107-108
subject-object inversion, 108